Topological vertex and its applications

Satoshi Minabe

February 19, 2007

Abstract

We give two new results on Gromov–Witten invariants of Calabi–Yau threefolds. The first one is a computation of local Gromov–Witten invariants of cubic surfaces at all genera. We give an explicit formula for the generating function of these invariants in a closed form. The second one is on the flop invariance of Gromov–Witten invariants of toric Calabi–Yau threefolds. We prove transformation formulas for generating functions of Gromov–Witten invariants on general toric Calabi–Yau threefolds under flops. Both results are based on the theory of the topological vertex. We present proofs of these two results together with required background on Gromov–Witten invariants.

This paper has been accepted as the author's doctoral thesis at the Graduate School of Mathematics, Nagoya University (defended in March 2007).

Contents

-	0.	ciview and Dackground	Э
1	Ove	erview	3
	1.1	Introduction	3
	1.2	Topological vertex and local Gromov–Witten invariants of del Pezzo surfaces	5
		1.2.1 Local Gromov–Witten invariants of algebraic surfaces	5
		1.2.2 Gromov–Witten invariants of local toric surfaces	6
		1.2.3 Local Gromov–Witten invariants of del Pezzo surfaces	6
	1.3	Flop invariance of the topological vertex	7
		1.3.1 An identity \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots	7
		1.3.2 Flop invariance of the Gromov–Witten invariants	8
	1.4	Organization of the paper	8
	1.5	Acknowledgments	9
2	Bac	kground	9
	2.1	Gromov–Witten invariants	9
		2.1.1 Stable maps \ldots	9
		2.1.2 Deformation theory of stable maps	10
		2.1.3 Virtual fundamental class	11
		2.1.4 Gromov–Witten invariants of Calabi–Yau threefolds	11
	2.2	Virtual localization	12
		2.2.1 Equivariant homology groups	12
		2.2.2 Classical localization theorem	13

	2.3	2.2.3Virtual localization formulaHodge integrals2.3.1Partitions2.3.2Hodge integrals2.3.3Formula for two-partition Hodge integrals	14 15 16 16 17
II	\mathbf{T}	opological vertex and local Gromov–Witten invariants	18
3	Loc 3.1 3.2	cal Gromov–Witten invariants of surfacesLocal Gromov–Witten invariants of surfaces	 18 19 19 19 20 21
4	Gro 4.1 4.2 4.3	omov–Witten invariants of local toric surfaces Equivariant Gromov–Witten invariants of local toric surfaces Topological vertex formula Gopakumar–Vafa invariants	 22 22 23 24
5	Loc 5.1 5.2 5.3 5.4 5.5	calization on moduli spaces of stable maps to toric surfacesThe T-fixed componentsPerfect obstruction theories on the T-fixed componentsVirtual normal bundles of the T-fixed componentsColspan="2">Colspan="2"Colspan="2">Colspan="2">Colspan="2"Colspan="2">Colspan="2"Colspan="2">Colspan="2"Colspan="2">Colspan="2"Colspan="2">Colspan="2" </td <td> 24 24 26 27 29 32 </td>	 24 24 26 27 29 32
6	Loc 6.1 6.2 6.3 6.4 6.5 6.6	ral Gromov–Witten invariants of cubic surfacesExample of $-K$ -nef toric degeneration	34 35 36 37 37 39 40 41 41 41 42
Π	I 7	Topological vertex and its flop invarinace	43
7	Top 7.1 7.2	Dological vertex Topological vertex	44 44 45 46

	7.3	Appendix: Combinatorial formulas	17
8	Tori	c Calabi–Yau threefolds and Partition functions 4	8
	8.1	Toric Calabi–Yau threefolds	18
		8.1.1 Toric Calabi–Yau threefolds	18
		8.1.2 Intersection theory	19
	8.2	Partition functions	50
		8.2.1 Toric graphs	50
		8.2.2 Partition functions	51
	8.3	Gromov–Witten invariants of toric Calabi–Yau threefolds	51
		8.3.1 Gromov–Witten invariants of toric Calabi–Yau threefolds	52
		8.3.2 Partition functions and Gromov–Witten invariants	53
9	Tra	nsformations of partition functions 5	53
	9.1	Flop invariance	53
		9.1.1 Flop	53
		9.1.2 Transformation of partition function	55
	9.2	Application to toric surface and its blowup	57
		9.2.1 Small modification of toric Calabi–Yau threefolds	57
		9.2.2 Toric surface and its blowup	59
	9.3	Example and geometric engineering	59

Part I Overview and Background

1 Overview

1.1 Introduction

Let X be a smooth projective Calabi–Yau threefold, i.e. complex three-dimensional algebraic manifold whose canonical bundle is trivial. Around 1991, string theorists made a new prediction on "numbers" of algebraic curves in X of genus zero. It had been known that a certain generating series F_X^0 whose coefficients are these numbers has a physical meaning (the "prepotential" or the "genus zero part of the free energy"). The mirror symmetry identified F_X^0 as a specific combination of hypergeometric series. For mathematical understanding of this phenomenon, an appropriate definition of "numbers" of curves was needed. Nowadays, these numbers are rigorously defined and called the Gromov-Witten invariants. See [12, 32] for reviews of these developments. The Gromov–Witten invariants of X is defined by counting the number of maps from curves to X as follows. The moduli space of stable maps $\overline{\mathcal{M}}_{g,0}(X,\beta)$ into X from the genus g curves and the prescribed homology class (called the degree) $\beta \in H_2(X,\mathbb{Z})$ of the images has virtual dimension zero. The Gromov–Witten invariant $N_{g,\beta}(X)$ of X with the genus g and the degree β is defined by the integral of the unit cohomology class 1 against the 0-dimensional virtual fundamental class of $\overline{\mathcal{M}}_{g,0}(X,\beta)$. The generating function F_X which sums $N_{g,\beta}(X)$ over all g and β with the weight $\lambda^{2g-2}Q^{\beta}$ (here λ and Q are parameters) is called the free energy and $Z_X := \exp F_X$ is called the partition function.

In this paper, we study the partition functions of toric Calabi–Yau threefolds ¹. The partition function Z_X of a toric Calabi–Yau threefold X can be written in an explicit form by the theory of "topological vertex" developed by Aganagic–Klemm–Mariño–Vafa [2]. They found an algorithm to write down a formula for Z_X from the toric fan of X, which is as follows. First, one assigns a certain amplitude to each vertex of the dual graph of the toric fan of X as a building block. Then the algorithm tells the gluing rule of vertex amplitudes between two vertices which are joined by an edge of the graph. The gluing rule was motivated by a geometric idea of localization (see [16]). On the other hand, the vertex amplitude, from which the name of the theory comes ², has a combinatorial expression in terms of special values of skew-Schur functions. It was obtained by using a certain duality between closed A-model topological string theory (Gromov–Witten theory) and Chern–Simons theory. See [40, 63] for expositions on these developments. Although the argument based on a string duality is not mathematically rigorous, a mathematical theory for the topological vertex was developed later by Li–Liu–Liu–Zhou [59]. In their approach, a combinatorial expression for the vertex amplitude comes from so-called Hodge integrals over moduli spaces of stable curves.

The topological vertex is an efficient tool to study Gromov–Witten invariants of toric Calabi– Yau threefolds. So far, many applications have been given: proofs of a conjecture of Nekrasov [69] on geometric engineering by Iqbal–Kahsani-Poor [36, 37], Eguchi–Kanno [22, 23] and Zhou [83], proofs of the Gopakumar–Vafa conjecture for toric Calabi–Yau threefolds by Konishi [48, 49] and Peng [75], and so on. Furthermore, it also shows us a surprising connection between string theory and some statistical model. By relating the combinatorial expression of the topological vertex to three dimensional partitions, Okounkov–Reshetikhin–Vafa [72] proposed a duality between topological A-model strings on Calabi–Yau threefolds and a statistical mechanical model of crystal melting. Inspired by this work, Maulik–Nekrasov–Okounkov–Pandharipande [64] proposed a conjecture which states the equivalence between Gromov–Witten theory and Donaldson–Thomas theory (which concerns integration over moduli of sheaves on threefolds) for Calabi–Yau threefolds. As an evidence, they proved the conjecture for local toric del Pezzo surfaces ³ based on the topological vertex. This conjecture was generalized to general projective threefolds later in [65].

The aim of this paper is to present two new applications of the topological vertex. The first main result gives explicit formulas for the partition functions of local Gromov–Witten invariants of del Pezzo surfaces S_d of degree d = 3, 4, 5, which are non-toric ⁴.

Theorem A. We have explicit formulas for the partition functions of local Gromov–Witten invariants of non-toric del Pezzo surfaces S_3 , S_4 and S_5 .

Formulas for toric del Pezzo surfaces $(d \ge 6)$ was obtained by Zhou [82] (see also [1, 14, 15, 35]). The exact form of our formula is given in Theorems 1.4 and 6.8. Our main idea is to use the deformation invariance of local Gromov–Witten invariants and reduce the computations to those of -K-nef toric surfaces whose local Gromov–Witten invariants can be evaluated by the topological vertex formula (Theorems 1.3 and 4.6).

¹ A Calabi–Yau threefold X is toric if it contains the algebraic torus $(\mathbb{C}^*)^3$ as a dense open subset and the action of $(\mathbb{C}^*)^3$ on itself extends to X. Particular examples are total spaces of the canonical line bundles of toric surfaces, which are called *local toric surfaces*.

²The vertex amplitude itself is also called the topological vertex.

³A nonsingular algebraic surface S is called a del Pezzo surface if its canonical bundle K_S is negative.

⁴For a del Pezzo surface S, the positive integer $d = c_1(K_S)^2$ is called the degree of S. It is known that del Pezzo surfaces are classified by their degrees which take all integers from 1 to 9. A del Pezzo surface S_d of degree d is obtained as a blow up of the projective plane \mathbb{P}^2 at (9 - d) points if $d \neq 8$ and S_8 is isomorphic to either \mathbb{P}^2 blown up at a point or $\mathbb{P}^1 \times \mathbb{P}^1$. In particular, S_d is toric if $d \geq 6$. See e.g.[67, §0] for a brief account of the classification of del Pezzo surfaces.

The second main result is on the flop invariance of Gromov–Witten invariants of toric Calabi– Yau threefolds. The same problem for projective Calabi–Yau threefolds was studied by Li–Ruan [55]. For related physical works, we refer to the references in *loc. cit*..

Theorem B. Let $\phi : X \dashrightarrow X^+$ be a flop with respect to a (-1, -1)-curve between toric Calabi–Yau threefolds. Then Gromov–Witten invariants of X and X^+ coincide under the identification of degrees given by ϕ .

See Theorem 1.6 for a precise statement. We show the invariance of the partition functions under flops (Theorem 9.3), which implies that for Gromov–Witten invariants (Corollary 9.4). By virtue of the topological vertex and local analysis of fans of toric Calabi–Yau threefolds, the flop invariance of the partition functions is obtained from a combinatorial identity on skew-Schur functions (Theorems 1.5 and 7.9).

The above two results were obtained by joint works [50, 51] with Yukiko Konishi. The author's main contribution to the first result was the idea of using the topological vertex and the invariance of Gromov–Witten invariants under deformations to compute the local Gromov–Witten invariants of the non-toric del Pezzo surfaces which admit -K-nef toric degenerations. That to the second one was to prove the combinatorial identity in Theorems 1.5 which is a key to our proof of the flop invariance. In the next two sections, outlines of results and their proofs will be explained in further detail. Since we are going to address two problems which are related but different in nature, we treat them separately.

1.2 Topological vertex and local Gromov–Witten invariants of del Pezzo surfaces

The first problem is computations of local Gromov–Witten invariants of del Pezzo surfaces. These invariants have been intensively studied in physics by various methods: local mirror symmetry, the holomorphic anomaly equation, Seiberg–Witten curve technique, and so on. See e.g. [11, 44, 54] and the references therein.

1.2.1 Local Gromov–Witten invariants of algebraic surfaces

First we give a definition of local Gromov–Witten invariants for any algebraic surface. See §3 for details.

Let S be a nonsingular projective surface and K_S be the canonical bundle of S. We denote by $\overline{\mathcal{M}}_{g,n}(S,\beta)$ the moduli space of n-pointed stable maps into S with genus g, degree $\beta \in H_2(S,\mathbb{Z})$. Consider the following universal diagram:

$$\begin{array}{ccc} \overline{\mathcal{M}}_{g,1}(S,\beta) & \stackrel{\mu}{\longrightarrow} S \\ & & \pi \\ & & \\ \overline{\mathcal{M}}_{g,0}(S,\beta) \ , \end{array}$$

where $\mu = ev_1$ is the evaluation at the marked point and $\pi = \pi_1$ is the map forgetting the marked point, which form the universal stable map and the universal curve respectively (cf. §2.1.1).

Definition 1.1. The local Gromov–Witten invariant $I_{g,\beta}(S)$ of S with genus g, degree β is defined as follows:

$$I_{g,\beta}(S) := \int_{[\overline{\mathcal{M}}_{g,0}(S,\beta)]^{vir}} c(-R^{\bullet}\pi_*\mu^*K_S) \in \mathbb{Q},$$

where $c(\cdot)$ is the total Chern class and $[\overline{\mathcal{M}}_{g,0}(S,\beta)]^{vir}$ is the virtual fundamental class of $\overline{\mathcal{M}}_{g,0}(S,\beta)$ (cf. §2.1.3).

1.2.2 Gromov–Witten invariants of local toric surfaces

Next, we introduce the Gromov–Witten invariants of local toric surfaces. The topological vertex, which is our main tool, gives a closed formula for their generating functions. See §4 for details.

Let S be a nonsingular complete toric surface, which is equipped with an action of the 2-dimensional algebraic torus $\mathbb{T} = (\mathbb{C}^*)^2$. We regard the canonical bundle K_S of S as a \mathbb{T} -equivariant line bundle $\mathcal{O}_S(-\sum_{i=1}^r C_i)$, where C_1, \cdots, C_r are irreducible toric divisors on S. It is known that they generate $H_2(S, \mathbb{Z})$.

Definition 1.2. The Gromov–Witten invariants $N_{g,\beta}(K_S)$ of K_S with genus g, degree β is defined by the following \mathbb{T} -equivariant integral:

$$N_{g,\beta}(K_S) := \int_{[\overline{\mathcal{M}}_{g,0}(S,\beta)]_{\mathbb{T}}^{vir}} e_{\mathbb{T}}(-R^{\bullet}\pi_*\mu^*K_S),$$

where $e_{\mathbb{T}}(\cdot)$ is the T-equivariant Euler class.

If S is a -K-nef (i.e. $-K_S$ is nef) toric surface then

$$N_{g,\beta}(K_S) = I_{g,\beta}(S), \tag{1.1}$$

for β satisfying $\int_{\beta} c_1(K_S) < 0^{-5}$ (cf. Lemma 4.4). Let us introduce

$$F_{K_S}(\lambda, Q) := \sum_{\beta \in H_2(S,\mathbb{Z}), \beta \neq 0} \sum_{g \ge 0} N_{g,\beta}(K_S) \lambda^{2g-2} Q^{\beta},$$

where Q is a formal variable satisfying $Q^{\beta}Q^{\beta'} = Q^{\beta+\beta'}$, and define

$$Z_{K_S}(\lambda, Q) := \exp F_{K_S}(\lambda, Q)$$

Theorem 1.3 (Topological vertex formula). Let s_i be the self-intersection number of C_i and $t_i := Q^{[C_i]} (1 \le i \le r)$. Then we have

$$Z_{K_S}(\lambda,Q) = \sum_{(\nu^1,\dots,\nu^r)\in\mathcal{P}^r} \prod_{i=1}^r ((-1)^{s_i} t_i)^{|\nu^i|} e^{\sqrt{-1\lambda}s_i \frac{\kappa(\nu^i)}{2}} W_{\nu^i,\nu^{i+1}}(e^{\sqrt{-1\lambda}}).$$

Here ν^i runs over all partitions and we regard $i \in \mathbb{Z}/r$. See §2.3.1 and §2.3.3 for notations on partitions and combinatorial functions. In §5, we give a proof of Theorem 1.3 which is essentially due to Zhou [82] and Liu–Liu–Zhou [60].

1.2.3 Local Gromov–Witten invariants of del Pezzo surfaces

Let S_d be a del Pezzo surface of degree d. By eq. (1.1) and Theorem 1.3, we have a closed formula for the generating functions of $I_{g,\beta}(S_d)$ for toric del Pezzo surfaces S_d ($6 \le d \le 9$). We extend this result for $d \ge 3$. It is enough to state the result for S_3 (see Remark 6.6). Let $p: S_3 \to \mathbb{P}^2$ be a blow-up at six points in a general position and e_1, \dots, e_6 be the classes of the exceptional curves of p, l be the class of a line in \mathbb{P}^2 pulled back by p. They form a basis for $H_2(S_3, \mathbb{Z})$. We consider the partition function of local Gromov–Witten invariants of S_3 :

$$Z_{S_3}^{\mathrm{loc}}(\lambda, Q) := \exp\left[\sum_{\beta \in H_2(S_3, \mathbb{Z}), -K_{S_3}, \beta > 0} \sum_{g \ge 0} I_{g,\beta}(S_3) \lambda^{2g-2} Q^{\beta}\right].$$

⁵Both of the two conditions $(-K\text{-nef} + \int_{\beta} c_1(K_S) < 0)$ are essential for the equality.

Our goal here is to obtain an explicit formula for $Z_{S_3}^{\text{loc}}$. A key fact is that S_3 is deformation equivalent to a -K-nef (but not a del Pezzo) toric surface S_3^0 , whose toric fan can be found in Figure 2 (cf. §6.2). It has nine irreducible toric divisors C_1, \ldots, C_9 . Then, by deformation invariance of local Gromov–Witten invariants for -K-nef surfaces (Proposition 3.7), local Gromov-Witten invariants of S_3 can be obtained from those of S_3^0 under appropriate identifications of second homology classes. Applying eq. (1.1) and Theorem 1.3 to S_3^0 and subtracting the contributions from chains of (-2)-curves which are orthogonal to the canonical divisor $K_{S_3^0}$, we obtain the following

Theorem 1.4 ([51]).

$$Z_{S_3}^{\text{loc}}(\lambda, Q) = \frac{Z_{K_{S_3^0}}(\lambda, t_1, \cdots, t_9)}{\prod_{i=1,4,7} Z_{(-2,0)}(\lambda, t_i) Z_{(-2,0)}(\lambda, t_{i+1}) Z_{(-2,0)}(\lambda, t_i t_{i+1})},$$
(1.2)

with the following identification of the parameters

$$t_1 = Q^{e_2 - e_5}, \quad t_2 = Q^{l - e_2 - e_3 - e_6}, \quad t_3 = Q^{e_6}, \quad t_4 = Q^{e_3 - e_6}, \quad t_5 = Q^{l - e_1 - e_3 - e_4}, \\ t_6 = Q^{e_4}, \quad t_7 = Q^{e_1 - e_4}, \quad t_8 = Q^{l - e_1 - e_2 - e_5}, \quad t_9 = Q^{e_5}.$$

$$(1.3)$$

Here $Q = (Q_1, \ldots, Q_6, Q_7)$ is a set of formal variables and $Q^{\beta} = Q_1^{a_1} Q_2^{a_2} \ldots Q_7^{a_7}$ for $\beta = a_1 e_1 + \cdots + a_6 e_6 + a_7 l \in H_2(S_3, \mathbb{Z})$ and $Z_{(-2,0)}(\lambda, t)$ has been defined by

$$Z_{(-2,0)}(\lambda,t) = \prod_{k=1}^{\infty} (1 - te^{\sqrt{-1}\lambda k})^{-k}$$

This is the first main result of this paper. Its proof will be given in §6. Note that an explicit formula for $Z_{K_{S_3^0}}(\lambda, t_1, \dots, t_9)$ is given by Theorem 1.3. The change of the variables (1.3) corresponds to an identification $H_2(S_3^0, \mathbb{Z}) \cong H_2(S_3, \mathbb{Z})$. The denominator of the formula (1.2) is the contributions from (-2)-curves on S_3^0 .

1.3 Flop invariance of the topological vertex

The second problem is on the flop invariance of Gromov–Witten invariants of toric Calabi– Yau threefolds. In this paper, we mean by a flop a birational transformation obtained by contracting a rational curve whose normal bundle is isomorphic to $\mathcal{O}(-1) \oplus \mathcal{O}(-1)$, which is called a (-1, -1)-curve, and resolving the resulting singularity appropriately (cf. §9.1.1).

1.3.1 An identity

Let \mathcal{P} be the set of partitions. For $\lambda_1, \lambda_2, \lambda_3 \in \mathcal{P}$, the topological vertex $C_{\lambda_1, \lambda_2, \lambda_3}(q) \in \mathbb{Q}(q^{\frac{1}{2}})$ is defined by certain special values of skew-Schur functions associated to $\lambda_1, \lambda_2, \lambda_3$. See §7.1 for details. Take four partitions $\lambda_1, \lambda_2, \lambda_3, \lambda_4$ and introduce:

$$Z_0'(q,Q_0) = \frac{Z_0(q,Q_0)}{Z_{(-1,-1)}(q,Q_0)} , \quad Z_0^{+\prime}(q,Q_0^+) = \frac{Z_0^+(q,Q_0^+)}{Z_{(-1,-1)}(q,Q_0^+)} , \quad (1.4)$$

where

$$Z_0(q,Q_0) = \sum_{\mu \in \mathcal{P}} (-Q_0)^{|\mu|} C_{\lambda_1,\lambda_2,\mu^t}(q) C_{\lambda_3,\lambda_4,\mu}(q) ,$$

$$Z_0^+(q,Q_0^+) = \sum_{\mu \in \mathcal{P}} (-Q_0^+)^{|\mu|} C_{\lambda_1,\mu^t,\lambda_4}(q) C_{\lambda_3,\mu,\lambda_2}(q) ,$$

and $Z_{(-1,-1)}(q,Q) = \prod_{k=1}^{\infty} (1 - Qq^k)^k$. These are formal power series in Q_0 (resp. Q_0^+). A key result to our proof of the flop invariance of Gromov–Witten invariants of toric Calabi–Yau threefolds is the following combinatorial identity.

Theorem 1.5 ([50]). Under the identification $Q_0^+ = Q_0^{-1}$, we have $Z_0^{+\prime}(q, Q_0^+) = (-Q_0)^{-(|\lambda_1|+|\lambda_2|+|\lambda_3|+|\lambda_4|)} q^{\frac{1}{2}(\kappa(\lambda_1)-\kappa(\lambda_2)+\kappa(\lambda_3)-\kappa(\lambda_4))} Z_0^{\prime}(q, Q_0)$.

Theorem 1.5 will be proved in §7.2. Our proof is based on the technique which has been developed in [36, 37, 22, 23, 83] to prove a conjecture of Nekrasov [69] on the equivalence between the partition function of Gromov–Witten invariants of a certain toric Calabi–Yau threefold and the Nekrasov partition function in instanton counting.

1.3.2 Flop invariance of the Gromov–Witten invariants

Let X and X^+ be toric Calabi–Yau threefolds which are birationally equivalent under a flop $\phi : X \dashrightarrow X^+$ with respect to a (-1, -1)-curve C_0 in X. Let $C_0^+ \subset X^+$ be the flopped (-1, -1)-curve. See §9.1 for details. Note that we have an isomorphism between $H_2(X, \mathbb{Z})$ and $H_2(X^+, \mathbb{Z})$ induced by ϕ .

Let us consider the partition functions of X and X^+ :

$$Z_X(\lambda, Q) = \exp\left[\sum_{\beta \in H_2(X,\mathbb{Z}), \beta \neq 0} \sum_{g \ge 0} N_{g,\beta}(X) \lambda^{2g-2} Q^{\beta}\right],$$

$$Z_{X^+}(\lambda, Q^+) = \exp\left[\sum_{\beta \in H_2(X^+,\mathbb{Z}), \beta \neq 0} \sum_{g \ge 0} N_{g,\beta}(X^+) \lambda^{2g-2} (Q^+)^{\beta}\right].$$

where $N_{g,\beta}(X)$ and $N_{g,\beta}(X^+)$ are Gromov–Witten invariants of X and X^+ respectively. The topological vertex enables us to write down a combinatorial formula for $Z_X(\lambda, Q)$ and $Z_{X^+}(\lambda, Q^+)$. See §8.2 for a summary of the theory of the topological vertex. Under appropriate identifications of the expansion parameters Q and Q^+ , one can show that they are equal except for factors coming from multiples of $[C_0]$ and $[C_0^+]$ (Theorem 9.3). Since a flop is a local operation, the difference between the two can only appear at the local contributions from neighborhoods of C_0 and C_0^+ . If we set $q = e^{\sqrt{-1}\lambda}$, they are given by $Z'_0(q,Q_0)$ and $Z''_0(q,Q_0^+)$, defined in (1.4), respectively. Then, showing the equality between the two partition functions reduces to the identity in Theorem 1.5. As a result, we obtain the following

Theorem 1.6 ([50]). For $\beta \in H_2(X, \mathbb{Z})$ which is not a multiple of $[C_0]$, we have

$$N_{g,\beta}(X) = N_{g,\phi_*(\beta)}(X^+).$$

For a multiple of flopping curve class, we have

$$N_{g,d[C_0]}(X) = N_{g,d[C_0^+]}(X^+).$$

Theorem 1.6 is the second main result of this paper. Its proof and some applications will be given in §9.

1.4 Organization of the paper

The paper is organized as follows. The rest of the part I covers the background material. Notations introduced in §2, especially those on moduli of stable maps (§2.1) and partitions (§2.3) are used throughout the paper. Parts II and III are devoted to prove the results presented in §1.2 and §1.3 respectively. In principle, they can be read separately. More details will be explained at the beginning of each section.

1.5 Acknowledgments

I would like to thank my advisor Prof. Hiroaki Kanno for his guidance and support. I learned physical aspects on the subject mostly from his papers, lectures and discussions with him. I wish to thank Dr. Yukiko Konishi for collaboration and instruction in the virtual localization technique. She had the patience to answer a lot of my questions on it. I am also grateful to Dr. Yuichi Nohara, Prof. Hiroyuki Ochiai, Prof. Akihiro Tsuchiya, Prof. Tohru Uzawa and the preliminary reviewers for reading the manuscript carefully and giving me useful comments. The paper was written at the Mittag-Leffler Institute (Djursholm, Sweden) during the fall term of 2006-07 program on Moduli Spaces. I would like to thank the staff and the organizers there for providing me a wonderful research environment.

2 Background

In §2.1, a definition of Gromov–Witten invariants for nonsingular projective varieties is given. Special attention will be paid to projective Calabi–Yau threefolds. §2.2 is devoted to the virtual localization formula, which is one of the most powerful tools in the study of Gromov–Witten invariants. In §2.3, a definition of Hodge integrals and necessary formulas are given.

2.1 Gromov–Witten invariants

2.1.1 Stable maps

The notion of stable maps was introduced by Kontsevich [52]. Let X be a non-singular projective variety over \mathbb{C} . Stable maps into X are defined as follows. Recall that an *n*-pointed pre-stable curve of genus g is a tuple (C, p_1, \dots, p_n) where C is a projective, connected, reduced, nodal curve of arithmetic genus g, and p_1, \dots, p_n are distinct non-singular points on C. Nodal points on C and marked points p_1, \dots, p_n are called special points of a pre-stable curve (C, p_1, \dots, p_n) .

Definition 2.1. (i) An *n*-pointed pre-stable map $f : (C, p_1, \dots, p_n) \to X$ into X of genus g, degree $\beta \in H_2(X, \mathbb{Z})$ is a morphism from *n*-pointed genus g pre-stable curves (C, p_1, \dots, p_n) to X such that $f_*[C] = \beta$.

(ii) An infinitesimal automorphism of pre-stable map is a tangent vector field v on the domain curve C which vanishes at the special points and satisfying df(v) = 0.

(iii) A pre-stable map is called stable if it has no nontrivial infinitesimal automorphisms.

We denote by $\overline{\mathcal{M}}_{g,n}(X,\beta)$ the moduli space of *n*-pointed genus *g* stable maps of degree β into *X*. The existence of the fine moduli spaces $\overline{\mathcal{M}}_{g,n}(X,\beta)$ of stable maps as Deligne–Mumford stacks and their properness was stated in [52] and proved in [6] (see also [28] together with [30, Appendix A]). These spaces come equipped with several natural morphisms. For each of the *n* marked points, there is an evaluation map

$$ev_i: \overline{\mathcal{M}}_{q,n}(X,\beta) \to X$$

which takes $(f; C, p_1, \dots, p_n) \in \overline{\mathcal{M}}_{g,n}(X, \beta)$ to $f(p_i) \in X$. There is the map

$$\pi_{n+1}: \overline{\mathcal{M}}_{g,n+1}(X,\beta) \to \overline{\mathcal{M}}_{g,n}(X,\beta)$$

which forgets the last marked point (and stabilizes if necessary). The forgetful morphism π_{n+1} and the evaluation morphism ev_{n+1} form the universal curve and the universal stable map respectively:

$$\overline{\mathcal{M}}_{g,n+1}(X,\beta) \xrightarrow{ev_{n+1}} X$$

$$\begin{array}{c} \pi_{n+1} \\ \overline{\mathcal{M}}_{g,n}(X,\beta) . \end{array}$$
(2.1)

2.1.2 Deformation theory of stable maps

Following [30, Appendix B], we summarize the obstruction theory on the moduli space $\overline{\mathcal{M}} := \overline{\mathcal{M}}_{g,n}(X,\beta)$ of stable maps into X. Let us set $\mathcal{U} := \overline{\mathcal{M}}_{g,n+1}(X,\beta)$ and consider the universal curve $\pi = \pi_{n+1} : \mathcal{U} \to \overline{\mathcal{M}}$ and the universal stable map $\mu = ev_{n+1} : \mathcal{U} \to X$ (cf.(2.1)). The stability condition and the projectivity of X imply that there exists a two-term complex

$$E_{\bullet} = [E_0 \to E_1] \tag{2.2}$$

of vector bundles over $\overline{\mathcal{M}}$ which satisfies the following condition.

Define coherent sheaves \mathcal{T}^1 , \mathcal{T}^2 on $\overline{\mathcal{M}}$ by the following exact sequence:

$$0 \longrightarrow \mathcal{T}^1 \longrightarrow E_0 \longrightarrow E_1 \longrightarrow \mathcal{T}^2 \longrightarrow 0.$$
 (2.3)

Then \mathcal{T}^1 and \mathcal{T}^2 fit into the following exact sequence:

$$0 \longrightarrow R^{0}\pi_{*}\underline{Hom}(\Omega_{\pi}(P), \mathcal{O}_{\mathcal{U}}) \longrightarrow R^{0}\pi_{*}\mu^{*}T_{X} \longrightarrow \mathcal{T}^{1}$$

$$\longrightarrow R^{1}\pi_{*}\underline{Hom}(\Omega_{\pi}(P), \mathcal{O}_{\mathcal{U}}) \longrightarrow R^{1}\pi_{*}\mu^{*}T_{X} \longrightarrow \mathcal{T}^{2} \longrightarrow 0,$$

$$(2.4)$$

where Ω_{π} is the sheaf of π -relative differentials on \mathcal{U} and P is the divisor of marked points.

The exact sequence (2.4) is the tangent-obstruction sequence of the deformation theory of stable maps which consists of two parts:

(i) Deformation of maps $f: C \to X$ from a fixed domain curve C,

(ii) Deformation of the *n*-pointed domain curve (C, p_1, \dots, p_n) .

The tangent and obstruction spaces for (i) are $H^0(C, f^*T_X)$ and $H^1(C, f^*T_X)$ respectively. For (ii), $\operatorname{Ext}^0(\Omega_C(P), \mathcal{O}_C)$ and $\operatorname{Ext}^1(\Omega_C(P), \mathcal{O}_C)$ are the space of infinitesimal automorphisms and the tangent space respectively (there is no obstruction space). This explains the geometrical meaning of (2.4). We call \mathcal{T}^1 and \mathcal{T}^2 the tangent sheaf and the obstruction sheaf respectively. The complex E_{\bullet} is called the (dual) *perfect obstruction theory* on $\overline{\mathcal{M}}$.

From (2.4), we can compute the virtual (or expected) dimension vdim $\overline{\mathcal{M}}_{g,n}(X,\beta)$ of the moduli space $\overline{\mathcal{M}}_{g,n}(X,\beta)$. By Riemann–Roch theorem, it is given by

vdim
$$\overline{\mathcal{M}}_{g,n}(X,\beta) = -K_X \cdot \beta + (\dim X - 3)(1 - g) + n.$$
 (2.5)

If there are no obstructions, the moduli space $\overline{\mathcal{M}}_{g,n}(X,\beta)$ is smooth of expected dimension. This is the case when $H^1(C, f^*T_X)$ vanishes at all points in the moduli spaces. However, in general, $\overline{\mathcal{M}}_{g,n}(X,\beta)$ may be singular, non-reduced, reducible, and of impure dimension. It may have a component whose dimension is bigger than the expected dimension.

2.1.3 Virtual fundamental class

We want to define Gromov-Witten invariants of X as integrals over $\overline{\mathcal{M}}_{g,n}(X,\beta)$. However, this space is not so nice as we mentioned earlier. In addition, it does not behave well under deformations of X. For instance, $\overline{\mathcal{M}}_{g,n}(X,\beta)$ does not necessarily deform in a flat family under a flat deformations of X. To overcome these problems, the virtual fundamental class was introduced: $\overline{\mathcal{M}}_{g,n}(X,\beta)$ carries a natural class

$$[\overline{\mathcal{M}}_{g,n}(X,\beta)]^{vir} \in A_{\mathrm{vdim}}(\overline{\mathcal{M}}_{g,n}(X,\beta)),$$

in the expected dimensional Chow group of rational coefficients. It has good properties including nice behaviors under deformations of X. In Gromov–Witten theory, all integrals are taken against the virtual fundamental class. It is constructed from the perfect obstruction theory E_{\bullet} (cf.(2.2)). For example, if the moduli space is smooth of expected dimension then the virtual fundamental class is equal to the ordinary one. If the moduli space is smooth but not of expected dimension, that is the case when obstruction sheaf \mathcal{T}^2 is a vector bundle, then the virtual fundamental class is given by

$$[\overline{\mathcal{M}}_{g,n}(X,\beta)]^{vir} = e(\mathcal{T}^2) \cap [\overline{\mathcal{M}}_{g,n}(X,\beta)]_{\mathcal{T}}$$

where $e(\cdot)$ is the Euler class. The construction is quite subtle in general. This was the main technical point in defining Gromov–Witten invariants which was achieved by Behrend–Fantechi [7, 8] and Li–Tian [56] in algebraic category, and by Fukaya–Ono [25], Li–Tian [57], and Siebert [76] in symplectic category. Proofs of the coincidence of these constructions in two different categories were given by Li–Tian [58] and Siebert [77].

The Gromov–Witten invariants of X are defined by integrals of the following form:

$$\langle \gamma_1, \cdots, \gamma_n \rangle_{g,\beta}^X := \int_{[\overline{\mathcal{M}}_{g,n}(X,\beta)]^{vir}} \prod_{i=1}^n e v_i^* \gamma_i,$$

where $\gamma_i \in H^*(X, \mathbb{Z})$. The geometrical meaning of this invariant can be explained as follows. Take *n* homology cycles Γ_i on X $(i = 1, \dots, n)$. Then Gromov–Witten invariant $\langle \gamma_1, \dots, \gamma_n \rangle_{g,\beta}^X$ for $\gamma_i = \text{PD}[\Gamma_i]$, where $\text{PD}[\Gamma_i]$ represents the cohomology class which is dual to the homology class $[\Gamma_i]$ under the Poincaré duality, is a virtual counting of *n*-pointed stable maps into X of genus g, degree β which intersect with the cycle Γ_i at the image of the *i*-th marked point. Note however that Gromov–Witten invariants are rational numbers since the virtual fundamental class is a rational class.

The Gromov–Witten invariants are invariant under smooth deformations of X (see, e.g. [56, Theorem 4.2]). A meaning of this statement can be explained as follows. Since smooth families of varieties are locally trivial in the C^{∞} -topology, there are canonical isomorphisms between nearby fibers of the family. With respect to this identification, the Gromov–Witten invariants are independent of the choice of a fiber.

2.1.4 Gromov–Witten invariants of Calabi–Yau threefolds

Let X be a nonsingular projective threefold. Then the virtual dimensions of the moduli spaces $\overline{\mathcal{M}}_{g,0}(X,\beta)$ do not depend upon the genus:

vdim
$$\overline{\mathcal{M}}_{q,0}(X,\beta) = -K_X \cdot \beta.$$

We say that X is a Calabi–Yau threefold if its canonical bundle K_X is trivial. Then $-K_X \cdot \beta = 0$ for any β . For a Calabi–Yau threefold X, the genus g, degree β Gromov-Witten invariant of

X is defined as

$$N_{g,\beta}(X) := \langle 1 \rangle_{\beta,0}^X = \int_{[\overline{\mathcal{M}}_{g,0}(X,\beta)]^{\mathrm{vir}}} 1 \in \mathbb{Q}.$$
 (2.6)

We consider the following generating function of Gromov–Witten invariants $N_{g,\beta}(X)$ of X in all genera and all nonzero degrees:

$$F_X(\lambda, Q) = \sum_{\beta \neq 0} \sum_{g=0}^{\infty} N_{g,\beta}(X) \lambda^{2g-2} Q^{\beta}.$$
(2.7)

 $F_X(\lambda, Q)$ is called the free energy of X and its exponential

$$Z_X(\lambda, Q) := \exp F_X(\lambda, Q), \qquad (2.8)$$

is called the partition function of X.

From considerations in M-theory, Gopakumar–Vafa [29] proposed a remarkable conjecture about integrality structures of the free energy of a Calabi–Yau threefold X.

Definition 2.2. Define the Gopakumar–Vafa invariants $n_{\beta}^{g}(X) \in \mathbb{Q}$ by the following formula:

$$F_X(\lambda, Q) = \sum_{\beta \neq 0} \sum_{g=0}^{\infty} \sum_{k=1}^{\infty} \frac{n_{\beta}^g(X)}{k} \left(2\sin\frac{k\lambda}{2}\right)^{2g-2} Q^{k\beta}.$$
(2.9)

Matching the coefficients of the two series yields equations determining $n_{\beta}^{g}(X)$ recursively in terms of $N_{g,\beta}(X)$.

Conjecture 2.3 (Gopakumar–Vafa conjecture). For a Calabi–Yau threefold X, $n_{\beta}^{g}(X) \in \mathbb{Z}$ for any $g \geq 0$ and $\beta \in H_{2}(X,\mathbb{Z})$. Moreover for any fixed β , $n_{\beta}^{g}(X) = 0$ for all but finite g.

Gopakumar–Vafa's original argument was that for each curve class $\beta \in H_2(X, \mathbb{Z})$ and genus g, there is an integer $n^g_\beta(X)$ counting certain BPS states in M-theory and the formula (2.9) holds for a Calabi–Yau threefold X. According to their physical argument, the number of BPS states should be defined in terms of the moduli space of D-branes on X. See [33, 43] for approaches for this problem.

2.2 Virtual localization

The virtual localization formula (Theorem 2.5) due to Kontsevich [52] and Graber–Pandharipande [30] is presented. We shall use it in §3.2.2 and §5.

2.2.1 Equivariant homology groups

First, we review equivariant homology groups. We work with Q-coefficients. Let X be a nonsingular irreducible algebraic variety equipped with an action of a linear algebraic group G (over \mathbb{C}). The G-equivariant homology groups of X are defined by the so-called Borel construction. Consider the universal principal bundle $EG \to BG$ of G. Here EG is a contractible space with a free G-action (these conditions determine EG uniquely up to homotopy) and BG = EG/G. Consider $EG \times X$ on which G acts freely and the quotient $X_G = (EG \times X)/G$, the homotopy quotient of X by G. Roughly speaking, the G-equivariant (co)homology groups are defined via ordinary (co)homology groups of X_G .

More precisely, we work with finite-dimensional approximations of the classifying space $EG \rightarrow BG$. For each $N \geq 1$, there exists a nonsingular irreducible variety EG_N with G-action which has the following properties:

(i) $H^i(EG_N) = 0$ for i = 1, ..., 2N.

(ii) A principal G-bundle quotient

$$EG_N \to BG_N := EG_N/G$$

exists.

Let $X_{G,N} := (EG_N \times X)/G$. For each *n*, we define the *n*-th *G*-equivariant homology group of X as follows. Take N such that n < 2N and define

$$H_n^G(X) := H_{n-2\dim G+2\dim EG_N}^{BM}(X_{G,N}),$$

where the RHS is the Borel–Moore homology group (see e.g. $[26, \S19.1]$). We also define the *n*-th equivariant cohomology group of X as

$$H^n_G(X) := H^n(X_{G,N}),$$

where the RHS is the ordinary cohomology group. These are independent of the choice of EG_N . If $n > 2 \dim X$, then $H_n^G(X) = 0$. On the other hand, $H_n^G(X)$ may be non-zero for n < 0. By the Poincaré duality, we have

$$H^n_G(X) \cong H^G_{2\dim X-n}(X).$$

The projection map $p: X_{G,N} \to BG_N$ is a flat map with the fiber X, which makes $H^*_G(X)$ into a module over the graded ring $H^*(BG) = H^*_G(pt)$. The inclusion *i* of a fiber of *p* gives a morphism

$$i^*: H^G_*(X) \to H_*(X)$$

via the Poincaré duality isomorphism. The top dimensional map

$$i^*: H^G_{2\dim X}(X) \to H_{2\dim X}(X),$$

is an isomorphism. The G-equivariant fundamental class $[X]_G$ is defined by $i^*[X]_G = [X]$, where [X] is the usual fundamental class of X.

Let $V \to X$ be a *G*-equivariant vector bundle. Consider $V_G := (EG \times V)/G$. Then $V_G \to X_G$ is a vector bundle. Define the *G*-equivariant Euler class $e_G(V)$ of *V* as

$$e_G(V) := e(V_G) \in H_G^{2\operatorname{rk} V}(X),$$

where the RHS is the ordinary Euler class (i.e. the Chern class of the top degree).

2.2.2 Classical localization theorem

In this section, we consider an action of an algebraic torus $\mathbb{T} = (\mathbb{C}^*)^r$ on a nonsingular irreducible algebraic variety X. First, we give a concrete description of $EG_N \to BG_N$ for $G = \mathbb{T} = (\mathbb{C}^*)^r$. Let us consider

$$E\mathbb{T}_N := (\mathbb{C}^{N+1} \setminus \{0\}) \times \cdots \times (\mathbb{C}^{N+1} \setminus \{0\}) \ (r \text{ times}),$$

equipped with the T-action defined by

$$(t_1, \ldots, t_r) \cdot (v_1, \ldots, v_r) = (t_1^{-1}v_1, \ldots, t_r^{-1}v_r).$$

Then $E\mathbb{T}_N$ satisfies the conditions (i) and (ii) above and

$$B\mathbb{T}_N = \mathbb{P}^N \times \cdots \times \mathbb{P}^N$$
 (*r* times)

In the limit $N \to \infty$, the map $E\mathbb{T} \to B\mathbb{T}$ is given by

$$(\mathbb{C}^{\infty} \setminus \{0\})^r \to (\mathbb{P}^{\infty})^r.$$

We have $H^*_{\mathbb{T}}(pt) = H^*(B\mathbb{T}) \cong \mathbb{Q}[\alpha_1, \ldots, \alpha_r]$, where α_i is given by $c_1(\mathcal{O}(1))$ of the *i*-th factor, hence deg $\alpha_i = 2$. We denote this graded ring by $R_{\mathbb{T}}$.

Let $\operatorname{Ch}(\mathbb{T})$ be the character group of \mathbb{T} . For each $\rho \in \operatorname{Ch}(\mathbb{T})$, we have a 1-dimensional representation \mathbb{C}_{ρ} of \mathbb{T} , hence the corresponding line bundle $L_{\rho} := (\mathbb{C}_{\rho})_{\mathbb{T}}$ over $B\mathbb{T}$. We have a homomorphism $w : \operatorname{Ch}(\mathbb{T}) \to H^2(B\mathbb{T})$ defined by $\rho \mapsto -c_1(L_{\rho})$. For $\rho \in \operatorname{Ch}(\mathbb{T})$, we call $w(\rho)$ the weight of ρ . Note that if $\rho_i \in \operatorname{Ch}(\mathbb{T})$ is the character defined by $\rho_i(t_1, \ldots, t_r) = t_i$ for $(t_1, \ldots, t_r) \in \mathbb{T}$, then we have $w(\rho_i) = \alpha_i$.

Let $X^{\mathbb{T}}$ be the \mathbb{T} -fixed point set and $\iota: X^{\mathbb{T}} \hookrightarrow X$ be the inclusion. We have the push-forward morphism

$$\iota_*: H^{\mathbb{T}}_*(X^{\mathbb{T}}) \to H^{\mathbb{T}}_*(X).$$

Let $R_{\mathbb{T}}^+$ be the multiplicative closed subset of $R_{\mathbb{T}}$ which consists of all the homogeneous elements of positive degree and the unity. Let $Q_{\mathbb{T}} := (R_{\mathbb{T}}^+)^{-1} R_{\mathbb{T}} \subset \mathbb{Q}(\alpha_1, \ldots, \alpha_r)$ be the localized ring. Then the localization theorem of Atiyah–Bott [3] states that

$$\iota_*: H^{\mathbb{T}}_*(X^{\mathbb{T}}) \otimes Q_{\mathbb{T}} \to H^{\mathbb{T}}_*(X) \otimes Q_{\mathbb{T}}$$

is an isomorphism.

Next, we give explicit localization formula for the equivariant fundamental class. Let $X^{\mathbb{T}} = \bigcup X_i$ be the decomposition into irreducible components. By [39], smoothness of X implies that of X_i . Let N_i be the normal bundle to X_i in X. Since \mathbb{T} acts on N_i with non-zero weights, the \mathbb{T} -equivariant Euler class $e_{\mathbb{T}}(N_i)$ is invertible in $H^*_{\mathbb{T}}(X_i) \otimes Q_{\mathbb{T}}$ (see e.g.[21, Proposition 4]). The localization formula for the \mathbb{T} -equivariant fundamental class $[X]_{\mathbb{T}}$ is given by

$$[X]_{\mathbb{T}} = \iota_* \sum_{i} \frac{[X_i]}{e_{\mathbb{T}}(N_i)} \in H^{\mathbb{T}}_*(X) \otimes Q_{\mathbb{T}} .$$

$$(2.10)$$

This formula enables us to express integrals over X as a sum of contributions from \mathbb{T} -fixed point components. For example, let A be a \mathbb{T} -equivariant vector bundle whose rank is equal to dim X and A_i be the pull-back of A to X_i . Then we have

$$\int_{X} e(A) = \sum_{i} \int_{X_i} \frac{e_{\mathbb{T}}(A_i)}{e_{\mathbb{T}}(N_i)},$$
(2.11)

which is called Bott's residue formula [9].

There is an algebraic version $A^{\mathbb{T}}_*(X)$ of the equivariant homology group, the equivariant Chow group of X with Q-coefficients defined by Edidin–Graham [20]. A construction of the equivariant Chow groups similar to that in §2.2.1 has been given in *loc.cit*.. Moreover, in [21], it has been shown that the localization theorem and the formula (2.10) hold in the localized equivariant Chow group $A^{\mathbb{T}}_*(X) \otimes Q_{\mathbb{T}}$. See [53] for an extension of these results to Deligne– Mumford stacks.

2.2.3 Virtual localization formula

Let X be a non-singular projective variety equipped with an action of an algebraic torus \mathbb{T} . A \mathbb{T} -action on X induces that on $\overline{\mathcal{M}} := \overline{\mathcal{M}}_{g,n}(X,\beta)$ by translating maps:

$$t \cdot [f] = [t \cdot f], \quad t \in \mathbb{T}, \ [f] \in \overline{\mathcal{M}}.$$

It was Kontsevich [52] who proposed to compute Gromov–Witten invariants of such X by using localization formula with respect to the above \mathbb{T} -action on $\overline{\mathcal{M}}$. However, integrals in Gromov– Witten theory are taken against the virtual fundamental class $[\overline{\mathcal{M}}]^{vir}$ of $\overline{\mathcal{M}}$. Therefore, we need a localization formula for the virtual fundamental class. This was established by Graber– Pandharipande [30]. Here we summarize their results.

Recall that we have the perfect obstruction theory $E_{\bullet} = [E_0 \to E_1]$ which is a two-term complex of vector bundles on $\overline{\mathcal{M}}$. It defines the virtual fundamental class $[\overline{\mathcal{M}}]^{vir}$ of $\overline{\mathcal{M}}$ (cf.§2.1.2, 2.1.3). Now if there is a T-action on $\overline{\mathcal{M}}$ we can take the complex E_{\bullet} so that it is T-equivariant. Then it defines the equivariant virtual fundamental class $[\overline{\mathcal{M}}]^{vir}_{\mathbb{T}}$ in the equivariant Chow group $A^{\mathbb{T}}_{vdim\overline{\mathcal{M}}}(\overline{\mathcal{M}})$ of expected dimension.

Let $\overline{\mathcal{M}}^{\mathbb{T}} = \bigcup_i \overline{\mathcal{M}}_i$ be the fixed point components. Let $E_{\bullet,i}$ be the restriction of E_{\bullet} to $\overline{\mathcal{M}}_i$. The complex $E_{\bullet,i}$ may be decomposed by \mathbb{T} -characters:

$$E_{\bullet,i} = E^f_{\bullet,i} \oplus E^m_{\bullet,i},$$

where the first summand is the \mathbb{T} -fixed part corresponding to the trivial character and the second summand is the \mathbb{T} -moving part corresponding to non-trivial characters. The \mathbb{T} -fixed part $E_{\bullet,i}^f$ defines the virtual fundamental class $[\overline{\mathcal{M}}_i]^{vir}$ of $\overline{\mathcal{M}}_i$. The normal 'bundle' to $\overline{\mathcal{M}}_i$ is defined as follows.

Definition 2.4. The virtual normal bundle N_i^{vir} to $\overline{\mathcal{M}}_i$ is defined to be the T-moving part $E_{\bullet,i}^m$. The equivariant Euler class of $N_i^{vir} = E_{0,i}^m \to E_{1,i}^m$ is defined to be

$$e_{\mathbb{T}}(N_i^{vir}) = \frac{e_{\mathbb{T}}(E_{0,i}^m)}{e_{\mathbb{T}}(E_{1,i}^m)}.$$
(2.12)

Then the virtual localization formula is stated as follows.

Theorem 2.5 (Graber–Pandharipande).

$$[\overline{\mathcal{M}}]^{vir}_{\mathbb{T}} = \iota_* \sum_{i} \frac{[\overline{\mathcal{M}}_i]^{vir}}{e_{\mathbb{T}}(N_i^{vir})} \in A^{\mathbb{T}}_{v\dim(\overline{\mathcal{M}})}(\overline{\mathcal{M}}) \otimes Q_{\mathbb{T}}.$$
(2.13)

Bott's formula is also valid in this setting;

$$\int_{[\overline{\mathcal{M}}]^{vir}} e(A) = \sum_{i} \int_{[\overline{\mathcal{M}}_{i}]^{vir}} \frac{e_{\mathbb{T}}(A_{i})}{e_{\mathbb{T}}(N_{i}^{vir})}.$$
(2.14)

2.3 Hodge integrals

If we consider stable maps to a point, we recover the notion of stable curves introduced in [13, 45]. We denote by $\overline{\mathcal{M}}_{g,n}$ the moduli space of stable curves of genus g with n marked points. It is an irreducible projective variety of dimension 3g - 3 + n. Since there are no obstructions to deformations of curves, it follows that the virtual fundamental class of $\overline{\mathcal{M}}_{g,n}$ is the usual fundamental class. Hodge integrals are integrals of natural cohomology classes over $\overline{\mathcal{M}}_{g,n}$. They arise naturally in Gromov–Witten theory. See [24] for more information. The main result in this section is a formula for two-partition Hodge integrals (Theorem 2.6) due to Liu–Liu–Zhou [60], which will be used in §5.

2.3.1 Partitions

First, we introduce some notations related to partitions, which will be used throughout the paper. Let μ be a partition, i.e. a non-increasing sequence of positive integers $\mu = (\mu_1 \ge \mu_2 \ge \cdots \ge \mu_{l(\mu)} > 0)$. The number $l(\mu)$ is called the length of μ and $|\mu| := \mu_1 + \cdots + \mu_{l(\mu)}$ is called the weight of μ . The automorphism group $\operatorname{Aut}(\mu)$ is the subgroup of the permutation group of $\{\mu_1, \ldots, \mu_{l(\mu)}\}$ which preserves μ , i.e. permutes μ_i and μ_j only if $\mu_i = \mu_j$. Its order is equal to $\prod_{k\ge 1} m_k(\mu)!$, where $m_k(\mu) := \#\{i \mid \mu_i = k\}$.

We denote by \mathcal{P} the set of partitions. For $\mu \in \mathcal{P}$, let χ_{μ} denote the character of the irreducible representation of symmetric group $S_{|\mu|}$ indexed by μ , and let C_{μ} denote the conjugacy class of $S_{|\mu|}$ indexed by μ . By the orthogonality of characters of S_d , where $d = |\mu| = |\nu|$, we have

$$\sum_{\substack{\sigma \in \mathcal{P} \\ |\sigma|=d}} \frac{\chi_{\mu}(C_{\sigma})\chi_{\nu}(C_{\sigma})}{z_{\sigma}} = \delta_{\mu\nu}, \qquad (2.15)$$

where

$$z_{\mu} = |\operatorname{Aut}(\mu)| \cdot \prod_{i=1}^{l(\mu)} \mu_i .$$

For $\mu \in \mathcal{P}$, we define the integer

$$\kappa(\mu) := |\mu| + \sum_{i=1}^{l(\mu)} \mu_i(\mu_i - 2i), \qquad (2.16)$$

which is always even.

Let $x = (x_1, x_2, ...)$ be a set of variables. For $\mu \in \mathcal{P}$, let $s_{\mu}(x)$ be the Schur function associated to μ . See [62] for a definition. It is known that the set of Schur functions $\{s_{\mu}(x) \mid \mu \in \mathcal{P}\}$ is a \mathbb{Z} -basis of the ring of symmetric functions. The skew Schur function $s_{\mu/\lambda}(x)$ for $\mu, \lambda \in \mathcal{P}$ is defined in terms of Schur functions as follows:

$$s_{\mu/\lambda}(x) = \sum_{\nu} c^{\mu}_{\lambda\nu} s_{\nu}(x)$$

where $c^{\mu}_{\lambda\nu}$ are integers (called the *Littlewood-Richardson coefficients*) defined by

$$s_{\lambda}(x)s_{\nu}(x) = \sum_{\mu} c^{\mu}_{\lambda\nu}s_{\mu}(x) \; .$$

2.3.2 Hodge integrals

There are natural cohomology classes on $\overline{\mathcal{M}}_{g,n}$ called the λ and ψ classes. Let $\pi : \overline{\mathcal{M}}_{g,n+1} \to \overline{\mathcal{M}}_{g,n}$ be the universal curve and ω_{π} be the relative dualizing sheaf. The Hodge bundle $\mathbb{E} := \pi_* \omega_{\pi}$ is the rank g vector bundle over $\overline{\mathcal{M}}_{g,n}$ whose fiber at (C, p_1, \ldots, p_n) is given by $H^0(C, \omega_C)$. The λ -classes are the Chern classes of \mathbb{E} :

$$\lambda_i := c_i(\mathbb{E}) \in H^{2i}(\overline{\mathcal{M}}_{q,n}, \mathbb{Q}).$$

Let $s_i : \overline{\mathcal{M}}_{g,n} \to \overline{\mathcal{M}}_{g,n+1}$ be the section of π which corresponds to the *i*-th marked point. Let $\mathbb{L}_i := s_i^* \omega_{\pi}$ be the line bundle over $\overline{\mathcal{M}}_{g,n}$ whose fiber at (C, p_1, \ldots, p_n) is the cotangent line $T_{p_i}^* C$. The ψ -classes are the first Chern classes of \mathbb{L}_i :

$$\psi_i := c_1(\mathbb{L}_i) \in H^2(\overline{\mathcal{M}}_{g,n}, \mathbb{Q}).$$

The Hodge integrals are intersection numbers of ψ -classes and λ -classes:

$$\int_{\overline{\mathcal{M}}_{g,n}} \psi_1^{j_1} \cdots \psi_n^{j_n} \lambda_1^{k_1} \cdots \lambda_g^{k_g} \in \mathbb{Q}.$$

In Gromov–Witten theory of threefolds, Hodge integrals of particular types often appear. For $h \ (1 \le h \le 3)$, let (μ^1, \dots, μ^h) be an *h*-tuple of partitions. Let us consider the integrals of the following type:

$$\int_{\overline{\mathcal{M}}_{g,l(\mu^{1})+\dots+l(\mu^{h})}} \frac{\Lambda_{g}^{\vee}(u_{1})\Lambda_{g}^{\vee}(u_{2})\Lambda_{g}^{\vee}(u_{3})}{\prod_{j=0}^{h-1}\prod_{i=1}^{l(\mu^{j+1})}(1-\mu_{i}^{j+1}\psi_{l(\mu^{j})+i})}$$

where we set $\mu^0 = \emptyset$ (the empty partition) and

$$\Lambda_g^{\vee}(u) := u^g - \lambda_1 u^{g-1} + \dots + (-1)^g \lambda_g.$$

By definition, the integral is the combination of Hodge integrals obtained by expanding the integrand as a series in the ψ and λ classes and then integrating the terms of the appropriate cohomological degree. They are often referred to as one-, two-, three-partition triple Hodge integrals for h = 1, 2, 3 respectively. If the parameters in the numerator satisfies $u_1+u_2+u_3=0$, these are called special (or Calabi-Yau) triple Hodge integrals. For example, the integrals that arise in the computation of the Gromov–Witten invariants of a general toric Calabi–Yau threefold are three-partition special triple Hodge integrals.

Let us consider the simplest cases, one-partition Hodge integrals for g = 0. In this case $\Lambda_0^{\vee}(u) = 1$ and

$$\int_{\overline{\mathcal{M}}_{0,l(\mu)}} \frac{1}{\prod_{i=1}^{l(\mu)} (1-\mu_i \psi_i)} = |\mu|^{l(\mu)-3},$$
(2.17)

for $l(\mu) \geq 3$ (see e.g.[52, Lemma in §3.3.2]). We use the equation (2.17) to extend the definition to the case of $l(\mu) < 3$. In that case, $\overline{\mathcal{M}}_{0,l(\mu)}$ is formally defined to be a point.

2.3.3 Formula for two-partition Hodge integrals

We shall encounter the following two-partition special triple Hodge integrals in §5, where we will study Gromov–Witten invariants of local toric surfaces. Given $\mu^+, \mu^- \in \mathcal{P}$, define

$$G_{g,\mu^{+},\mu^{-}}(\tau) = \frac{-\sqrt{-1}^{l(\mu^{+})+l(\mu^{-})}}{|\operatorname{Aut}(\mu^{+})||\operatorname{Aut}(\mu^{-})|} (\tau(\tau+1))^{l(\mu^{+})+l(\mu^{-})-1} \cdot \prod_{i=1}^{l(\mu^{+})} \frac{\prod_{a=1}^{\mu_{i}^{+}-1}(\mu_{i}^{+}\tau+a)}{(\mu_{i}^{+}-1)!} \prod_{j=1}^{l(\mu^{-})} \frac{\prod_{a=1}^{\mu_{j}^{-}-1}(\frac{\mu_{j}^{-}}{\tau}+a)}{(\mu_{j}^{-}-1)!} \cdot \int_{\overline{\mathcal{M}}_{g,l(\mu^{+})+l(\mu^{-})}} \frac{\Lambda_{g}^{\vee}(1)\Lambda_{g}^{\vee}(\tau)\Lambda_{g}^{\vee}(-\tau-1)}{\prod_{i=1}^{l(\mu^{+})}(1-\mu_{i}^{+}\psi_{i})\prod_{j=1}^{l(\mu^{-})}\tau(\tau-\mu_{j}^{-}\psi_{l(\mu^{+})+j})} .$$
(2.18)

We give a formula for $G_{g,\mu^+,\mu^-}(\tau)$ which will be used in §5. To state the formula, we introduce some generating functions.

We first define generating function of two-partition Hodge integrals. Introduce variables

$$p^+ = (p_1^+, p_2^+, \ldots), \quad p^- = (p_1^-, p_2^-, \ldots)$$

Given a partition μ , define

$$p_{\mu}^{\pm} = p_{\mu_1}^{\pm} \cdots p_{\mu_{l(\mu)}}^{\pm}$$

Define generating functions

$$G_{\mu^+,\mu^-}(\lambda;\tau) = \sum_{g=0}^{\infty} \lambda^{2g-2+l(\mu^+)+l(\mu^-)} G_{g,\mu^+,\mu^-}(\tau) , \qquad (2.19)$$

$$G(\lambda; p^{+}, p^{-}; \tau) = \sum_{\substack{(\mu^{+}, \mu^{-}) \neq (\emptyset, \emptyset)}} G_{\mu^{+}, \mu^{-}}(\lambda; \tau) p_{\mu^{+}}^{+} p_{\mu^{-}}^{-} ,$$

$$G^{\bullet}(\lambda; p^{+}, p^{-}; \tau) = \exp(G(\lambda; p^{+}, p^{-}; \tau))$$

$$= \sum_{\mu^{+}, \mu^{-}} G_{\mu^{+}, \mu^{-}}^{\bullet}(\lambda; \tau) p_{\mu^{+}}^{+} p_{\mu^{-}}^{-} .$$
(2.20)

 $G^{ullet}_{\mu^+,\mu^-}(\lambda;\tau)$ is the "disconnected" ⁶ version of $G_{\mu^+,\mu^-}(\lambda;\tau)$.

We next define generating functions of of symmetric group representations. Let $q = e^{\sqrt{-1}\lambda}$ and write

$$q^{-\rho} = (q^{\frac{1}{2}}, q^{\frac{3}{2}}, \ldots).$$

Introduce

$$W_{\mu,\nu}(q) := (-1)^{|\mu| + |\nu|} q^{\frac{\kappa(\mu) + \kappa(\nu)}{2}} \sum_{\eta \in \mathcal{P}} s_{\mu/\eta}(q^{-\rho}) s_{\nu/\eta}(q^{-\rho}) , \qquad (2.21)$$

and define

$$R^{\bullet}_{\mu^{+},\mu^{-}}(\lambda;\tau) = \sum_{\substack{\nu^{+},\nu^{-}\\ |\nu^{\pm}| = |\mu^{\pm}|}} \frac{\chi_{\nu^{+}}(C_{\mu^{+}})}{z_{\mu^{+}}} \frac{\chi_{\nu^{-}}(C_{\mu^{-}})}{z_{\mu^{-}}} e^{\sqrt{-1}(\kappa(\nu^{+})\tau + \kappa(\nu^{-})\tau^{-1})\lambda/2} W_{\nu^{+},\nu^{-}}(q) .$$
(2.22)

The following identity was conjectured by Zhou [81] and proved by Liu–Liu–Zhou [60].

Theorem 2.6 (Liu–Liu–Zhou).

$$G^{\bullet}_{\mu^{+},\mu^{-}}(\lambda;\tau) = R^{\bullet}_{\mu^{+},\mu^{-}}(\lambda;\tau).$$
(2.23)

See [61] and the references therein for various aspects of the formula and related works. A formula for three-partition special triple Hodge integrals can be found in [59].

Part II Topological vertex and local Gromov–Witten invariants

3 Local Gromov–Witten invariants of surfaces

In §3.1, local Gromov–Witten invariants of algebraic surfaces are introduced. We discuss the case of -K-nef surfaces in detail in §3.2.

⁶See [71] for a meaning and discussions of disconnected version.

3.1 Local Gromov–Witten invariants of surfaces

Let S be a non-singular projective surface and K_S be its canonical bundle. Let

be the universal curve and the universal map over the moduli space of stable maps to S (cf.(2.1)).

Lemma 3.1. There exists a two term complex $[K_0 \to K_1]$ of vector bundles on $\overline{\mathcal{M}}_{g,0}(S,\beta)$ whose cohomology sheaves are $R^0\pi_*\mu^*K_S$ and $R^1\pi_*\mu^*K_S$.

Proof. See [7, Proposition 5].

Note that the difference $[K_1] - [K_0]$ as an element of the Grothendieck group of vector bundles on $\overline{\mathcal{M}}_{q,0}(S,\beta)$ is independent of a choice of the complex.

Definition 3.2. (i) We define

$$c(-R^{\bullet}\pi_{*}\mu^{*}K_{S}) := c([K_{1}] - [K_{0}]) = \frac{c(K_{1})}{c(K_{0})}$$

where $c(\cdot)$ is the total Chern class defined on the Grothendieck group of vector bundles (cf. [26, Example 3.2.7.]).

(ii) For $g \ge 0$ and $\beta \in H_2(S, \mathbb{Z})$, the local Gromov-Witten invariant $I_{g,\beta}(S)$ of S with the genus g and the degree β is defined by

$$I_{g,\beta}(S) := \int_{[\overline{\mathcal{M}}_{g,0}(S,\beta)]^{vir}} c(-R^{\bullet}\pi_*\mu^*K_S) \in \mathbb{Q}.$$

A geometrical meaning of local Gromov–Witten invariants can be explained as follows. The total space of the bundle $K_S \to S$ can be regarded as a neighborhood of S smoothly embedded into a Calabi–Yau threefold X. Under some assumptions, the moduli space of stable maps $\overline{\mathcal{M}}_{g,0}(X,\iota_*\beta)$ (where $\iota: S \hookrightarrow X$ is the given imbedding) has a union of connected components which is isomorphic to $\overline{\mathcal{M}}_{g,0}(S,\beta)$. Ideally, local Gromov–Witten invariants can be seen as local contributions to Gromov–Witten invariants of the ambient space X from maps into S. See [74] for details.

Remark 3.3. The following statement seems to be well-known to specialists: the local Gromov–Witten invariants $I_{g,\beta}(S)$ is invariant under smooth deformations of S in the sense that is explained in the last paragraph in §2.1.3. However, because of the lack of appropriate references, we shall prove this statement under some assumptions in Proposition 3.7.

3.2 The case of -K-nef surfaces

3.2.1 -K-nef surfaces

Let S be a non-singular projective surface whose anti-canonical bundle $-K_S$ is nef, i.e. $-K_S \cdot [C] \ge 0$ for any irreducible curve C. We call such a surface a -K-nef surface.

Lemma 3.4. Let S be a -K-nef surface. Then for any $g \in \mathbb{Z}_{\geq 0}$ and $\beta \in H_2(X,\mathbb{Z})$ such that $\int_{\beta} c_1(K_S) < 0$, we have

$$R^0 \pi_* \mu^* K_S = 0.$$

Proof. Let $(C, f) \in \overline{\mathcal{M}}_{g,0}(S, \beta)$. By the -K-nef condition, f^*K_S has at most constant sections on each irreducible component of C. On the other hand, the negativity assumption $\int_{\beta} c_1(K_S) < 0$ implies that there exists an irreducible component of C on which the degree of f^*K_S is negative. Therefore, sections of f^*K_S on C must be zero. \Box

By Lemma 3.4, $R^1 \pi_* \mu^* K_S$ forms a vector bundle over $\overline{\mathcal{M}}_{g,0}(S,\beta)$ of rank $(1-g)(\dim S-3) - \int_{\beta} c_1(K_S) = \operatorname{vdim} \overline{\mathcal{M}}_{g,0}(S,\beta)$. We have

$$I_{g,\beta}(S) = \int_{[\overline{\mathcal{M}}_{g,0}(S,\beta)]^{vir}} e(R^1 \pi_* \mu^* K_S).$$
(3.2)

3.2.2 Comparison with projective bundles

Let S be a non-singular projective surface and $\mathbb{P}(K_S \oplus \mathcal{O}_S)$ be the projectivization of the total space of the vector bundle $K_S \oplus \mathcal{O}_S$. This is a \mathbb{P}^1 -bundle over S. Let $\iota : S \hookrightarrow \mathbb{P}(K_S \oplus \mathcal{O}_S)$ be the inclusion as the zero section of $K_S \subset \mathbb{P}(K_S \oplus \mathcal{O}_S)$. Although $\mathbb{P}(K_S \oplus \mathcal{O}_S)$ is not a Calabi–Yau threefold, it is *locally* Calabi–Yau near the zero section. We define

$$N_{g,\iota_*\beta}(\mathbb{P}(K_S \oplus \mathcal{O}_S)) = \int_{[\overline{\mathcal{M}}_{g,0}(\mathbb{P}(K_S \oplus \mathcal{O}_S),\iota_*\beta)]^{vir}} 1 ,$$

since vdim $\overline{\mathcal{M}}_{g,0}(\mathbb{P}(K_S \oplus \mathcal{O}_S), \iota_*\beta) = 0.$

Proposition 3.5. Let S be a -K-nef surface, $\iota : S \hookrightarrow \mathbb{P}(K_S \oplus \mathcal{O}_S)$ be the inclusion as the zero section of K_S . Then, for $g \in \mathbb{Z}_{\geq 0}$ and $\beta \in H_2(S, \mathbb{Z})$ such that $\int_{\beta} c_1(K_S) < 0$, we have

$$I_{g,\beta}(S) = N_{g,\iota_*\beta}(\mathbb{P}(K_S \oplus \mathcal{O}_S))$$

Consider the natural \mathbb{C}^* -action on $\mathbb{P}(K_S \oplus \mathcal{O}_S)$ as the scalar multiplication in the \mathbb{P}^1 -fiber direction. There is an induced \mathbb{C}^* -action on $\overline{\mathcal{M}}_{g,0}(\mathbb{P}(K_S \oplus \mathcal{O}_S), \iota_*\beta)$. First we show the following

Lemma 3.6. Let S be a -K-nef surface, $\iota : S \hookrightarrow \mathbb{P}(K_S \oplus \mathcal{O}_S)$ be the inclusion as the zero section of K_S . Let $\beta \in H_2(S,\mathbb{Z})$ be a class satisfying $\int_{\beta} c_1(K_S) < 0$. If a stable map $(f, C) \in \overline{\mathcal{M}}_{g,0}(\mathbb{P}(K_S \oplus \mathcal{O}_S), \iota_*\beta)$, where C is a connected curve of genus g and $f : C \to S$ a morphism such that $[f(C)] = \iota_*\beta$, is fixed by the \mathbb{C}^* -action, then the image f(C) is contained in the zero section $\iota(S)$.

Proof. Denote the \mathbb{P}^1 -fibration $\mathbb{P}(K_S \oplus \mathcal{O}_S) \to S$ by p, and let $P = [p^{-1}(a)] \in H_2(\mathbb{P}(K_S \oplus \mathcal{O}_S), \mathbb{Z})$ be the class of the fiber \mathbb{P}^1 where $a \in S$ is any point. Let $\iota^{\infty} : S \hookrightarrow \mathbb{P}(K_S \oplus \mathcal{O}_S)$ be the inclusion as the zero section of \mathcal{O}_S (the section at the infinity of the \mathbb{P}^1 -bundle). Note that for any $\alpha \in H_2(S, \mathbb{Z})$, we have

$$\iota_*^{\infty} \alpha = \iota_* \alpha - \left(\int_{\alpha} c_1(K_S) \right) P \quad . \tag{3.3}$$

Let $\gamma \in H_2(\mathbb{P}(K_S \oplus \mathcal{O}_S), \mathbb{Z})$. If a stable map $(f, C) \in \overline{\mathcal{M}}_{g,0}(\mathbb{P}(K_S \oplus \mathcal{O}_S), \gamma)$ is fixed by the \mathbb{C}^* -action, then the image of an irreducible component C_i of C must be either one of these:

- (i) $f(C_i) \subset \iota(S)$,
- (ii) $f(C_i) \subset \iota^{\infty}(S)$,
- (iii) $f(C_i) = p^{-1}(a_i) \ (a_i \in S) \text{ and } C_i \cong \mathbb{P}^1.$

So assume that irreducible components C_1, \ldots, C_k of C are of type (i) with $[f(C_i)] = \beta_i \in$

 $H_2(S,\mathbb{Z}), C_{k+1}, \ldots, C_r$ are of type (ii) with $[f(C_i)] = \beta_i \in H_2(S,\mathbb{Z})$, and that C_{r+1}, \ldots, C_s are of type (iii) with $f: C_i \to p^{-1}(a_i)$ the d_i -fold coverings. Then $[f(C)] = \gamma$ is equivalent to

$$\gamma = \sum_{i=1}^{k} \iota_* \beta_i + \sum_{i=k+1}^{r} \iota_*^{\infty} \beta_i + \sum_{i=r+1}^{s} d_i P = \sum_{i=1}^{r} \iota_* \beta_i + \left(\sum_{i=r+1}^{s} d_i - \sum_{i=k+1}^{r} \int_{\beta_i} c_1(K_S)\right) P$$

Now take $\gamma = \iota_*\beta$ with $\beta \in H_2(S, \mathbb{Z})$ satisfying $\int_\beta c_1(K_S) < 0$ and solve the above equation. The assumption that S is -K-nef implies that the coefficient of P in the last line is always nonnegative. Therefore it is zero if and only if there is no irreducible components of type (iii) and $\int_{\beta_i} c_1(K_S) = 0$ for those of type (ii). Then connectedness of the domain curve C implies either $f(C) \subset \iota(S)$ or $f(C) \subset \iota^{\infty}(S)$. For the latter case, $\int_{[f(C)]} c_1(K_S) = 0$ and this contradicts the assumption $\int_\beta c_1(K_S) < 0$. Thus $f(C) \subset \iota(S)$.

Proof. (of Proposition 3.5.) By Lemma 3.6, the \mathbb{C}^* -fixed point set in $\overline{\mathcal{M}}_{g,0}(\mathbb{P}(K_S \oplus \mathcal{O}_S), \iota_*\beta)$ is isomorphic to $\overline{\mathcal{M}}_{g,0}(S,\beta)$. We compare perfect obstruction theories on them. Let $E_{\bullet} = [E_0 \to E_1]$ be the perfect obstruction theory on $\overline{\mathcal{M}}_{g,0}(\mathbb{P}(K_S \oplus \mathcal{O}_S), \iota_*\beta)$ restricted to its \mathbb{C}^* -fixed point set. By exact sequences (2.3, 2.4), we have

$$E_{1} - E_{0} = R^{1} \pi_{*} \underline{Hom}(\Omega_{\pi}, \mathcal{O}_{\mathcal{U}}) - R^{0} \pi_{*} \underline{Hom}(\Omega_{\pi}, \mathcal{O}_{\mathcal{U}}) + R^{1} \pi_{*} \mu^{*} T_{X} - R^{0} \pi_{*} \mu^{*} T_{X}$$

$$= R^{1} \pi_{*} \underline{Hom}(\Omega_{\pi}, \mathcal{O}_{\mathcal{U}}) - R^{0} \pi_{*} \underline{Hom}(\Omega_{\pi}, \mathcal{O}_{\mathcal{U}}) + R^{1} \pi_{*} \mu^{*} T_{S} - R^{0} \pi_{*} \mu^{*} T_{S}$$

$$+ R^{1} \pi_{*} \mu^{*} K_{S} - R^{0} \pi_{*} \mu^{*} K_{S}.$$
(3.4)

Here we have used the normal bundle sequence at the zero-section

$$0 \longrightarrow T_S \longrightarrow T_{\mathbb{P}(K_S \oplus \mathcal{O}_S)}|_S \longrightarrow K_S \longrightarrow 0,$$

in the second equality. The first four terms in the second line of (3.4) are the \mathbb{C}^* -fixed part of E_{\bullet} which are identical to the perfect obstruction theory on $\overline{\mathcal{M}}_{g,0}(S,\beta)$. Hence we have

$$[\overline{\mathcal{M}}_{g,0}(\mathbb{P}(K_S \oplus \mathcal{O}_S), \iota_*\beta)^{\mathbb{C}^*}]^{vir} \cong [\overline{\mathcal{M}}_{g,0}(S, \beta)]^{vir}.$$

The two terms in the third line of (3.4) are the \mathbb{C}^* -moving part of E_{\bullet} which give the virtual normal bundle to the \mathbb{C}^* -fixed point set. By Lemma 3.4 we have $R^0\pi_*\mu^*K_S = 0$. Thus the virtual normal bundle is actually the bundle $R^1\pi_*\mu^*K_S$. Then by Bott's formula (2.14), we have

$$N_{g,\iota_*\beta}(\mathbb{P}(K_S \oplus \mathcal{O}_S)) = \int_{[\overline{\mathcal{M}}_{g,0}(S,\beta)]^{vir}} e_{\mathbb{C}^*}(R^1\pi_*\mu^*K_S)$$
$$= \int_{[\overline{\mathcal{M}}_{g,0}(S,\beta)]^{vir}} e(R^1\pi_*\mu^*K_S).$$

The second equality follows since the rank of $R^1 \pi_* \mu^* K_S$ is equal to the virtual dimension of $\overline{\mathcal{M}}_{g,0}(S,\beta)$.

3.2.3 Deformation invariance

Recall that Gomov–Witten invariants of projective varieties are invariant under deformations (cf. §2.1.3).

Proposition 3.7. Let S be a -K-nef surface and S' be a -K-nef surface which is deformation equivalent to S. Let $\beta \in H_2(S, \mathbb{Z})$ be a class satisfying $\int_{\beta} c_1(K_S) < 0$ and $\beta' \in H_2(S', \mathbb{Z})$ be the class corresponding to β under a deformation. Then $I_{g,\beta}(S) = I_{g,\beta'}(S')$ for $g \in \mathbb{Z}_{\geq 0}$. *Proof.* Since S and S' are deformation equivalent, $\mathbb{P}(K_S \oplus \mathcal{O}_S)$ and $\mathbb{P}(K_{S'} \oplus \mathcal{O}_{S'})$ are also deformation equivalent. Let $\iota : S \hookrightarrow \mathbb{P}(K_S \oplus \mathcal{O}_S)$ and $\iota' : S' \hookrightarrow \mathbb{P}(K_{S'} \oplus \mathcal{O}_{S'})$ be the inclusions as the zero sections of K_S and $K_{S'}$ respectively. We have

$$I_{g,\beta}(S) = N_{g,\iota_*\beta}(\mathbb{P}(K_S \oplus \mathcal{O}_S)) = N_{g,\iota'_*\beta'}(\mathbb{P}(K_{S'} \oplus \mathcal{O}_{S'})) = I_{g,\beta'}(S').$$

The middle equality follows from the deformation invariance of Gromov–Witten invariants. The first and the third equalities follow from Proposition 3.5.

4 Gromov–Witten invariants of local toric surfaces

First, we introduce the equivariant Gromov–Witten invariants of local toric surfaces. The Gromov–Witten invariants of local toric surfaces are defined as a certain non-equivariant limit of equivariant ones. Then we give a closed formula for the partition functions of local toric surfaces (Theorem 4.6), which has been stated in Theorem 1.3. Its proof will be given in §5.

4.1 Equivariant Gromov–Witten invariants of local toric surfaces

Let S be a nonsingular complete toric surface (see, e.g. [27, §2.5]). There is a canonical action of $\mathbb{T} = (\mathbb{C}^*)^2$ on S. We regard the canonical bundle K_S of S as a \mathbb{T} -equivariant line bundle

$$K_S = \mathcal{O}_S(-\sum_{i=1}^r C_i),$$

where C_1, \dots, C_r are the irreducible toric divisors (cf. [73, §2.1]). By Lemma 3.1, there is a two-term complex $[K_0 \to K_1]$ of vector bundles on $\overline{\mathcal{M}}_{g,0}(S,\beta)$ which is \mathbb{T} -equivariant and whose cohomology sheaves are $R^0 \pi_* \mu^* K_S$ and $R^1 \pi_* \mu^* K_S$. We define

$$e_{\mathbb{T}\times\mathbb{C}^*}(-R^\bullet\pi_*\mu^*K_S) := \frac{e_{\mathbb{T}\times\mathbb{C}^*}(K_1)}{e_{\mathbb{T}\times\mathbb{C}^*}(K_0)} \in A^{\operatorname{vdim}}_{\mathbb{T}\times\mathbb{C}^*}(\overline{\mathcal{M}}_{g,0}(S,\beta)) \otimes Q_{\mathbb{T}\times\mathbb{C}^*},$$
(4.1)

where the second factor in $\mathbb{T} \times \mathbb{C}^*$ acts on K_S by the scalar multiplication fiber-wisely and acts trivially on $\overline{\mathcal{M}}_{q,0}(S,\beta)$.

Definition 4.1. Let us consider the following equivariant integral:

$$N_{g,\beta}^{\mathbb{T}\times\mathbb{C}^*}(K_S) := \int_{[\overline{\mathcal{M}}_{g,0}(S,\beta)]_{\mathbb{T}\times\mathbb{C}^*}^{vir}} e_{\mathbb{T}\times\mathbb{C}^*}(-R^\bullet\pi_*\mu^*K_S) \in \mathbb{Q}(\alpha_1,\alpha_2,\alpha_3),$$
(4.2)

Here α_1 and α_2 are the weights of \mathbb{T} and α_3 is that of \mathbb{C}^* (cf. §2.2.2). We call $N_{g,\beta}^{\mathbb{T}\times\mathbb{C}^*}(K_S)$ the equivariant Gromov-Witten invariant of local toric surface K_S .

We shall compute $N_{g,\beta}^{\mathbb{T}\times\mathbb{C}^*}(K_S)$ by virtual localization formula (2.13). It turns out that the non-equivariant limit with respect to the second factor

$$\lim_{\alpha_3 \to 0} N_{g,\beta}^{\mathbb{T} \times \mathbb{C}^*}(K_S) =: N_{g,\beta}(K_S)$$
(4.3)

exists (cf. the last paragraph in $\S5.4$).

Definition 4.2. We call $N_{g,\beta}(K_S)$ the Gromov–Witten invariant of a local toric surface K_S .⁷

 $^{^7}$ This definition was suggested by Hiroshi Iritani. The author is grateful to him for discussions in July 2006 at Hokkaido University.

A priori, $N_{g,\beta}(K_S)$ takes values in $\mathbb{Q}(\alpha_1, \alpha_2)$. However, it turns out that $N_{g,\beta}(K_S) \in \mathbb{Q}$.

Remark 4.3. The total space of K_S is an example of toric Calabi–Yau threefolds, which is called a local toric surface. $N_{g,\beta}(K_S)$ are special cases of Gromov–Witten invariants of toric Calabi–Yau threefolds. See §8.3.

We discuss the relations between $N_{g,\beta}(K_S)$ and $I_{g,\beta}(S)$. In general, they are different. For instance, $I_{g,\beta}(S)$ is a deformation invariant as we mentioned, but $N_{g,\beta}(K_S)$ is not (see Remark 6.2 (i)). However, we have the following

Lemma 4.4. If S is a -K-nef toric surface, then

$$N_{g,\beta}(K_S) = I_{g,\beta}(S), \tag{4.4}$$

for β satisfying $\int_{\beta} c_1(K_S) < 0$.

Proof. This follows from (3.2) and Bott's formula (2.14).

Remark 4.5. The non-equivariant limit with respect to the first factor

$$\lim_{\alpha_1,\alpha_2\to 0} N_{g,\beta}^{\mathbb{T}\times\mathbb{C}^*}(K_S) =: I_{g,\beta}^{\mathbb{C}^*}(S) \in \mathbb{Q}[\alpha_3,\alpha_3^{-1}] ,$$

is equal to so-called the equivariant local Gromov–Witten invariants of S, which have been considered, e.g. in [10]. If we further take the specialization $\alpha_3 = 1$, we recover local Gromov– Witten invariants of S:

$$I_{g,\beta}^{\mathbb{C}^*}(S)|_{\alpha_3=1} = I_{g,\beta}(S).$$

In this sense, the equivariant Gromov–Witten invariants $N_{g,\beta}^{\mathbb{T}\times\mathbb{C}^*}(K_S)$ of K_S are the universal objects which contain both the Gromov–Witten invariants $N_{g,\beta}(K_S)$ of K_S and the local Gromov–Witten invariants $I_{g,\beta}(S)$ of S.

4.2 Topological vertex formula

Let us introduce the following generating function of Gromov–Witten invariants of a local toric surface K_S :

$$F_{K_S}(\lambda, Q) := \sum_{\beta \in H_2(S,\mathbb{Z}), \beta \neq 0} \sum_{g \ge 0} N_{g,\beta}(K_S) \lambda^{2g-2} Q^{\beta}, \tag{4.5}$$

and its partition function

$$Z_{K_S}(\lambda, Q) := \exp F_{K_S}(\lambda, Q). \tag{4.6}$$

The following formula (4.7) was obtained by Zhou [82] when S is a toric del Pezzo surface.

Theorem 4.6. Let S be a nonsingular complete toric surface, C_1, \dots, C_r the irreducible toric divisors on S, and s_i the self-intersection number of C_i . Set $t_i = Q^{[C_i]}$ $(1 \le i \le r)$. Then we have

$$Z_{K_S}(\lambda, Q) = \sum_{(\nu^1, \dots, \nu^r) \in \mathcal{P}^r} \prod_{i=1}^r ((-1)^{s_i} t_i)^{|\nu^i|} e^{\sqrt{-1}\lambda s_i \frac{\kappa(\nu^i)}{2}} W_{\nu^i, \nu^{i+1}}(e^{\sqrt{-1}\lambda}),$$
(4.7)

where $W_{\nu^{i},\nu^{i+1}}$ has been defined by (2.21).

See §2.3.3 for definitions of combinatorial functions on the set of partitions \mathcal{P} . In §5, we shall prove Theorem 4.6 following the arguments in [82]. It also follows from the results for general toric Calabi–Yau threefolds obtained in [59]. The term $W_{\mu,\nu}$ in (4.7) comes from a formula for two-partition Hodge integrals (Theorem 2.6).

4.3 Gopakumar–Vafa invariants

Gopakumar–Vafa invariants can be defined for Gromov–Witten invariants of K_S by exactly the same way as in Definition 2.2.

Definition 4.7. Define the Gopakumar–Vafa invariants $n^g_\beta(K_S) \in \mathbb{Q}$ by the following formula:

$$F_{K_S}(\lambda, Q) = \sum_{\beta \neq 0} \sum_{g=0}^{\infty} \sum_{k=1}^{\infty} \frac{n_{\beta}^g(K_S)}{k} \left(2\sin\frac{k\lambda}{2}\right)^{2g-2} Q^{k\beta}.$$
(4.8)

By using the topological vertex formula (4.7), Peng [75] and Konishi [48] showed the following

Theorem 4.8 (Konishi, Peng). Gopakumar–Vafa invariants $n_{\beta}^{g}(K_{S})$ are integers. Moreover, for each β , $n_{\beta}^{g}(K_{S})$ is equal to zero for all but finite g.

We will use this result later.

5 Localization on moduli spaces of stable maps to toric surfaces

The goal of this section is to prove Theorem 4.6 by using virtual localization. Let S be a nonsingular complete toric surface and $\mathbb{T} := (\mathbb{C}^*)^2$. The \mathbb{T} -action on S induces that on $\overline{\mathcal{M}}_{g,0}(S,\beta)$ (cf. 2.2.3). We carry out virtual localization on $\overline{\mathcal{M}}_{g,0}(S,\beta)$ as follows. First, in §5.1–5.3, the following four properties are established:

- The \mathbb{T} -fixed point set $\overline{\mathcal{M}}_{g,0}(S,\beta)^{\mathbb{T}}$ is a disjoint union of irreducible components.
- Each T-fixed component is isomorphic to a quotient of products of moduli spaces of pointed stable curves by a finite group.
- The virtual fundamental class of a T-fixed component is the ordinary fundamental class of moduli spaces of stable curves.
- The T-equivariant Euler class of the virtual normal bundle to a T-fixed component is expressed in terms of ψ and λ classes on moduli spaces of stable curves.

Here we closely follow [30]. It is straightforward to extend the arguments of *loc.cit*. for any nonsingular projective toric varieties (see [78]). As a result, an equivariant Gromov–Witten invariant of K_S is expressed as a sum over equivalence classes of appropriate graphs (§5.4). Then Theorem 4.6 is proven in §5.5. There we follow [82] in which the case when S is a toric del Pezzo surface has been treated.

5.1 The T-fixed components

We identify the components of the \mathbb{T} -fixed point set of $\overline{\mathcal{M}}_{g,0}(S,\beta)$ with a set of (isomorphism classes of) certain decorated graphs.

Definition 5.1. (1) An $\overline{\mathcal{M}}_{g,0}(S,\beta)$ -decorated graph G consists of the data (V, E, g, p, d) where:

(i) V is the vertex set,

- (ii) $g: V \to \mathbb{Z}_{\geq 0}$ is a genus label,
- (iii) $p: V \to S^{\mathbb{T}}$ is a fixed point label, where $S^{\mathbb{T}}$ is the \mathbb{T} -fixed point set on S,
- (iv) E is the edge set,
 - (a) If the edge e connects $v, v' \in V$, then $p(v) \neq p(v')$ and there exists a unique irreducible \mathbb{T} -invariant divisor C_e on S which contains p(v) and p(v'),
 - (b) G is connected,
- (v) $d: E \to \mathbb{Z}_{>0}$ is a degree label,
- (vi) $g = \sum_{v \in V} g(v) + h^1(G)$, where $h^1(G)$ is the first Betti number of G,
- (vii) $\beta = \sum_{e \in E} d(e)[C_e]$, where $[C_e] \in H_2(S, \mathbb{Z})$ is the second homology class of C_e .
- (2) An isomorphism of $\overline{\mathcal{M}}_{g,0}(S,\beta)$ -decorated graphs

$$f: G = (V, E, g, p, d) \to G' = (V', E', g', p', d')$$

is a pair (f_v, f_e) of bijections

$$f_v: V \to V', \quad f_e: E \to E'$$

which preserve all the structures of decorated graphs. An isomorphism of G to itself is called an automorphism of G.

(3) We denote by $G_{g,\beta}(S)$ the set of isomorphism classes of $\overline{\mathcal{M}}_{g,0}(S,\beta)$ -decorated graphs. The automorphism group $\operatorname{Aut}(G)$ of an element $G \in G_{q,\beta}(S)$ is that of a representative of G.

Lemma 5.2. The \mathbb{T} -fixed components of $\overline{\mathcal{M}}_{g,0}(S,\beta)$ are in bijective correspondence with the graph set $G_{g,\beta}(S)$.

Proof. Let $f: C \to S$ be a T-fixed stable map. Then f(C) must be a T-invariant curve on S and the images of nodes, contracted components, and ramification points must be mapped to $S^{\mathbb{T}}$. It follows that each non-contracted component D must be mapped to an irreducible T-invariant curve on S, which is isomorphic to \mathbb{P}^1 . Since D is only ramified over two points on \mathbb{P}^1 , it follows that D is nonsingular and rational. The restriction of f to D is uniquely determined by the degree.

To a T-fixed stable map $f: C \to S$, we associate a graph $G \in G_{q,\beta}(S)$ as follows.

- (i) V is the set of connected components of $f^{-1}(S^{\mathbb{T}})$,
- (ii) g(v) is the arithmetic genus of the component corresponding to v (defined to be 0 if the component is an isolated point),
- (iii) For $v \in V$, p(v) := f(v),
- (iv) E is the set of non-contracted irreducible components of C and $C_e := f(e)$ for $e \in E$,
- (v) For $e \in E$, $d(e) := \deg(f|_e)$.

This gives a one-to-one correspondence between the T-fixed components of $\overline{\mathcal{M}}_{g,0}(S,\beta)$ and the graph set $G_{g,\beta}(S)$.

Let $G \in G_{g,\beta}(S)$. The set of T-fixed stable maps with given graph G can be identified with a quotient of a product of moduli space of pointed stable curves by a finite group as follows. Let

$$\overline{\mathcal{M}}_G = \prod_{v \in V} \overline{\mathcal{M}}_{g(v), \operatorname{val}(v)},$$

where $\overline{\mathcal{M}}_{g(v), \text{val}(v)}$ is the moduli space of stable curves of genus g(v) with val(v) marked points and val(v) is the valence of v. We consider the automorphism group A_G of $\overline{\mathcal{M}}_G$ which fits in the following exact sequence:

$$1 \longrightarrow \prod_{e \in E} \mathbb{Z}/d(e) \longrightarrow A_G \longrightarrow \operatorname{Aut}(G) \longrightarrow 1.$$

Here Aut(G) acts naturally on $\prod_{e \in E} \mathbb{Z}/d(e)$ and A_G is the semi-direct product $A_G = \prod_{e \in E} \mathbb{Z}/d(e) \ltimes Aut(G)$.

Lemma 5.3. Let \mathcal{Q}_G be the \mathbb{T} -fixed point component of $\overline{\mathcal{M}}_{g,0}(S,\beta)$ labeled by $G \in G_{g,\beta}(S)$. Then \mathcal{Q}_G can be identified with $\overline{\mathcal{M}}_G/A_G$ as a set.

5.2 Perfect obstruction theories on the \mathbb{T} -fixed components

We study the scheme structure of the T-fixed component \mathcal{Q}_G labeled by $G \in G_{g,\beta}(S)$. It is determined by the perfect obstruction theory on $\overline{\mathcal{M}}_{g,0}(S,\beta)$ restricted to \mathcal{Q}_G . Let $E_{\bullet} = [E_0 \to E_1]$ be the perfect obstruction theory on $\overline{\mathcal{M}}_{g,0}(S,\beta)$. Let $E_{0,G} \to E_{1,G}$ be the restriction of E_{\bullet} to \mathcal{Q}_G . The tangent sheaf \mathcal{T}_G^1 and the obstruction sheaf \mathcal{T}_G^2 on \mathcal{Q}_G are defined by the following exact sequence:

$$0 \longrightarrow \mathcal{T}_G^1 \longrightarrow E_{0,G} \longrightarrow E_{1,G} \longrightarrow \mathcal{T}_G^2 \longrightarrow 0.$$
 (5.1)

Recall that we have the following exact sequence of sheaves on \mathcal{Q}_G (cf. (2.4)):

$$0 \longrightarrow \operatorname{Ext}^{0}(\Omega_{\mathrm{C}}, \mathcal{O}_{\mathrm{C}}) \longrightarrow H^{0}(C, f^{*}T_{S}) \longrightarrow \mathcal{T}_{G}^{1}$$

$$\longrightarrow \operatorname{Ext}^{1}(\Omega_{\mathrm{C}}, \mathcal{O}_{\mathrm{C}}) \longrightarrow H^{1}(C, f^{*}T_{S}) \longrightarrow \mathcal{T}_{G}^{2} \longrightarrow 0.$$
(5.2)

Note that the four term other than \mathcal{T}_G^i in (5.2) are vector bundles on \mathcal{Q}_G . Therefore, we have represented them by their fibers at each point $(f, C) \in \mathcal{Q}_G$.

Lemma 5.4. The \mathbb{T} -fixed part $\mathcal{T}_G^{1,f}$ of \mathcal{T}_G^1 is isomorphic to that of the fourth term in (5.2). The \mathbb{T} -fixed part $\mathcal{T}_G^{2,f}$ of \mathcal{T}_G^2 is zero.

Proof. See [30, the second paragraph in p.503]. (Actually, this follows form proofs of Lemmas 5.8 and 5.10 below.) \Box

As a result, we have the following

Lemma 5.5. \mathcal{Q}_G is isomorphic to $\overline{\mathcal{M}}_G/A_G$. Moreover the perfect obstruction theory on \mathcal{Q}_G is trivial in the sense that its virtual fundamental class is a usual fundamental class of $\overline{\mathcal{M}}_G/A_G$.

Proof. By Lemma 5.4, $\mathcal{T}_G^{1,f}$ is isomorphic to the tangent bundle of $\overline{\mathcal{M}}_G$ and $\mathcal{T}_G^{2,f} = 0$. Then the assertion follows from the exact sequence (5.1). See [30, p.505–506] for detail.

Then, by Theorem 2.5, we have the following

Proposition 5.6.

$$\left[\overline{\mathcal{M}}_{g,0}(S,\beta)\right]_{\mathbb{T}}^{vir} = \iota_* \sum_{G \in G_{g,\beta}(S)} \frac{1}{|A_G|} \frac{|\mathcal{M}_G|}{e_{\mathbb{T}}(N_G^{vir})} \quad \in A_{\mathrm{vdim}}^{\mathbb{T}}(\overline{\mathcal{M}}_{g,0}(S,\beta)) \otimes Q_{\mathbb{T}} \quad .$$
(5.3)

5.3 Virtual normal bundles of the T-fixed components

We compute the equivariant Euler class of the virtual normal bundle of Q_G . By definition of the virtual normal bundle (cf. Definition 2.4) and the exact sequences (5.1, 5.2), we have

$$\frac{1}{e_{\mathbb{T}}(N_G^{vir})} = \frac{e_{\mathbb{T}}(\operatorname{Ext}^0(\Omega_C, \mathcal{O}_C)^m)}{e_{\mathbb{T}}(\operatorname{Ext}^1(\Omega_C, \mathcal{O}_C)^m)} \cdot \frac{e_{\mathbb{T}}(H^1(C, f^*T_S)^m)}{e_{\mathbb{T}}(H^0(C, f^*T_S)^m)}.$$
(5.4)

First, we compute the first factor in (5.4). Let us introduce some notations. Let $G = (V, E, g, p, d) \in G_{g,\beta}(S)$. We set

$$V_1 := \{ v \in V \mid (g(v), \operatorname{val}(v)) = (0, 1) \}, V_2 := \{ v \in V \mid (g(v), \operatorname{val}(v)) = (0, 2) \}, V_3 := \{ v \in V \mid (g(v), \operatorname{val}(v)) \neq (0, 1), (0, 2) \}.$$

Definition 5.7. (i) A flag F is a pair (v, e) of a vertex $v \in V$ and an edge $e \in E$ which is incident on v. A flag F = (v, e) is called a flag incident on v We denote the set of flags incident on v by \mathcal{F}_v .

(ii) For $v \in V_1$, \mathcal{F}_v consists of a unique element, which is denoted by F(v).

(iii) For $v \in V_2$, there are exactly two elements in \mathcal{F}_v which are denoted by $F_1(v)$ and $F_2(v)$.

(iv) Let $v \in V_2 \cup V_3$. For $F = (v, e) \in \mathcal{F}_v$, we define $\psi_F := c_1(L_F)$ (the ψ -class associated to F), where L_F is the line bundle over $\overline{\mathcal{M}}_G$ whose fiber is the cotangent line to the component e of the domain curve at the node v.

(v) For a T-invariant divisor C and a T-fixed point $p \in S^{\mathbb{T}}$ on S, we denote the T-weight of $\mathcal{O}_S(C)$ at p by $\omega_p^C = \omega_p^C(\alpha_1, \alpha_2) \in R_{\mathbb{T}}$. Here α_1, α_2 are the weights of the standard representations of factors of T (cf. §2.2.2).

(vi) For a flag
$$F = (v, e)$$
, we define $\omega_F := \frac{\omega_{p(v)}^{C_e}}{d(e)} \in R_{\mathbb{T}}$.

We have the following

Lemma 5.8. (i)

$$e_{\mathbb{T}}(\operatorname{Ext}^{0}(\Omega_{C}, \mathcal{O}_{C})^{m}) = \prod_{v \in V_{1}} \omega_{F(v)}.$$

(ii)

$$e_{\mathbb{T}}(\operatorname{Ext}^{1}(\Omega_{C}, \mathcal{O}_{C})^{m}) = \prod_{v \in V_{3}} \prod_{F \in \mathcal{F}_{v}} (\omega_{F} - \psi_{F}) \times \prod_{v \in V_{2}} (\omega_{F_{1}(v)} + \omega_{F_{2}(v)}).$$

Proof. See [30, p.503–504] or [78, §7.1–7.2].

Next, we compute the second factor in (5.4). Let us introduce notations. Let Σ be the fan of S and Σ_n be the set of n-cones of Σ . Note that Σ_2 is in one-to-one correspondence with $S^{\mathbb{T}}$, and Σ_1 is in one-to-one correspondence with the set of irreducible \mathbb{T} -invariant divisors on S. Let $\Sigma_2 = \{\sigma_1, \ldots, \sigma_r\}, \Sigma_1 = \{\rho_1 \ldots, \rho_r\}$, where numberings are given as in Figure 1. Let p_i be the \mathbb{T} -fixed point corresponding to σ_i and C_i be the irreducible \mathbb{T} -invariant divisor corresponding to ρ_i . For $i = 1, \ldots, r$, we set

$$\begin{split} V_j^{(i)} &:= \{ v \in V_j \mid p(v) = p_i \}, \quad j = 1, 2, 3, \\ V^{(i)} &:= V_1^{(i)} \cup V_2^{(i)} \cup V_3^{(i)}, \\ E^{(i)} &:= \{ e \in E \mid C_e = C_i \}. \end{split}$$



Figure 1: Σ (left) and its dual graph (right).

Definition 5.9. (i) We set

$$\omega_i^{\rm in}:=\omega_{p_i}^{C_i},\quad \omega_i^{\rm out}:=\omega_{p_i}^{C_{i+1}},$$

where we have regarded *i* as an element of \mathbb{Z}/r . (ii) Let $s_i := C_i^2$ be the self intersection number of C_i . For $d \in \mathbb{Z}_{\geq 1}$, we define

$$P_{(i)}(d) := \frac{d^{2d}}{(d!)^2} \frac{1}{(\omega_{i-1}^{\text{in}})^d (\omega_i^{\text{out}})^d} \times \begin{cases} \prod_{a=1}^{-s_i d-1} \left(\frac{a}{d} \,\omega_i^{\text{out}} + \omega_i^{\text{in}}\right) & \text{if } s_i < 0, \\ \prod_{a=0}^{s_i d} \left(-\frac{a}{d} \,\omega_i^{\text{out}} + \omega_i^{\text{in}}\right)^{-1} & \text{if } s_i \ge 0. \end{cases}$$
(5.5)

Then we have the following

Lemma 5.10.

$$\frac{e_{\mathbb{T}}(H^{1}(C, f^{*}T_{S})^{m})}{e_{\mathbb{T}}(H^{0}(C, f^{*}T_{S})^{m})} = \prod_{i=1}^{r} \left\{ \prod_{v \in V_{2}^{(i)}} \omega_{i}^{\mathrm{in}} \cdot \omega_{i}^{\mathrm{out}} \times \prod_{v \in V_{3}^{(i)}} (\omega_{i}^{\mathrm{in}} \cdot \omega_{i}^{\mathrm{out}})^{\mathrm{val}(v)-1} \cdot \Lambda_{g(v)}^{\vee}(\omega_{i}^{\mathrm{in}}) \cdot \Lambda_{g(v)}^{\vee}(\omega_{i}^{\mathrm{out}}) \right\} \times \prod_{i=1}^{r} \prod_{e \in E^{(i)}} P_{(i)} \left(d(e) \right) ,$$
(5.6)

where $\Lambda_{g(v)}^{\vee}(u)$ has been defined in §2.3.

Proof. The proof is identical to that of Lemma 5.13 below. See [30, p.504–505] or [78, §7.3.]. \Box By eq.(5.4), Lemmas 5.8 and 5.10, we have the following

Proposition 5.11.

$$\frac{1}{e_{\mathbb{T}}(N_G^{vir})} = \prod_{i=1}^r \left\{ \prod_{v \in V_1^{(i)}} \omega_{F(v)} \times \prod_{v \in V_2^{(i)}} \frac{\omega_i^{\text{in}} \cdot \omega_i^{\text{out}}}{\omega_{F_1(v)} + \omega_{F_2(v)}} \\
\times \prod_{v \in V_3^{(i)}} \frac{(\omega_i^{\text{in}} \cdot \omega_i^{\text{out}})^{\text{val}(v)-1} \cdot \Lambda_{g(v)}^{\vee}(\omega_i^{\text{in}}) \cdot \Lambda_{g(v)}^{\vee}(\omega_i^{\text{out}})}{\prod_{F \in \mathcal{F}_v} (\omega_F - \psi_F)} \right\}$$
(5.7)
$$\times \prod_{i=1}^r \prod_{e \in E^{(i)}} P_{(i)} \left(d(e) \right) .$$

5.4 Equivariant Gromov–Witten invariants of K_S

We compute the $\mathbb{T} \times \mathbb{C}^*$ -equivariant Euler class $e_{\mathbb{T} \times \mathbb{C}^*}(-R^{\bullet}\pi_*\mu^*K_S)$ on $\overline{\mathcal{M}}_G$. First, we introduce some notations.

Definition 5.12. (i) Let s_i^{can} be the degree of the canonical bundle $\mathcal{O}_S(-\sum_{j=1}^r C_j)$ of S pulled back to C_i . By adjunction formula, it is given by

$$s_i^{\rm can} = -s_i - 2$$

(ii) Let ω_i^{can} be the \mathbb{T} -weight of the canonical bundle $\mathcal{O}_S(-\sum_{j=1}^r C_j)$ at p_i . It is given by

$$\omega_i^{\rm can} = -\omega_i^{\rm in} - \omega_i^{\rm out}.$$

(iii) For $d \in \mathbb{Z}_{\geq 1}$, we define

$$Q_{(i)}(d) := \begin{cases} \prod_{a=1}^{-ds_i^{\operatorname{can}}} \left(\frac{a}{d} \,\omega_i^{\operatorname{out}} + \omega_i^{\operatorname{can}} + \alpha_3\right) & \text{if } s_i^{\operatorname{can}} < 0, \\ \prod_{a=0}^{ds_i^{\operatorname{can}}} \left(-\frac{a}{d} \,\omega_i^{\operatorname{out}} + \omega_i^{\operatorname{out}} + \alpha_3\right)^{-1} & \text{if } s_i^{\operatorname{can}} \ge 0. \end{cases}$$
(5.8)

Lemma 5.13. We have

$$\iota_{G}^{*} e_{\mathbb{T} \times \mathbb{C}^{*}} (-R^{\bullet} \pi_{*} \mu^{*} K_{S}) = \prod_{i=1}^{r} \left\{ \prod_{v \in V_{3}^{(i)}} \Lambda_{g(v)}^{\vee}(\omega_{i}^{\operatorname{can}}) \cdot (\omega_{i}^{\operatorname{can}} + \alpha_{3})^{\operatorname{val}(v)-1} \times \prod_{v \in V_{2}^{(i)}} (\omega_{i}^{\operatorname{can}} + \alpha_{3}) \right\} \times \prod_{i=1}^{r} \prod_{e \in E^{(i)}} Q_{(i)} (d(e)) .$$
(5.9)

Proof. Let $(f, C) \in \overline{\mathcal{M}}_G$. We have the following normalization sequence

$$0 \to \mathcal{O}_C \to \bigoplus_{e \in E} \mathcal{O}_{C_e} \oplus \bigoplus_{v \in V_3} \mathcal{O}_{C_v} \to \bigoplus_{v \in V_2} \mathcal{O}_{p(v)} \oplus \bigoplus_{v \in V_3} \mathcal{O}_{p(v)}^{\oplus |\mathcal{F}_v|} \to 0,$$

where in the second term, we have denote by C_v the *f*-contracted component corresponding to $v \in V_3$. By twisting f^*K_S and taking cohomology, we have

$$0 \to H^{0}(C, f^{*}K_{S}) \to \bigoplus_{e \in E} H^{0}(C_{e}, f^{*}K_{S}) \oplus \bigoplus_{v \in V_{3}} H^{0}(C_{v}, f^{*}K_{S})$$
$$\to \bigoplus_{v \in V_{2}} (K_{S})_{p(v)} \oplus \bigoplus_{v \in V_{3}} (K_{S})_{p(v)}^{\oplus |\mathcal{F}_{v}|} \to H^{1}(C, f^{*}K_{S})$$
$$\to \bigoplus_{e \in E} H^{1}(C_{e}, f^{*}K_{S}) \oplus \bigoplus_{v \in V_{3}} H^{1}(C_{v}, f^{*}K_{S}) \to 0.$$

Therefore, as an element in K-group, we have

$$H^{1}(C, f^{*}K_{S}) - H^{0}(C, f^{*}K_{S}) = \sum_{e \in E} H^{1}(C_{e}, f^{*}K_{S}) + \sum_{v \in V_{3}} H^{1}(C_{v}, f^{*}K_{S}) + \sum_{v \in V_{2}} (K_{S})_{p(v)} + \sum_{v \in V_{3}} \operatorname{val}(v)(K_{S})_{p(v)} - \sum_{e \in E} H^{0}(C_{e}, f^{*}K_{S}) - \sum_{v \in V_{3}} (K_{S})_{p(v)}.$$
(5.10)

For the second term in the RHS of (5.10), we have

$$H^1(C_v, f^*K_S) \cong H^1(C_v, \mathcal{O}_{C_v}) \otimes (K_S)_{p(v)} \cong H^0(C_v, \omega_{C_v})^{\vee} \otimes (K_S)_{p(v)}.$$

Other terms in the RHS of (5.10) are trivial vector bundles over $\overline{\mathcal{M}}_G$. Then (5.9) follows immediately.

Here we collect some useful formulas:

$$\omega_i^{\rm in} = -\omega_{i-1}^{\rm out}, \tag{5.11}$$

$$\omega_i^{\text{out}} = -\omega_{i-1}^{\text{in}}, \tag{5.12}$$

$$\omega_i^{\rm in} - s_i \omega_i^{\rm out} = \omega_{i-1}^{\rm out}. \tag{5.13}$$

Lemma 5.14. Let $Q'_{(i)}(d) := Q_{(i)}(d)|_{\alpha_3=0}$. Then we have

$$P_{(i)}(d) \cdot Q'_{(i)}(d) = \frac{d^{2d}}{(d!)^2} \frac{(-1)^{ds_i+1}}{(\omega_{i-1}^{\text{out}})^d (\omega_i^{\text{out}})^d} \\ \times \prod_{a=1}^{d-1} \left(\frac{a}{d} \,\omega_i^{\text{out}} + \omega_i^{\text{in}}\right) \cdot \prod_{a=1}^{d-1} \left(\frac{a}{d} \,\omega_{i-1}^{\text{in}} + \omega_{i-1}^{\text{out}}\right) .$$
(5.14)

Proof. We have three cases:

$$\begin{cases} \text{(i)} \ s_i \ge 0, \ s_i^{\text{can}} < 0 \iff s_i \ge 0, \\ \text{(ii)} \ s_i < 0, \ s_i^{\text{can}} < 0 \iff s_i = -1, \\ \text{(iii)} \ s_i < 0, \ s_i^{\text{can}} \ge 0 \iff s_i \le -2. \end{cases}$$

For the case (i), we have

$$Q'_{(i)}(d) = \underbrace{\prod_{a=1}^{d-1} \left(\frac{a}{d} \,\omega_i^{\text{out}} + \omega_i^{\text{can}}\right)}_{(a)} \times \underbrace{\prod_{a=d}^{s_i d+d} \left(\frac{a}{d} \,\omega_i^{\text{out}} + \omega_i^{\text{can}}\right)}_{(b)} \times \underbrace{\prod_{a=s_i d+1}^{s_i d+2d-1} \left(\frac{a}{d} \,\omega_i^{\text{out}} + \omega_i^{\text{can}}\right)}_{(c)}.$$

We have

$$P_{(i)}(d) \times (b) = (-1)^{s_i d+1}, \tag{5.15}$$

$$(a) = \prod_{a=1}^{d-1} \left(\frac{a-d}{d} \omega_i^{\text{out}} - \omega_i^{\text{in}} \right) = \prod_{a=1}^{d-1} \left(-\frac{a}{d} \omega_i^{\text{out}} - \omega_i^{\text{in}} \right) = (-1)^{d-1} \left(\frac{a}{d} \omega_i^{\text{out}} + \omega_i^{\text{in}} \right),$$
(5.16)

$$(c) = \prod_{a=1}^{d-1} \left\{ (s_i + \frac{a}{d}) \omega_i^{\text{out}} - \omega_i^{\text{in}} \right\} = \prod_{a=1}^{d-1} \left\{ \frac{a}{d} \omega_i^{\text{out}} + (s_i \omega_i^{\text{out}} - \omega_i^{\text{in}}) \right\} = (-1)^{d-1} \prod_{a=1}^{d-1} \left(\frac{a}{d} \omega_{i-1}^{\text{in}} + \omega_{i-1}^{\text{out}} \right) .$$
(5.17)

In the third equality in (5.17) we have used (5.12, 5.13). By equations (5.15, 5.16, 5.17), we get the claim for the case (i). Proofs for other cases are similar.

Let us further introduce notations.

Definition 5.15. (i) For $\mu^1, \mu^2 \in \mathcal{P}$, we define

$$\Omega(\omega^{1}, \omega^{2}, \alpha ; \mu^{1}, \mu^{2}) := (\sqrt{-1})^{l(\mu^{1})+l(\mu^{2})} \\
\times \prod_{i=1}^{l(\mu^{1})} \frac{(\mu_{i}^{1})^{\mu_{i}^{1}}}{\mu_{i}^{1!}} \prod_{a=1}^{\mu^{1}-1} \left(\frac{a}{\mu_{i}^{1}} \omega^{2} + \omega^{1}\right) \times \prod_{i=1}^{l(\mu^{2})} \frac{(\mu_{i}^{2})^{\mu_{i}^{2}}}{\mu_{i}^{2!}} \prod_{a=1}^{\mu^{2}-1} \left(\frac{a}{\mu_{i}^{2}} \omega^{1} + \omega^{2}\right) \quad (5.18) \\
\times \left\{\omega^{1}\omega^{2} \left(-\omega^{1} - \omega^{2} - \alpha\right)\right\}^{l(\mu^{1})+l(\mu^{2})-1} \times \frac{1}{(\omega^{2})^{|\mu^{1}|}} \cdot \frac{1}{(\omega^{1})^{|\mu^{2}|}} ,$$

where ω^i and α are parameters.

(ii) For $\delta \in \mathbb{Z}$ and a partition μ , we define

$$R_{\delta}(\omega^{1}, \omega^{2}, \alpha \; ; \; \mu) := \prod_{a=1}^{l(\mu)} \frac{R_{\delta}^{0}(\omega^{1}, \omega^{2}, \alpha \; ; \; \mu_{i})}{R_{\delta}^{0}(\omega^{1}, \omega^{2}, 0 \; ; \; \mu_{i})},$$
(5.19)

where

$$R^{0}_{\delta}(\omega^{1}, \omega^{2}, \alpha \; ; \; d) := \begin{cases} \prod_{a=1}^{-d\delta} \left\{ \frac{a}{d} \; \omega^{1} + (-\omega^{1} - \omega^{2} + \alpha) \right\} & \text{if } \delta < 0, \\ \prod_{a=0}^{d\delta} \left\{ -\frac{a}{d} \; \omega^{1} + (-\omega^{1} - \omega^{2} + \alpha) \right\}^{-1} & \text{if } \delta \ge 0, \end{cases}$$

for $d \in \mathbb{Z}_{>1}$.

Definition 5.16. Let $G = (V, E, g, p, d) \in G_{g,\beta}(S)$. For $v \in V^{(i)}$, we define partitions μ_v^{in} and μ_v^{out} as follows:

$$\mu_v^{\text{in}} := \{ d(e) \mid (v, e) \in \mathcal{F}_v, \ C_e = C_i \},$$
(5.20)

$$\mu_v^{\text{out}} := \{ d(e) \mid (v, e) \in \mathcal{F}_v, \ C_e = C_{i+1} \}.$$
(5.21)

For $i = 1, \ldots, r$, we define partitions $\nu^{(i)}$ by

$$\nu^{(i)} := \left\{ d(e) \mid e \in E^{(i)} \right\} \,. \tag{5.22}$$

We have

$$l(\mu_v^{\rm in}) + l(\mu_v^{\rm out}) = \operatorname{val}(v) , \qquad (5.23)$$

$$\bigsqcup_{v \in V^{(i)}} \mu_v^{\text{in}} = \bigsqcup_{v \in V^{(i-1)}} \mu_v^{\text{out}} = \nu^{(i)} .$$

$$(5.24)$$

Theorem 5.17. We have

$$N_{g,\beta}^{\mathbb{T}\times\mathbb{C}^{*}}(K_{S}) = \sum_{G\in G_{g,\beta}(S)} \frac{1}{|\operatorname{Aut}(G)|} \cdot \frac{1}{\prod_{i=1}^{r} \prod_{j=1}^{l(\nu^{(i)})} \nu_{j}^{(i)}} \times \prod_{i=1}^{r} R_{-2-s_{i}}(\omega_{i}^{\text{in}}, \omega_{i}^{\text{out}}, \alpha_{3} ; \nu^{(i)}) \cdot (-1)^{|\nu^{(i)}|s_{i}} \times \prod_{i=1}^{r} \{\prod_{v\in V^{(i)}} \Omega(\omega_{i}^{\text{in}}, \omega_{i}^{\text{out}}, \alpha_{3} ; \mu_{v}^{\text{in}}, \mu_{v}^{\text{out}}) \cdot (-1)^{|\nu^{(i)}|s_{i}} \times \prod_{i=1}^{r} \{\prod_{v\in V^{(i)}} \Omega(\omega_{i}^{\text{in}}, \omega_{i}^{\text{out}}, \alpha_{3} ; \mu_{v}^{\text{in}}, \mu_{v}^{\text{out}}) \cdot (-1)^{|\nu^{(i)}|s_{i}} \times \prod_{i=1}^{r} \{\prod_{v\in V^{(i)}} \Omega(\omega_{i}^{\text{in}}, \omega_{i}^{\text{out}}, \alpha_{3} ; \mu_{v}^{\text{in}}, \mu_{v}^{\text{out}}) \cdot (-1)^{|\nu^{(i)}|s_{i}} \cdot (-1)^{|\nu^{(i)}|s_{i}} \cdot (-1)^{|\mu^{(i)}|s_{i}} \times \prod_{i=1}^{r} \{\prod_{v\in V^{(i)}} \Omega(\omega_{i}^{\text{in}}, \omega_{i}^{\text{out}}, \alpha_{3} ; \mu_{v}^{\text{in}}, \mu_{v}^{\text{out}}) \cdot (-1)^{|\mu^{(i)}|s_{i}} \cdot (-1)^{|\mu^{(i)}|s_{i}}$$

Proof. Note that the first line of the RHS comes from $|A_G|$, the second line is the contributions from edges and the last two lines are the contributions from vertices. We deal with the the signs coming form Lemma 5.14, as follows.

$$\prod_{i=1}^{r} \prod_{e \in E^{(i)}} (-1)^{d(e)s_i+1} = \prod_{i=1}^{r} \prod_{e \in E^{(i)}} (-1)^{d(e)s_i} \times \prod_{i=1}^{r} \prod_{e \in E^{(i)}} (-1)$$
$$= \prod_{i=1}^{r} (-1)^{|\nu^{(i)}|s_i} \times \prod_{i=1}^{r} \prod_{v \in V^{(i)}} (\sqrt{-1})^{l(\mu_v^{\text{in}}) + l(\mu_v^{\text{out}})}$$
(5.26)

We insert the term (*) into Ω . The rest of the verification is straightforward.

From (5.25), we can see that the non-equivariant limit

$$\lim_{\alpha_3 \to 0} N_{g,\beta}^{\mathbb{T} \times \mathbb{C}^*}(K_S) = N_{g,\beta}(K_S)$$

exists.

5.5 Gromov–Witten invariants of K_S

In this section, we complete the proof of Theorem 4.6. Let us introduce some notations on graphs.

Definition 5.18. (i) We set $G_{\beta}(S) := \bigsqcup_{g \ge 0} G_{g,\beta}(S)$. Let $G^{\bullet}_{\beta}(S)$ be the disconnected version of $G_{\beta}(S)$ (i.e. graphs may be disconnected).

(ii) For $\vec{\eta} = (\eta_1, \ldots, \eta_r) \in \mathcal{P}^r$, we set

$$G_{\beta,\vec{\eta}}(S) := \left\{ G \in G_{\beta}(S) \mid \nu^{(i)} = \eta^{i} \right\}, \quad G^{\bullet}_{\beta,\vec{\eta}}(S) := \left\{ G \in G^{\bullet}_{\beta}(S) \mid \nu^{(i)} = \eta^{i} \right\}.$$

(iii) For $G \in G_{\beta}(S)$ or $G^{\bullet}_{\beta,\tilde{\eta}}(S)$, we define

$$\chi(G) := -2\left(\sum_{v \in V} g(v) + |E| - |V|\right).$$

Definition 5.19. (i) Given a decorated graph $G = (V, E, g, p, d) \in G_{\beta}(S)$, the decorated graph $\mathbb{G} = (V, E, p, d)$ obtained from G by ignoring the genus label g is called the type of G. We denote by $\mathbb{G}_{\beta}(S)$ the set of types of decorated graphs in $G_{\beta}(S)$. Its disconnected version is denoted by $\mathbb{G}_{\beta}^{\bullet}(S)$. We have natural maps

$$\psi: G_{\beta}(S) \to \mathbb{G}_{\beta}(S), \quad G^{\bullet}_{\beta}(S) \to \mathbb{G}^{\bullet}_{\beta}(S).$$

(ii) For $\vec{\eta} \in \mathcal{P}^r$ we define $\mathbb{G}_{\beta,\vec{\eta}}(S)$ and $\mathbb{G}^{\bullet}_{\beta,\vec{\eta}}(S)$ as in Definition 5.18 (ii).

By Theorem 5.17, we have

$$N_{g,\beta}(K_S) = \sum_{G \in G_{g,\beta}(S)} F(G), \qquad (5.27)$$

here F(G) is defined by

$$F(G) := \frac{1}{|\operatorname{Aut}(G)|} \cdot \frac{1}{\prod_{i=1}^{r} \prod_{j=1}^{l(\nu^{(i)})} \nu_{j}^{(i)}} \times \prod_{i=1}^{r} \prod_{v \in V^{(i)}} z_{\mu_{v}^{\text{in}}} \cdot z_{\mu_{v}^{\text{out}}} \cdot G_{g(v), \mu_{v}^{\text{in}}, \mu_{v}^{\text{out}}} \left(\frac{\omega_{i}^{\text{in}}}{\omega_{i}^{\text{out}}}\right) ,$$
(5.28)

where $G_{g,\mu^+,\mu^-}(\tau)$ has been defined by (2.18).

Lemma 5.20. For $\mathbb{G} \in \mathbb{G}_{\beta,\vec{\eta}}$, we have

$$\sum_{G \in \psi^{-1}(\mathbb{G})} F(G) \,\lambda^{-\chi(G)} = H(\mathbb{G}),\tag{5.29}$$

here $H(\mathbb{G})$ is given by

$$H(\mathbb{G}) := \frac{1}{|\operatorname{Aut}(\mathbb{G})|} \cdot \frac{1}{\prod_{i=1}^{r} \prod_{j=1}^{l(\eta^{i})} \eta_{j}^{i}} \times \prod_{i=1}^{r} \prod_{v \in V^{(i)}} z_{\mu_{v}^{\text{in}}} \cdot z_{\mu_{v}^{\text{out}}} \cdot G_{\mu_{v}^{\text{in}}, \mu_{v}^{\text{out}}} \left(\lambda; \frac{\omega_{i}^{\text{in}}}{\omega_{i}^{\text{out}}}\right) ,$$

$$(5.30)$$

where $G_{\mu^+, \mu^-}(\lambda; \tau)$ has been define by (2.19).

Proof. By expanding the RHS of (5.29), we have a sum over all decorated graphs⁸ which have the type \mathbb{G} . For each $G \in \psi^{-1}(\mathbb{G})$, there are exactly $|\operatorname{Aut}(\mathbb{G})|/|\operatorname{Aut}(G)|$ representatives of G. This completes the proof.

Let

$$F_{K_S}(\lambda)_{\beta} := \sum_{g \ge 0} N_{g,\beta}(K_S) \lambda^{2g-2}.$$

Then by Lemma 5.20, we have

$$F_{K_S}(\lambda)_{\beta} = \sum_{\vec{\eta} \in \mathcal{P}^r} \sum_{\mathbb{G} \in \mathbb{G}_{\beta, \vec{\eta}}} H(\mathbb{G})$$

We have

$$F_{K_{S}}(\lambda, Q) = \sum_{\substack{\beta \in H_{2}(S,\mathbb{Z}) \\ \beta \neq 0}} F_{K_{S}}(\lambda)_{\beta} Q^{\beta}$$

$$= \sum_{\substack{\beta \in H_{2}(S,\mathbb{Z}) \\ \beta \neq 0}} \left(\sum_{\vec{\eta} \in \mathcal{P}^{r}} \sum_{\mathbb{G} \in \mathbb{G}_{\beta,\vec{\eta}}} H(\mathbb{G}) \right) Q^{\beta}$$

$$= \sum_{\substack{\vec{\eta} \in \mathcal{P}^{r} \\ \eta \neq \vec{\emptyset}}} \sum_{\mathbb{G} \in \mathbb{G}_{\beta,\vec{\eta}}} H(\mathbb{G}) Q^{\beta} , \qquad (5.31)$$

where $\vec{\emptyset} = (\emptyset, \dots, \emptyset)$. If we set $t_i := Q^{[C_i]}$ $(1 \le i \le r)$, we have

$$F_{K_S}(\lambda, t_1, \dots, t_r) = \sum_{\substack{\vec{\eta} \in \mathcal{P}^r \\ \eta \neq \vec{\emptyset}}} \sum_{\mathbb{G} \in \mathbb{G}_{\beta, \vec{\eta}}} H(\mathbb{G}) \prod_{i=1}^r t_i^{|\eta^i|} .$$
(5.32)

Now we take exponentials of the both sides of (5.32). As is well-known, we obtain a sum over disconnected graphs (see e.g. [48, Appendix A]):

$$Z_{K_S}(\lambda, t_1, \dots, t_r) = 1 + \sum_{\substack{\vec{\eta} \in \mathcal{P}^r \\ \eta \neq \vec{\emptyset}}} \sum_{\mathbb{G} \in \mathbb{G}_{\beta, \vec{\eta}}} H(\mathbb{G}) \prod_{i=1}^r t_i^{|\eta^i|} .$$
(5.33)

 $^{^{8}}$ Note that this is *not* a sum over isomorphism classes of decorated graphs.

Lemma 5.21.

$$\sum_{\mathbb{G}\in\mathbb{G}^{\bullet}_{\beta,\vec{\eta}}(S)} H(\mathbb{G}) = \frac{1}{\prod_{i=1}^{r} |\operatorname{Aut}(\eta^{i})| \prod_{j=1}^{l(\eta^{i})} \eta_{j}^{i}} \times \prod_{i=1}^{r} (-1)^{|\eta^{i}|s_{i}} \times \prod_{i=1}^{r} z_{\eta^{i}} \cdot z_{\eta^{i+1}} \cdot G^{\bullet}_{\eta^{i},\eta^{i+1}} \left(\lambda; \frac{\omega_{i}^{\operatorname{in}}}{\omega_{i}^{\operatorname{out}}}\right) .$$
(5.34)

Proof. This can be shown by expanding the RHS using the definition (2.20) of G^{\bullet} and (5.24). \Box *Proof. (of Themrem 4.6.)* By Lemma 5.21, we have

$$Z_{K_S}(\lambda, t_1, \dots, t_r) = 1 + \sum_{\substack{\vec{\eta} \in \mathcal{P}^r \\ \eta \neq \vec{\emptyset}}} \prod_{i=1}^r z_{\eta^i} \times \prod_{i=1}^r \left((-1)^{s_i} t_i \right)^{|\eta^i|} \times \prod_{i=1}^r G^{\bullet}_{\eta^i, \eta^{i+1}}\left(\lambda; \frac{\omega_i^{\text{in}}}{\omega_i^{\text{out}}}\right) .$$
(5.35)

Now we apply Theorem 2.6. We have

$$Z_{K_{S}} = 1 + \sum_{\substack{\vec{\eta} \in \mathcal{P}^{r} \\ \eta \neq \vec{\emptyset}}} \prod_{i=1}^{r} z_{\eta^{i}} \times \prod_{i=1}^{r} ((-1)^{s_{i}} t_{i})^{|\eta^{i}|} \\ \times \prod_{i=1}^{r} \sum_{\substack{\mu^{i}, \mu^{i}_{+} \\ |\mu^{i}| = |\eta^{i}| \\ |\mu^{i}| = |\eta^{i}| \\ |\mu^{i}| = |\eta^{i}| \\ |\mu^{i}| = |\eta^{i+1}|}} \frac{\chi_{\mu^{i}}(C_{\eta^{i}})}{z_{\eta^{i}}} \frac{\chi_{\mu^{i}}(C_{\eta^{i+1}})}{z_{\eta^{i+1}}} e^{\sqrt{-1} \left(\kappa(\mu^{i}) \frac{\omega_{i}^{\text{in}}}{\omega_{i}^{\text{out}}} + \kappa(\mu^{i}_{+}) \frac{\omega_{i}^{\text{out}}}{\omega_{i}^{\text{in}}}\right) \lambda/2} W_{\mu^{i}, \mu^{i}_{+}}(e^{\sqrt{-1}\lambda}) .$$
(5.36)

By taking sum over η^i and using orthogonality of characters (2.15) for each *i*, we can see that only terms which satisfy

$$\mu^i_+ = \mu^{i+1}$$
 for all $i \in \mathbb{Z}/r$

survive. Hence we have

$$Z_{K_{S}} = \sum_{(\mu^{1},...,\mu^{r})\in\mathcal{P}^{r}} \prod_{i=1}^{r} \left((-1)^{s_{i}} t_{i} \right)^{|\mu^{i}|} \cdot e^{\sqrt{-1} \left(\kappa(\mu^{i}) \frac{\omega_{i}^{\mathrm{in}}}{\omega_{i}^{\mathrm{out}}} + \kappa(\mu^{i+1}) \frac{\omega_{i}^{\mathrm{out}}}{\omega_{i}^{\mathrm{in}}} \right) \lambda/2} \cdot W_{\mu^{i},\mu^{i+1}} (e^{\sqrt{-1}\lambda})$$

$$= \sum_{(\mu^{1},...,\mu^{r})\in\mathcal{P}^{r}} \prod_{i=1}^{r} \left((-1)^{s_{i}} t_{i} \right)^{|\mu^{i}|} \cdot e^{\sqrt{-1}\kappa(\mu^{i}) \left(\frac{\omega_{i}^{\mathrm{in}}}{\omega_{i}^{\mathrm{out}}} + \frac{\omega_{i-1}^{\mathrm{out}}}{\omega_{i-1}^{\mathrm{in}}} \right) \lambda/2} \cdot W_{\mu^{i},\mu^{i+1}} (e^{\sqrt{-1}\lambda})$$

$$= \sum_{(\mu^{1},...,\mu^{r})\in\mathcal{P}^{r}} \prod_{i=1}^{r} \left((-1)^{s_{i}} t_{i} \right)^{|\mu^{i}|} \cdot e^{\sqrt{-1}\kappa(\mu^{i})s_{i}\lambda/2} \cdot W_{\mu^{i},\mu^{i+1}} (e^{\sqrt{-1}\lambda}) .$$

In the last equality, we have used (5.12, 5.13).

6 Local Gromov–Witten invariants of cubic surfaces

In this section, we give a proof of Theorem 1.4 (Theorem 6.8). In §6.1, an example which illustrate the argument is given. We summarize necessary facts on cubic surfaces S_3 in §6.2. In §6.3, the -K-nef toric surface S_3^0 is introduced. (A proof of the deformation equivalence of S_3 and S_3^0 is given in §6.6.) Formulas for the generating functions of local Gromov–Witten invariants of S_3^0 and S_3 are obtained in §6.4. Numerical results are presented in §6.4.3. In §6.5, local Gopakumar–Vafa invariants of S_3 are studied.

6.1 Example of -K-nef toric degeneration

We illustrate our arguments in §6 with an example. Consider the Hirzebruch surface $\mathbb{F}_n = \mathbb{P}(\mathcal{O}_{\mathbb{P}^1} \oplus \mathcal{O}_{\mathbb{P}^1}(-n))$ $(n \ge 0)$. We denote the homology class of the canonical section of $\mathbb{F}_n \to \mathbb{P}^1$ with self-intersection number -n by B_n and that of the fiber by F. It is well-known that $\mathbb{F}_0 = \mathbb{P}^1 \times \mathbb{P}^1$ is deformation equivalent to \mathbb{F}_2 . Note that \mathbb{F}_2 is a -K-nef toric surface but not a del Pezzo surface. We call \mathbb{F}_2 is a -K-nef toric degeneration of \mathbb{F}_0 .

There is a one parameter deformation $\pi : \mathcal{X} \to T$ of \mathbb{F}_2 such that $X_0 = \pi^{-1}(t_0) \cong \mathbb{F}_2$ and $X_t = \pi^{-1}(t) \cong \mathbb{F}_0$ if $t \neq t_0$, where t_0 is some fixed base point. See e.g. [68, Ch.1, Theorem 4.2] for an explicit description of the deformation space. The relative Picard group of $\pi : \mathcal{X} \to T$ is a trivial rank 2 local system, where the class F on X_t deforms to the class F on X_0 , and the class B_0 on X_t deforms to $B_2 + F$ on X_0 . By Lemma 4.4 and Proposition 3.7, we have

$$N_{g,aB_0+bF}(K_{\mathbb{F}_0}) = I_{g,aB_0+bF}(\mathbb{F}_0) = I_{g,aB_2+(a+b)F}(\mathbb{F}_2) = N_{g,aB_2+(a+b)F}(K_{\mathbb{F}_2}), \tag{6.1}$$

for $a \ge 0, b \ge 0, (a, b) \ne (0, 0)$. Note that

- $N_{g,aB_0+bF}(K_{\mathbb{F}_0}) = 0$ unless $a \ge 0, b \ge 0$.
- $N_{g,cB_2+dF}(K_{\mathbb{F}_2}) = 0$ unless $c \ge 0, d \ge 0$.
- Therefore, the difference between them only appears at a multiple of $[B_2]$ which does not corresponds to an effective class on \mathbb{F}_0 .

The first two statements follow from the virtual localization formula (2.13). Now we translate these into an identity between partition functions $Z_{K_{\mathbb{F}_0}}(\lambda, Q_{B_0}, Q_F)$ and $Z_{K_{\mathbb{F}_2}}(\lambda, Q_{B_2}, Q_F)$, where $Q_{B_0} = Q^{[B_0]}$, $Q_{B_2} = Q^{[B_2]}$ and $Q_F = Q^{[F]}$.

Proposition 6.1. As an identity between formal power series in two variables Q_1, Q_2 , we have

$$Z_{K_{\mathbb{F}_0}}(\lambda, Q_1 Q_2, Q_2) = \frac{Z_{K_{\mathbb{F}_2}}(\lambda, Q_1, Q_2)}{Z_{K_{\mathbb{F}_2}}(\lambda, Q_1, 0)}.$$
(6.2)

Proof. The denominator of the RHS of (6.2) subtracts the contributions from multiples of $[B_2]$. The substitution of the variable in the LHS is determined by the identification $Q_{B_2} = Q_{B_0}Q_F^{-1}$.

Note that the identity (6.2) is a nontrivial combinatorial identity by Theorem 4.6. Our strategy is to apply the same arguments to a cubic surface S_3 (which is *non-toric*) and its -K-nef toric degeneration S_3^0 . This time, we shall obtain a closed formula for the partition function of local Gromov–Witten invariants of S_3 from that of S_3^0 .

Remark 6.2. (i) In general, \mathbb{F}_n and \mathbb{F}_m are deformation equivalent if and only if $n \equiv m \mod 2$. However, there is no relation such as (6.1) between Gromov–Witten invariants of $K_{\mathbb{F}_n}$ and $K_{\mathbb{F}_m}$ if one of which is not a -K-nef surface. Therefore Gromov–Witten invariants of local toric surfaces are not deformation invariant. Hence -K-nef condition is essential for our argument. We cannot apply our method to S_1 and S_2 , del Pezzo surfaces of degrees 1 and 2, since they do not admits -K-nef toric degenerations.

(ii) In [16], another approach to local Gromov–Witten invariants of non-toric del Pezzo surfaces based on the ruled vertex formula [17]. Although it is still conjectural, it may cover all del Pezzo surfaces including S_1 and S_2 .

(iii) The -K-nef toric degeneration S_3^0 of S_3 has been used in [79] to study quantum cohomology of S_3 .



Figure 2: $S_3^0 \to S_4^0 \to S_5^0 \to S_6$.

6.2 Cubic surfaces S_3

Here we summarize basic facts on cubic surfaces, see e.g. [31, Ch. V.4] for details.

Let $S_3 \subset \mathbb{P}^3$ be a nonsingular cubic surface. S_3 is realized as a blowing up $\pi : S_3 \to \mathbb{P}^2$ at six points in a general position. Let e_1, \dots, e_6 be the classes of the exceptional curves of π and l be the class of a line in \mathbb{P}^2 pulled back by π . Then l, e_1, \dots, e_6 is a basis of $\operatorname{Pic}(S_3)$. Their intersections are

$$l^2 = 1$$
, $e_i^2 = -1$, $l.e_i = 0$, $e_i.e_j = 0$ if $i \neq j$.

Let h be the class of the hyperplane section of \mathbb{P}^3 . Then we have

$$h = -K_{S_3} = 3l - \sum_{i=1}^{6} e_i$$
.

It is a classical fact that S_3 contains exactly twenty-seven lines which are given as follows:

$$e_i \ (i = 1, \cdots, 6), \quad l - e_i - e_j \ (1 \le i < j \le 6), \quad 2l - \sum_{i \ne j} e_i \ (j = 1, \cdots, 6)$$

Each one of these is an exceptional curve of the first kind. These twenty-seven lines are the minimal generators of the Mori cone (the cone generated by effective 1-cycles modulo numerical equivalence) (cf. [67, (0.6)]).

It is well-known that the Weyl group W_{E_6} of type E_6 acts on $Pic(S_3)$ as symmetries of configurations of twenty seven lines. Its generators are given as follows.

$$s_{i}: e_{i} \leftrightarrow e_{i+1} \quad (1 \leq i \leq 5),$$

$$s_{6}: e_{1} \mapsto l - e_{2} - e_{3}, \quad e_{2} \mapsto l - e_{1} - e_{3},$$

$$e_{3} \mapsto l - e_{1} - e_{2}, \quad l \mapsto 2l - e_{1} - e_{2} - e_{3}$$

It is known [19, §4] that W_{E_6} coincides with the group of automorphisms of $\operatorname{Pic}(S_3)$ which preserve the intersection form, the canonical class, and the semigroup of effective classes. Hereafter we identify $\operatorname{Pic}(S_3)$ with $H^2(S_3, \mathbb{Z}) \cong H_2(S_3, \mathbb{Z})$. We have the following

Lemma 6.3. $I_{g,\beta}(S_3) = I_{g,w(\beta)}(S_3)$ for $w \in W_{E_6}$. *Proof.* See [34, §2.4].

6.3 -K-nef toric degenerations of S_3 , S_4 , and S_5

Let S_3^0 , S_4^0 , and S_5^0 be the -K-nef toric surfaces whose fans are given in Figure 2. Here the nine one-dimensional cones of S_3^0 are generated by

$$v_1 = (1,0), v_2 = (0,1), v_3 = (-1,2), v_4 = (-1,1), v_5 = (-1,0), v_6 = (-1,-1), v_7 = (0,-1), v_8 = (1,-1), v_9 = (2,-1).$$

Let the fan of the toric del Pezzo surface S_6 be given in Figure 2 and let p_1, p_2, p_3 be the torus fixed points of S_6 corresponding to the two-dimensional cones generated by $(v_5, v_7), (v_8, v_1), (v_2, v_4)$. S_3^0 (resp. S_4^0, S_5^0) is obtained by blowing up S_6 at p_1, p_2, p_3 (resp. p_1, p_2 and p_1). S_k^0 contains (-2)-curves and its anticanonincal divisor is nef but not ample.

Proposition 6.4. S_k^0 (k = 3, 4, 5) is deformation equivalent to S_k .

A proof will be given in $\S6.6$ (see Proposition 6.13).

Now let us explain the geometry of the -K-nef toric surface S_3^0 . The torus-invariant divisors C_i $(1 \le i \le 9)$ corresponding to v_i have the intersections:

$$C_i \cdot C_{i+1} = 1, \quad C_i \cdot C_j = 0 \ (j \neq i, i \pm 1), \quad C_i^2 = \begin{cases} -1 & (i = 3, 6, 9), \\ -2 & (i = 1, 2, 4, 5, 7, 8), \end{cases}$$
(6.3)

and the canonical divisor $K_{S_3^0}$ is rationally equivalent to $-C_1 - \cdots - C_9$. The Mori cone of S_3^0 is generated by C_1, \ldots, C_9 [73, Proposition 2.26].

Note that $\operatorname{Pic}(S_3^0) \cong \operatorname{Pic}(S_3)$ and an isomorphism is given by the following.

$$C_{1} \mapsto e_{2} - e_{5}, \qquad C_{2} \mapsto l - e_{2} - e_{3} - e_{6}, \qquad C_{3} \mapsto e_{6}, C_{4} \mapsto e_{3} - e_{6}, \qquad C_{5} \mapsto l - e_{1} - e_{3} - e_{4}, \qquad C_{6} \mapsto e_{4}, C_{7} \mapsto e_{1} - e_{4}, \qquad C_{8} \mapsto l - e_{1} - e_{2} - e_{5}, \qquad C_{9} \mapsto e_{5}.$$
(6.4)

This is explained as follows. First, in S_6 , we regard the torus-invariant divisors C'_1, C'_4, C'_7 corresponding to v_1, v_4, v_7 as the exceptional curves of blowing up of \mathbb{P}^2 and identify them with e_2, e_3, e_1 . The torus-invariant divisors C'_2, C'_5, C'_8 corresponding to v_2, v_5, v_8 are identified with the proper transforms $l - e_2 - e_3, l - e_1 - e_3, l - e_1 - e_2$ of lines in \mathbb{P}^2 . Then in S^0_3, C_3, C_6, C_9 are exceptional curves of the blowup at p_3, p_1, p_2 and we identify them with e_6, e_4, e_5 . For $i = 1, 2, 4, 5, 7, 8, C_i$ is the proper transform of C'_i . (This identification can be seen from the construction of a deformation in the proof of Proposition 6.13.)

From here on, we identify $\operatorname{Pic}(S_3^0)$ with $H^2(S_3^0, \mathbb{Z}) \cong H_2(S_3^0, \mathbb{Z})$.

Corollary 6.5. For $g \in \mathbb{Z}_{>0}$ and $\beta \in H_2(S_3, \mathbb{Z})$ such that $K_{S_3} \cdot \beta < 0$,

$$I_{g,\beta}(S_3) = I_{g,\beta'}(S_3^0)$$

where $\beta' \in H_2(S_3^0, \mathbb{Z})$ is the class corresponding to β by eq. (6.4).

Proof. This follows from Propositions 3.7 and 6.4.

Remark 6.6. The statements similar to Theorem 6.5 hold for S_4, S_5 : local Gromov–Witten invariants of S_4 and S_5 are the same as those of S_4^0 and S_5^0 and the latter is obtained from those of S_3^0 (see Corollary 9.7). Their generating functions also have expressions analogous to the formula for S_3 (which will be stated in Theorem 6.8).

6.4 Generating function of local Gromov–Witten invariants

6.4.1 Formula for S_3^0

First we consider the generating function of local Gromov–Witten invariants of S_3^0 with $\beta \in H_2(S_3^0, \mathbb{Z})$ such that $K_{S_2^0}.\beta < 0$. Take a basis c_1, \ldots, c_7 of $H_2(S_3^0, \mathbb{Z})$ and let X_1, \ldots, X_7 be

associated formal variables. For $\beta = a_1c_1 + \cdots + a_7c_7 \in H_2(S_3^0, \mathbb{Z})$, denote $X_1^{a_1} \dots X_7^{a_7}$ by X^{β} . We define the generating function as

$$F_{S_3^0}^{\text{loc}}(\lambda, X) := \sum_{\substack{\beta \in H_2(S_3^0, \mathbb{Z}), \ g \ge 0\\K_{S_3^0}.\beta < 0}} \sum_{g \ge 0} I_{g,\beta}(S_3^0) \lambda^{2g-2} X^{\beta},$$

and its partition function

$$Z^{\mathrm{loc}}_{S^0_3}(\lambda,X):=\exp F^{\mathrm{loc}}_{S^0_3}(\lambda,X)$$

Second we consider the partition function of Gromov–Witten invariants of the local toric surface $Z_{K_{S_2^0}}$. By Theorem 4.6, we have the following formula:

$$Z_{K_{S_3^0}}(\lambda, t_1, \cdots, t_9) = \prod_{i=1}^{9} \sum_{\nu^i} ((-1)^{s_i} t_i)^{|\nu^i|} e^{\sqrt{-1}\lambda s_i \frac{\kappa(\nu^i)}{2}} W_{\nu^i, \nu^{i+1}}(e^{\sqrt{-1}\lambda}), \tag{6.5}$$

where $t_i = X^{[C_i]}$ $(1 \le i \le 9)$ and $s_i = C_i^2$ (See (6.3)), and each ν^i $(1 \le i \le 9)$ runs over the set of partitions ($\nu^{10} = \nu^1$ is assumed). Define $Z_{(-2,0)}(t)$ by

$$Z_{(-2,0)}(\lambda,t) := \exp\left[-\sum_{j\geq 1} \frac{1}{j} \left(2\sin\frac{j\lambda}{2}\right)^{-2} t^{j}\right].$$

Lemma 6.7. Under $t_i = X^{[C_i]}$ $(1 \le i \le 9)$, we have

$$Z_{S_3^0}^{\rm loc}(\lambda, X) = \frac{Z_{K_{S_3^0}}(\lambda, t_1, \cdots, t_9)}{\prod_{i=1,4,7} Z_{(-2,0)}(t_i) Z_{(-2,0)}(t_{i+1}) Z_{(-2,0)}(t_i t_{i+1})}$$

Proof. By Lemma 4.4, we have $I_{g,\beta}(S_3^0) = N_{g,\beta}(K_{S_3^0})$ for β such that $K_{S_3^0}.\beta < 0$. Note also that $N_{g,\beta}(K_{S_3^0}) = 0$ if there is no effective divisors of the form $\sum_{1 \le i \le 9} a_i[C_i]$ $(a_i \in \mathbb{Z}_{\ge 0})$ which are rationally equivalent to β because $\overline{\mathcal{M}}_{g,0}(S_3^0,\beta)^{\mathbb{T}}$ is empty.

We subtract the contributions coming from classes β which does not satisfy $K_{S_3^0}$. $\beta < 0$ from $Z_{K_{S_3^0}}$. By the above remark, such effective classes are of the forms $a[C_1] + b[C_2]$, $a[C_4] + b[C_5]$ or $a[C_7] + b[C_8]$ $(a, b \in \mathbb{Z}_{\geq 0})$. Therefore

$$\exp\left[\sum_{\substack{\beta \in H_2(S_3^0,\mathbb{Z}) \\ K_{S_3^0}, \beta \ge 0}} \sum_{g \ge 0} N_{g,\beta}(K_{S_3^0}) \lambda^{2g-2} X^{\beta}\right] = \prod_{i=1,4,7} \exp\left[\sum_{a,b \in \mathbb{Z}_{\ge 0}} \sum_{g \ge 0} N_{g,a[C_i]+b[C_{i+1}]}(K_{S_3^0}) \lambda^{2g-2} t_i^a t_{i+1}^b\right].$$
(6.6)

The i = 1 factor is easily obtained by setting $t_3 = t_4 = \cdots = t_9 = 0$ in (6.5). It is equal to

$$Z_{K_{S_3^0}}(\lambda, t_1, \cdots, t_9)|_{t_3=t_4=\cdots=t_9=0} = Z_{(-2,0)}(\lambda, t_1)Z_{(-2,0)}(\lambda, t_2)Z_{(-2,0)}(\lambda, t_1t_2) \quad .$$

The i = 4, 7 factors are similar. Dividing (6.5) by (6.6), we obtain

$$\exp\left[\sum_{\substack{\beta \in H_2(S_3^0,\mathbb{Z}), g \ge 0\\K_{S_3^0}.\beta < 0}} \sum_{g \ge 0} N_{g,\beta}(K_{S_3^0}) \lambda^{2g-2} X^{\beta}\right] = \frac{Z_{K_{S_3^0}}(\lambda, t_1, \cdots, t_9)}{\prod_{i=1,4,7} Z_{(-2,0)}(\lambda, t_i) Z_{(-2,0)}(\lambda, t_{i+1}) Z_{(-2,0)}(\lambda, t_i t_{i+1})}.$$

Thus we complete our proof.

6.4.2 Formula for S_3

Next we study the generating function of local Gromov–Witten invariants of S_3 . Let $Q = (Q_1, \ldots, Q_6, Q_7)$ be a set of formal variables and denote $Q_1^{a_1}Q_2^{a_2}\ldots Q_7^{a_7}$ by Q^β for $\beta = a_1e_1 + \cdots + a_6e_6 + a_7l \in H_2(S_3, \mathbb{Z})$. Define

$$F_d^{\mathrm{loc}}(\lambda, Q) := \sum_{\substack{\beta \in H_2(S_3, \mathbb{Z}), \ g \in \mathbb{Z}_{\ge 0} \\ -K_{S_3}, \beta = d}} \sum_{\substack{I_{g,\beta}(S_3)\lambda^{2g-2}Q^{\beta} \ ,} \qquad (d \in \mathbb{Z}_{\ge 1}),$$

$$F_{S_3}^{\mathrm{loc}}(\lambda, Q) := \sum_{d \ge 1} F_d^{\mathrm{loc}}(\lambda, Q),$$

$$Z_{S_3}^{\mathrm{loc}}(\lambda, Q) := \exp F_{S_3}^{\mathrm{loc}}(\lambda, Q).$$

The following is the main result of the part II.

Theorem 6.8. With the following identification of the parameters

$$t_1 = Q^{e_2 - e_5}, \quad t_2 = Q^{l - e_2 - e_3 - e_6}, \quad t_3 = Q^{e_6}, \quad t_4 = Q^{e_3 - e_6}, \quad t_5 = Q^{l - e_1 - e_3 - e_4}, \\ t_6 = Q^{e_4}, \quad t_7 = Q^{e_1 - e_4}, \quad t_8 = Q^{l - e_1 - e_2 - e_5}, \quad t_9 = Q^{e_5},$$
(6.7)

we have

$$Z_{S_3}^{\rm loc}(\lambda,Q) = \frac{Z_{K_{S_3^0}}(\lambda,t_1,\cdots,t_9)}{\prod_{i=1,4,7} Z_{(-2,0)}(\lambda,t_i) Z_{(-2,0)}(\lambda,t_{i+1}) Z_{(-2,0)}(\lambda,t_i t_{i+1})}.$$
(6.8)

Proof. The identification (6.7) is determined by (6.4). Under this identifications of parameters, we have

$$Z_{S_3}^{\mathrm{loc}}(\lambda, Q) = Z_{S_3^0}^{\mathrm{loc}}(\lambda, t_1, \cdots, t_9),$$

by Corollary 6.5. Then (6.8) follows from Lemma 6.7.

Remark 6.9. In [18], Diaconescu and Florea obtained a formula for F_{S_3} which is different from ours (eq.(3.14) for k = 5 in *loc.cit.*). It would be an interesting problem to show that these two formulas are equivalent.

6.4.3 Numerical results up to degree 6

Define the orbit sum $m(\beta)$ for $\beta \in H_2(S_3, \mathbb{Z})$ by

$$m(\beta) := \frac{1}{\#\{w \in W_{E_6} \mid w(\beta) = \beta\}} \sum_{w \in W_{E_6}} Q^{w(\beta)} .$$

By Lemma 6.3, $F_d^{\text{loc}}(\lambda, Q)$ can be expanded in terms of these. Here we list $F_d^{\text{loc}}(\lambda, Q)$ up to d = 6. Let $\mathbf{b}[k] := (2 \sin \frac{k\lambda}{2})^2$.

$$F_1^{\rm loc} = \frac{1}{\mathsf{b}[1]} \, m(e_6) \, , \qquad F_2^{\rm loc} = \frac{1}{2 \cdot \mathsf{b}[2]} \, m(2e_6) + \frac{-2}{\mathsf{b}[1]} \, m(-e_1 + l) \, ,$$

$$F_3^{\text{loc}} = \frac{1}{3 \cdot \mathsf{b}[3]} m(3e_6) + \frac{3}{\mathsf{b}[1]} m(l) + \left(-4 + \frac{27}{\mathsf{b}[1]}\right) m(-e_1 - e_2 - e_3 - e_4 - e_5 - e_6 + 3l) ,$$

$$F_4^{\text{loc}} = \frac{1}{4 \cdot \mathbf{b}[4]} m(4e_6) + \frac{-2}{2 \cdot \mathbf{b}[2]} m(-2e_1 + 2l) + \frac{-4}{\mathbf{b}[1]} m(-e_1 - e_2 + 2l) + \left(5 + \frac{-32}{\mathbf{b}[1]}\right) m(-e_1 - e_2 - e_3 - e_4 - e_5 + 3l)$$

$$\begin{split} F_5^{\rm loc} &= \frac{1}{5 \cdot \mathbf{b}[5]} \, m(5e_6) + \frac{5}{\mathbf{b}[1]} \, m(-e_1 + 2l) + \left(-6 + \frac{35}{\mathbf{b}[1]} \right) m(-e_1 - e_2 - e_3 - e_4 + 3l) \\ &+ \left(7 \cdot \mathbf{b}[1] - 68 + \frac{205}{\mathbf{b}[1]} \right) m(-2e_1 - e_2 - e_3 - e_4 - e_5 - e_6 + 4l) \;, \end{split}$$

$$\begin{split} F_6^{\text{loc}} &= \frac{1}{6 \cdot \mathbf{b}[6]} \, m(6e_6) - \frac{2}{3 \cdot \mathbf{b}[3]} \, m(-3e_1 + 3l) + \left(\frac{3}{2 \cdot \mathbf{b}[2]} - \frac{6}{\mathbf{b}[1]}\right) m(2l) \\ &+ \left(7 - \frac{36}{\mathbf{b}[1]}\right) m(-e_1 - e_2 - e_3 + 3l) \\ &+ \left(-8 \cdot \mathbf{b}[1] + 72 - \frac{198}{\mathbf{b}[1]}\right) m(-2e_1 - e_2 - e_3 - e_4 - e_5 + 4l) \\ &+ \left(9 \cdot \mathbf{b}[1]^2 - 108 \cdot \mathbf{b}[1] + 498 - \frac{936}{\mathbf{b}[1]}\right) m(-e_1 - e_2 - e_3 - e_4 - e_5 - e_6 + 4l) \\ &+ \left(\frac{1}{2}\left(-4 + \frac{27}{\mathbf{b}[2]}\right) - 10 \cdot \mathbf{b}[1]^3 + 141 \cdot \mathbf{b}[2]^2 - 846 \cdot \mathbf{b}[1] + 2636 - \frac{3780}{\mathbf{b}[1]}\right) \\ &\times m(-2e_1 - 2e_2 - 2e_3 - 2e_4 - 2e_5 - 2e_6 + 6l) \;. \end{split}$$

6.5 Gopakumar–Vafa invariants

The local Gipakumar–Vafa invariants $n^g_\beta(S_3)$ $(g \in \mathbb{Z}_{\geq 0}, \beta \in H_2(S_3, \mathbb{Z}))$ are defined by the following expansion:

$$F_{S_3}^{\mathrm{loc}}(\lambda,Q) = \sum_{\beta \in H_2(S_3,\mathbb{Z})} \sum_{g \in \mathbb{Z}_{\geq 0}} \sum_{k \geq 1} \frac{n_\beta^g(S_3)}{k} \Big(2\sin\frac{k\lambda}{2} \Big)^{2g-2} Q^{k\beta}.$$

Proposition 6.10. Gopakumar–Vafa invariants $n^g_\beta(S_3)$ of S_3 are integers. Moreover, for each β , $n^g_\beta(S_3)$ is equal to zero for all but finite g.

Proof. This follows from Theorem 6.8 and the same statement for the toric surface S_3^0 (cf. Theorem 4.8).

From the results in §6.4.3, we can obtain $n_{\beta}^{g}(S_{3})$ for all β satisfying $d = -K_{S_{3}} \cdot \beta \leq 6$. They are listed in Table 1, in which we set $\beta = a_{1}e_{1} + \cdots + a_{6}e_{6} + a_{7}l$.

Remark 6.11. The results in Table 1 are in agreement with previous results in [66, Table 3], [54, Table 1, n = 6], [11, Table 7, $X_3(1, 1, 1, 1)$] obtained by the B-model calculation based on local mirror symmetry. Also compare with [43, Table 7].

d	$\beta = (a_1, \dots, a_6, a_7)$	$ \mathcal{O}(\beta) $	$g(\beta)$	g	0	1	2	3	4	5
1	(0,0,0,0,0,1,0)	27	0		1					
2	(-1, 0, 0, 0, 0, 0, 1)	27	0		-2					
3	(0,0,0,0,0,0,1)	72	0		3					
	(-1, -1, -1, -1, -1, -1, 3)	1	1		27	-4				
4	(-1, -1, 0, 0, 0, 0, 2)	216	0		-4					
	(-1, -1, -1, -1, -1, 0, 3)	27	1		-32	5				
5	(-1, 0, 0, 0, 0, 0, 2)	432	0		5					
	(-1, -1, -1, -1, 0, 0, 3)	216	1		35	-6				
	(-2, -1, -1, -1, -1, -1, 4)	27	2		205	-68	7			
6	(-2, -1, 0, 0, 0, 0, 3)	432	0		-6					
	(0, 0, 0, 0, 0, 0, 2)	72	0		-6					
	(-1, -1, -1, 0, 0, 0, 3)	720	1		-36	7				
	(-2, -1, -1, -1, -1, 0, 4)	270	2		-198	72	-8			
	(-1, -1, -1, -1, -1, -1, 4)	72	3		-936	498	-108	9		
	(-2, -2, -2, -2, -2, -2, 6)	1	4	-	-3780	2636	-846	141	-10	

Table 1: Gopakumar–Vafa invariants $n_{\beta}^{g}(S_{3})$ up to $d \leq 6$. Here $\mathcal{O}(\beta)$ is the $W_{E_{6}}$ -orbit of β and $g(\beta)$ is the arithmetic genus of a curve which belongs to β which is given by $\beta . (\beta + K_{S_{3}})/2 + 1$. One can observe that $n_{\beta}^{g}(S_{3})$ are zero if $g > g(\beta)$.

6.6 Appendix: -K-nef toric surfaces

6.6.1 Classification of -K-nef toric surfaces

The following classification is due to Batyrev [5] (see also [11, Table 1]).

Lemma 6.12. There are exactly sixteen -K-nef toric surfaces, whose fans are shown in Figure 3.

We will refer the -K-nef toric surfaces using the numbers shown in frames in Figure 3.

Proof. The minimal -K-nef toric surfaces are \mathbb{P}^2 , $\mathbb{P}^1 \times \mathbb{P}^1$, and the Hirzebruch surface \mathbb{F}_2 , which are No. 1, No. 2, and No. 4 respectively. -K-nef toric surfaces are obtained from them by blowing up at a torus-fixed point successively. By the -K-nef condition, we must blow-up at a torus-fixed point which is not on a torus-fixed (-2)-curve. All possible patterns of blowing-ups are listed in Figure 3. Note that No. 13, 15, and 16 can no longer be blown-up to -K-nef toric surfaces, since all of their torus-fixed points are on a torus-fixed (-2)-curve. This completes the classification.

6.6.2 Deformation of -K-nef toric surfaces

Proposition 6.13. A -K-nef toric surface has a smooth versal deformation family of dimension $h^1(\Theta)$, whose general member is a del Pezzo surface of degree c_1^2 .

 $h^1(\Theta)$ and c_1^2 are given in Table 2.



Figure 3: Classification of -K-nef toric surfaces. The arrows indicate blow-downs. The numbers in frames are reference numbers. Note that S_3^0 , S_4^0 , and S_5^0 introduced in §6.3 are No. 16, 14, and 12, respectively.

Proof. Note that $h^2(\Theta) = 0$ for any smooth compact toric surface (Corollary 6.15). This implies smoothness of a versal deformation family [46].

Versal deformation families of -K-nef toric surfaces are constructed inductively as follows. Let $\pi: \tilde{S} \to S$ be one of the blowing-ups in Figure 3. Let $P \in S$ be the center of the blowing-up π which is the intersection of two torus-fixed curves C_1 and C_2 (see Figure 4). By comparing Table 2 with Figure 3, we have

$$h^{1}(\tilde{S},\Theta) = \begin{cases} h^{1}(S,\Theta) & \text{if } C_{1}^{2} > -1, \ C_{2}^{2} > -1, \\ h^{1}(S,\Theta) + 1 & \text{if } C_{1}^{2} = -1, \ C_{2}^{2} > -1, \\ h^{1}(S,\Theta) + 2 & \text{if } C_{1}^{2} = C_{2}^{2} = -1. \end{cases}$$
(6.9)

Since smooth rational curves on complex surfaces with self-intersection ≥ -1 is stable under small deformations [47, Example in p.86] (see also [4, IV. 3.1]), a complete deformation family of \tilde{S} can be found as a simultaneous blowing-up of a complete deformation family of S. Furthermore, by eq. (6.9), we can find a versal deformation family of \tilde{S} as follows. First, we consider a versal deformation family S of S on which C_1 and C_2 deform holomorphically. If both of C_1 and C_2 have self-intersection > -1, simultaneous blowing up of S at P gives a versal deformation family of \tilde{S} which is of dimension $h^1(S, \Theta)$. If $C_1^2 = -1$ and $C_2^2 > -1$, we move the center P in the C_2 direction (see Figure 4) and blow S up simultaneously to get a versal deformation family of \tilde{S} which is of dimension $h^1(S, \Theta) + 1$. If $C_1^2 = C_2^2 = -1$, we move the center P in the whole direction and blow S up simultaneously to get a versal deformation family of \tilde{S} which is of dimension $h^1(S, \Theta) + 1$. If $C_1^2 = C_2^2 = -1$, we move the center P in the whole direction and blow S up simultaneously to get a versal deformation family of \tilde{S} which is of dimension $h^1(S, \Theta) + 1$. If $C_1^2 = C_2^2 = -1$, we move the center P in the whole direction and blow S up simultaneously to get a versal deformation family of \tilde{S} which is of dimension $h^1(S, \Theta) + 2$.

Thus we can find versal deformation families of -K-nef toric surfaces inductively. It is easy to see that their general members are del Pezzo surfaces.

6.6.3 Unobstructedness

Let X be a smooth compact toric surface, $D := D_1 + \cdots + D_r$ be the sum of all torus invariant divisors D_1, \cdots, D_r , and $\Theta(-\log D)$ be the sheaf of germs of holomorphic vector fields with



Figure 4: The center P of a blowing-up (C_1 and C_2 are torus-fixed curves) and its moving. The left is the case with $C_1^2 = -1$, $C_2^2 \ge 0$ and the right is the case with $C_1^2 = C_2^2 = -1$.

	Ι	II	III		IV		V				VI		VII			VIII
No.	1	3	2	4	5	6	7	8	9	10	11	12	13	14	15	16
c_{1}^{2}	9	8	8		7		6				5		4			3
c_2	3	4	4		5		6				7		8			9
$-\frac{7c_1^2-5c_2}{6}$	-8	-6	-6		-4		-2				0		2			4
$h^0(\Theta)$	8	6	6	7	4	5	2	4	3	5	3	2	3	2	2	2
$h^2(\Theta)$	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0
$h^1(\Theta)$	0	0	0	1	0	1	0	2	1	3	3	2	5	4	4	6

Table 2: Eight deformation types and $h^1(\Theta) \left(= -\frac{7c_1^2 - 5c_2}{6} + h^0(\Theta) + h^2(\Theta) \right)$.

logarithmic zeros along D.

Lemma 6.14. $H^2(X, \Theta(-\log D)) = 0.$

Proof. Since $\Theta(-\log D) = \mathcal{O} \otimes_{\mathbb{Z}} N$ (cf. [73, Proposition 3.1]), where N is the 2-dimensional lattice such that the fan of X sits in $N \otimes \mathbb{R}$. $H^2(X, \Theta(-\log D)) = H^2(X, \mathcal{O} \otimes_{\mathbb{Z}} N) = H^2(X, \mathcal{O} \oplus \mathcal{O}) = 0$, since $H^2(X, \mathcal{O}) = 0$ (cf. [73, Corollary 2.8]).

Corollary 6.15. $H^{2}(X, \Theta) = 0.$

Proof. From the exact sequence (cf. [73, Theorem 3.12])

$$0 \longrightarrow \Theta(-\log D) \longrightarrow \Theta \longrightarrow \oplus_{i=1}^{r} \mathcal{O}(D_i)|_{D_i} \longrightarrow 0$$

and Lemma 6.14, we have $H^2(X, \Theta) = 0$.

Part III Topological vertex and its flop invarinace

In this part, we prove Theorem 1.6. In §7.1, the topological vertex is introduced. In §7.2, Theorem 1.5 is proved (Theorem 7.9). Some combinatorial formulas are collected in §7.3. In §8, definitions of toric Calabi–Yau threefolds and their Gromov–Witten invariants are given. Then, a method to write down their partition functions is explained. In §9.1, we study the transformations of partition functions under a flop and prove Theorem 1.6 (Corollary 9.4). As an application, we consider the canonical bundles K_S and $K_{\hat{S}}$ of a complete smooth toric surface S and its blow-up \hat{S} (§9.2), and show that Gromov–Witten invariants of K_S are obtained from those of $K_{\hat{S}}$ (Corollary 9.8). In §9.3, an example is studied and the relationship with Nekrasov's partition function is discussed.

7 Topological vertex

7.1 Topological vertex

We use the following definition of the topological vertex given in [72]. See §2.3.1 for notations on combinatorial quantities.

Definition 7.1.

$$C_{\lambda_1,\lambda_2,\lambda_3}(q) \stackrel{\text{def.}}{=} q^{\frac{1}{2}\kappa(\lambda_3)} s_{\lambda_2}(q^{\rho}) \sum_{\mu \in \mathcal{P}} s_{\lambda_1/\mu}(q^{\lambda_2^t + \rho}) s_{\lambda_3^t/\mu}(q^{\lambda_2 + \rho}) , \qquad (7.1)$$

where $s_{\mu/\nu}(q^{\mu+\rho})$ (resp. $s_{\mu}(q^{\rho})$) is the skew-Schur function with the specialization of variables:

$$s_{\mu/\nu}(x_i = q^{\mu_i - i + \frac{1}{2}})$$
 (resp. $s_{\mu}(x_i = q^{-i + \frac{1}{2}})$)

Remark 7.2. (i) We have

$$W_{\mu,\nu}(q) = q^{\frac{1}{2}\kappa(\nu)} C_{\emptyset,\mu,\nu^{t}}(q), \qquad (7.2)$$

where \emptyset is the empty partition. (See (2.21) for definition of $W_{\mu,\nu}(q)$.) (ii) In [72], it has been shown that $C_{\lambda_1,\lambda_2,\lambda_3}(q)$ is invariant under the cyclic permutation of $\lambda_1, \lambda_2, \lambda_3$. This is not manifest from the expression (7.1).

7.2 Topological vertex under flops

Take four partitions $\lambda_1, \lambda_2, \lambda_3, \lambda_4$. These will be fixed throughout the rest of §7.2. We define

$$Z_0(q,Q_0) \stackrel{\text{def.}}{=} \sum_{\mu \in \mathcal{P}} (-Q_0)^{|\mu|} C_{\lambda_1,\lambda_2,\mu^t}(q) C_{\lambda_3,\lambda_4,\mu}(q) , \qquad (7.3)$$

$$Z_0^+(q, Q_0^+) \stackrel{\text{def.}}{=} \sum_{\mu \in \mathcal{P}} (-Q_0^+)^{|\mu|} C_{\lambda_1, \mu^t, \lambda_4}(q) C_{\lambda_3, \mu, \lambda_2}(q) .$$
(7.4)

We also set

$$Z_{(-1,-1)}(q,Q) = \prod_{k=1}^{\infty} (1 - Qq^k)^k,$$
(7.5)

and

$$Z_0'(q,Q_0) \stackrel{\text{def.}}{=} \frac{Z_0(q,Q_0)}{Z_{(-1,-1)}(q,Q_0)} , \qquad Z_0^{+\prime}(q,Q_0^+) \stackrel{\text{def.}}{=} \frac{Z_0^+(q,Q_0^+)}{Z_{(-1,-1)}(q,Q_0^+)}$$

The goal of this section is to show an identity relating $Z'_0(q, Q_0)$ and $Z'_0(q, Q_0)$ under the identification $Q_0^+ = Q_0^{-1}$ (Theorem 7.9). Formulas necessary for proofs can be found in §7.3.

Remark 7.3. Let us mention the geometrical meaning of the above formal power series. $Z_{(-1,-1)}(q, Q_0)$ is the partition function of the toric Calabi–Yau threefold $\mathcal{O}_{\mathbb{P}^1}(-1) \oplus \mathcal{O}_{\mathbb{P}^1}(-1)$ (see §8.3, also [24, Theorem 3] and [80]). $Z_0(q, Q_0)$ and $Z_0^+(q, Q_0^+)$ appear as local contributions in the partition functions of toric Calabi–Yau threefolds related by a flop such that both a flopping curve and a flopped curve are rational and have normal bundles isomorphic to $\mathcal{O}_{\mathbb{P}^1}(-1) \oplus \mathcal{O}_{\mathbb{P}^1}(-1)$ (see Figure 8).

7.2.1 Individual calculations

First, we compute $Z'_0(q, Q_0)$ and $Z'_0(q, Q_0^+)$ respectively. Following [36], introduce

$$f_{\mu}(q) = \frac{q}{q-1} \sum_{i \ge 1} (q^{\mu_i - i} - q^{-i}) ,$$

$$f_{\mu,\nu}(q) = (q - 2 + q^{-1}) f_{\mu}(q) f_{\nu}(q) + f_{\mu}(q) + f_{\nu}(q)$$

,

and let $C_k(\mu,\nu)$ be the expansion coefficients in the Laurent polynomial $f_{\mu,\nu}(q)$:

$$f_{\mu,\nu}(q) = \sum_{k\in\mathbb{Z}} C_k(\mu,\nu) q^k$$

Proposition 7.4. We have

$$Z_{0}'(q,Q_{0}) = q^{\frac{1}{2}\kappa(\lambda_{2}) + \frac{1}{2}\kappa(\lambda_{4})} s_{\lambda_{1}}(q^{\rho}) s_{\lambda_{3}}(q^{\rho}) \prod_{k \in \mathbb{Z}} (1 - Q_{0}q^{k})^{C_{k}(\lambda_{1}^{t},\lambda_{3}^{t})} \\ \times \sum_{\tau} (-Q_{0})^{|\tau|} s_{\lambda_{2}^{t}/\tau^{t}}(q^{\lambda_{1}+\rho}, Q_{0}q^{-\lambda_{3}-\rho}) s_{\lambda_{4}^{t}/\tau}(q^{\lambda_{3}+\rho}, Q_{0}q^{-\lambda_{1}-\rho}) ,$$

$$(7.6)$$

$$Z_{0}^{+\prime}(q,Q_{0}^{+}) = s_{\lambda_{1}}(q^{\rho})s_{\lambda_{3}}(q^{\rho})\prod_{k\in\mathbb{Z}}(1-Q_{0}^{+}q^{k})^{C_{k}(\lambda_{1},\lambda_{3})}$$

$$\times \sum_{\tau}(-Q_{0}^{+})^{|\tau|}s_{\lambda_{2}/\tau}(q^{\lambda_{3}^{t}+\rho},Q_{0}^{+}q^{-\lambda_{1}^{t}-\rho})s_{\lambda_{4}/\tau^{t}}(q^{\lambda_{1}^{t}+\rho},Q_{0}^{+}q^{-\lambda_{3}^{t}-\rho}).$$
(7.7)

Proof. By definition (7.1) of the topological vertex, we have

$$\begin{split} Z_{0}(q,Q_{0}) &= \sum_{\mu} (-Q_{0})^{|\mu|} q^{\frac{1}{2}\kappa(\lambda_{2})} s_{\lambda_{1}}(q^{\rho}) \sum_{T} s_{\mu^{t}/T}(q^{\lambda_{1}^{t}+\rho}) s_{\lambda_{2}^{t}/T}(q^{\lambda_{1}+\rho}) \\ & q^{\frac{1}{2}\kappa(\lambda_{4})} s_{\lambda_{3}}(q^{\rho}) \sum_{T'} s_{\mu/T'}(q^{\lambda_{3}^{t}+\rho}) s_{\lambda_{4}^{t}/T'}(q^{\lambda_{3}+\rho}) \\ &= q^{\frac{1}{2}\kappa(\lambda_{2})+\frac{1}{2}\kappa(\lambda_{4})} s_{\lambda_{1}}(q^{\rho}) s_{\lambda_{3}}(q^{\rho}) \sum_{T,T'} (-Q_{0})^{|T|} s_{\lambda_{2}^{t}/T}(q^{\lambda_{1}+\rho}) s_{\lambda_{4}^{t}/T'}(q^{\lambda_{3}+\rho}) \\ & \sum_{\mu} s_{\mu^{t}/T}(-Q_{0}q^{\lambda_{1}^{t}+\rho}) s_{\mu/T'}(q^{\lambda_{3}^{t}+\rho}) \; . \end{split}$$

We perform the sum with respect to μ by using (7.13):

$$\begin{split} Z_{0}(q,Q_{0}) &= q^{\frac{1}{2}\kappa(\lambda_{2}) + \frac{1}{2}\kappa(\lambda_{4})} s_{\lambda_{1}}(q^{\rho}) s_{\lambda_{3}}(q^{\rho}) \\ &\prod_{i,j\geq 1} (1 - Q_{0}q^{h_{\lambda_{1}^{t},\lambda_{3}^{t}}(i,j)}) \sum_{\tau} s_{T^{t}/\tau}(q^{\lambda_{3}^{t}+\rho}) s_{(T')^{t}/\tau^{t}}(-Q_{0}q^{\lambda_{1}^{t}+\rho}) \\ &\sum_{T,T'} (-Q_{0})^{|T|} s_{\lambda_{2}^{t}/T}(q^{\lambda_{1}+\rho}) s_{\lambda_{4}^{t}/T'}(q^{\lambda_{3}+\rho}) \\ &= q^{\frac{1}{2}\kappa(\lambda_{2}) + \frac{1}{2}\kappa(\lambda_{4})} s_{\lambda_{1}}(q^{\rho}) s_{\lambda_{3}}(q^{\rho}) \prod_{i,j\geq 1} (1 - Q_{0}q^{h_{\lambda_{1}^{t},\lambda_{3}^{t}}(i,j)}) \\ &\sum_{\tau} (-Q_{0})^{|\tau|} \sum_{T} s_{\lambda_{2}^{t}/T}(q^{\lambda_{1}+\rho}) s_{T/\tau^{t}}(Q_{0}q^{-\lambda_{3}-\rho}) \sum_{T'} s_{\lambda_{4}^{t}/T'}(q^{\lambda_{3}+\rho}) s_{T'/\tau}(Q_{0}q^{-\lambda_{1}-\rho}) \;. \end{split}$$

Here for $\mu, \nu \in \mathcal{P}$,

$$h_{\mu,\nu}(i,j) \stackrel{\text{def.}}{=} \mu_i - i + \nu_j - j + 1$$

In passing to the second line, we have used (7.16). By using (7.14), we have

$$Z_{0}(q,Q_{0}) = q^{\frac{1}{2}\kappa(\lambda_{2}) + \frac{1}{2}\kappa(\lambda_{4})} s_{\lambda_{1}}(q^{\rho}) s_{\lambda_{3}}(q^{\rho}) \prod_{i,j \ge 1} (1 - Q_{0}q^{h_{\lambda_{1}^{t},\lambda_{3}^{t}}(i,j)})$$
$$\sum_{\tau} (-Q_{0})^{|\tau|} s_{\lambda_{2}^{t}/\tau^{t}}(q^{\lambda_{1}+\rho}, Q_{0}q^{-\lambda_{3}-\rho}) s_{\lambda_{4}^{t}/\tau}(q^{\lambda_{3}+\rho}, Q_{0}q^{-\lambda_{1}-\rho}) .$$

Applying Lemma 7.10, we obtain (7.6). One can also compute $Z_0^+(q, Q_0^+)$ in a similar way. \Box

Remark 7.5. This type of computation also appears in a proof of the large N factorization formula of the S-matrix of U(N) Chern–Simons theory. See [41].

The next corollary is a consequence of Proposition 7.4.

Corollary 7.6. $Z_0^{+\prime}(q, Q_0^+)$ is a polynomial in Q_0^+ of degree at most $|\lambda_1| + |\lambda_2| + |\lambda_3| + |\lambda_4|$. Moreover, if $\lambda_3 = \lambda_4 = \emptyset$, $Z_0^{+\prime}(q, Q_0^+)$ is a polynomial in Q_0^+ of degree $|\lambda_1| + |\lambda_2|$.

Similar statement also holds for $Z'_0(q, Q_0)$.

Proof. The first statement follows if we apply (7.10) and (7.15) to the expression (7.7). To prove the second statement, we show that the top term does not vanish. By (7.10), we have

$$\prod (1 - Q_0^+ q^k)^{C_k(\lambda_1^t, \emptyset)} = (-1)^{|\lambda|} q^{-\frac{1}{2}\kappa_{\lambda_1}} (Q_0^+)^{|\lambda_1|} + (\text{terms of lower degree in } Q_0^+).$$

Substituting this into (7.7) with λ_3, λ_4 set to \emptyset , and using (7.15), we obtain the claim.

7.2.2 Comparison

Next, we compare $Z'_0(q, Q_0)$ with $Z'_0(q, Q_0)$ under the identification $Q_0^+ = Q_0^{-1}$. First we have the following

Lemma 7.7. Under the identification $Q_0^+ = Q_0^{-1}$, we have

$$\sum_{\tau} (-Q_0^+)^{|\tau|} s_{\lambda_2/\tau} (q^{\lambda_3^t + \rho}, Q_0^+ q^{-\lambda_1^t - \rho}) s_{\lambda_4/\tau^t} (q^{\lambda_1^t + \rho}, Q_0^+ q^{-\lambda_3^t - \rho}) = (-Q_0)^{-|\lambda_2| - |\lambda_4|} \sum_{\tau} (-Q_0)^{|\tau|} s_{\lambda_2^t/\tau^t} (q^{\lambda_1 + \rho}, Q_0 q^{-\lambda_3 - \rho}) s_{\lambda_4^t/\tau} (q^{\lambda_3 + \rho}, Q_0 q^{-\lambda_1 - \rho}) .$$
(7.8)

Proof. Under $Q_0^+ = Q_0^{-1}$, we have

$$\begin{aligned} \text{(LHS)} &= \sum_{\tau} (-Q_0^{-1})^{|\tau|} s_{\lambda_2/\tau} (q^{\lambda_3^t + \rho}, Q_0^{-1} q^{-\lambda_1^t - \rho}) s_{\lambda_4/\tau^t} (q^{\lambda_1^t + \rho}, Q_0^{-1} q^{-\lambda_3^t - \rho}) \\ &= \sum_{\tau} (-Q_0^{-1})^{|\tau|} (-1)^{|\lambda_2| + |\lambda_4|} s_{\lambda_2^t/\tau^t} (q^{-\lambda_3 - \rho}, Q_0^{-1} q^{\lambda_1 + \rho}) s_{\lambda_4^t/\tau} (q^{-\lambda_1 - \rho}, Q_0^{-1} q^{\lambda_3 + \rho}) , \\ &= \sum_{\tau} (-Q_0)^{-|\lambda_2| + |\tau| - |\lambda_4| + |\tau|} (-Q_0^{-1})^{|\tau|} s_{\lambda_2^t/\tau^t} (q^{\lambda_1 + \rho}, Q_0 q^{-\lambda_3^t - \rho}) s_{\lambda_4^t/\tau} (q^{\lambda_3 + \rho}, Q_0 q^{-\lambda_1^t - \rho}) \\ &= (\text{RHS}) . \end{aligned}$$

Note that we have used the property (7.16) in the second line and the homogeneity (7.15) of skew Schur functions in the third line.

Lemma 7.8. The following identity holds:

$$\prod_{k} (1 - Q_0^{-1} q^k)^{C_k(\lambda_1, \lambda_3)} = (-Q_0)^{-|\lambda_1| - |\lambda_3|} q^{\frac{1}{2}\kappa(\lambda_1) + \frac{1}{2}\kappa(\lambda_3)} \prod_{k} (1 - Q_0 q^k)^{C_k(\lambda_1^t, \lambda_3^t)}.$$

Proof. By (7.11), we have

$$\begin{split} \prod_{k} (1 - Q_0^{-1} q^k)^{C_k(\lambda_1, \lambda_3)} &= \prod_{k} (1 - Q_0^{-1} q^{-k})^{C_k(\lambda_1^t, \lambda_3^t)} \\ &= Q_0^{-\frac{1}{2}(|\lambda_1| + |\lambda_3|)} q^{-\frac{1}{4}(\kappa(\lambda_1^t) + \kappa(\lambda_3^t))} \prod_{k} \left(Q_0^{\frac{1}{2}} q^{\frac{k}{2}} - Q_0^{-\frac{1}{2}} q^{-\frac{k}{2}} \right)^{C_k(\lambda_1^t, \lambda_3^t)} \end{split}$$

On the other hand, we have

$$\begin{split} \prod_{k} (1 - Q_{0}q^{k})^{C_{k}(\lambda_{1}^{t},\lambda_{3}^{t})} &= Q_{0}^{\frac{1}{2}(|\lambda_{1}| + |\lambda_{3}|)} q^{\frac{1}{4}(\kappa(\lambda_{1}^{t}) + \kappa(\lambda_{3}^{t}))} \prod_{k} \left(Q_{0}^{-\frac{1}{2}}q^{-\frac{k}{2}} - Q_{0}^{\frac{1}{2}}q^{\frac{k}{2}} \right)^{C_{k}(\lambda_{1}^{t},\lambda_{3}^{t})} \\ &= (-1)^{|\lambda_{1}| + |\lambda_{3}|} Q_{0}^{\frac{1}{2}(|\lambda_{1}| + |\lambda_{3}|)} q^{\frac{1}{4}(\kappa(\lambda_{1}^{t}) + \kappa(\lambda_{3}^{t}))} \prod_{k} \left(Q_{0}^{\frac{1}{2}}q^{\frac{k}{2}} - Q_{0}^{-\frac{1}{2}}q^{-\frac{k}{2}} \right)^{C_{k}(\lambda_{1}^{t},\lambda_{3}^{t})} \end{split}$$

By comparing the above two equations and by using a symmetry (7.9) of a κ -factor, we get the claim.

The following is the main result in this section 9

Theorem 7.9. Under the identification $Q_0^+ = Q_0^{-1}$, we have

$$Z_0^{+\prime}(q,Q_0^+) = (-Q_0)^{-(|\lambda_1|+|\lambda_2|+|\lambda_3|+|\lambda_4|)} q^{\frac{1}{2}(\kappa(\lambda_1)-\kappa(\lambda_2)+\kappa(\lambda_3)-\kappa(\lambda_4))} Z_0^{\prime}(q,Q_0)$$

Proof. This follows from Proposition 7.4 and Lemmas 7.7 and 7.8.

7.3 Appendix: Combinatorial formulas

We collect some combinatorial formulas which are used in §7. Our basic references are [62, 22, 72, 83]. For $\mu \in \mathcal{P}$, $\kappa(\mu)$ has the following important property:

$$\kappa(\mu^t) = -\kappa(\mu) , \qquad (7.9)$$

where μ^t denotes the conjugate partition (the partition obtained by the transposition of the Young diagram of μ).

It is known that $C_k(\mu, \nu)$ are nonzero integers for finitely many values of k ([83, Theorem 5.1]), and have the following properties ([22, §3.1], [83, §5.3]):

$$\sum_{k} C_{k}(\mu, \nu) = |\mu| + |\nu| , \qquad \sum_{k} k C_{k}(\mu, \nu) = \frac{1}{2} (\kappa(\mu) + \kappa(\nu)) , \qquad (7.10)$$

$$C_k(\mu,\nu) = C_{-k}(\mu^t,\nu^t) .$$
(7.11)

The following lemma is proved in [22, Lemma in §C], [83, Proposition 6.1]:

⁹ The case of $\lambda_1 = \lambda_4 = \emptyset$ was proved by Hiroaki Kanno in his unpublished note in January 2004. The author is grateful to him for kindly providing the note.

Lemma 7.10. For $\mu, \nu \in \mathcal{P}$, the following identity holds:

$$\prod_{i,j\geq 1} \left(1 - Qq^{h_{\mu,\nu}(i,j)}\right) = Z_{(-1,-1)}(q,Q) \prod_k \left(1 - Qq^k\right)^{C_k(\mu,\nu)}$$

Here are some properties of skew-Schur function. The following formulas are useful in performing the summations over partitions ([62, p.93,(5.10)]). Let $x = (x_1, x_2, ...)$ and $y = (y_1, y_2, ...)$ be sets of variables and $(x, y) = (x_1, x_2, ..., y_1, y_2, ...)$.

$$\sum_{\lambda \in \mathcal{P}} s_{\lambda/\lambda_1}(x) s_{\lambda/\lambda_2}(y) = \prod_{i,j \ge 1} (1 - x_i y_j)^{-1} \sum_{\mu \in \mathcal{P}} s_{\lambda_2/\mu}(x) s_{\lambda_1/\mu}(y) , \qquad (7.12)$$

$$\sum_{\lambda \in \mathcal{P}} s_{\lambda/\lambda_1}(x) s_{\lambda^t/\lambda_2}(y) = \prod_{i,j \ge 1} (1 + x_i y_j) \sum_{\mu \in \mathcal{P}} s_{\lambda_2^t/\mu}(x) s_{\lambda_1^t/\mu^t}(y) , \qquad (7.13)$$

$$\sum_{\xi \in \mathcal{P}} s_{\mu/\xi}(x) s_{\xi/\nu}(y) = s_{\mu/\nu}(x, y) .$$
(7.14)

Other properties are ([83, Proposition 4.1]):

$$s_{\mu/\nu}(Qx) = Q^{|\mu| - |\nu|} s_{\mu/\nu}(x) , \qquad (7.15)$$

where $Qx = (Qx_1, Qx_2, \dots)$.

$$s_{\lambda/\mu}(q^{\nu+\rho}) = (-1)^{|\lambda|-|\mu|} s_{\lambda^t/\mu^t}(q^{-\nu^t-\rho}) .$$
(7.16)

8 Toric Calabi–Yau threefolds and Partition functions

8.1 Toric Calabi–Yau threefolds

In this section, we give definitions of toric Calabi–Yau threefolds and their partition functions. Basic references for toric varieties are [27, 73]. We follow [49] for a definition of toric Calabi–Yau threefolds and their partition functions.

8.1.1 Toric Calabi–Yau threefolds

Definition 8.1. A toric Calabi–Yau threefold is a three-dimensional smooth toric variety X associated with a finite fan Σ satisfying following conditions:

- (i) the primitive generator $\vec{\omega}$ of every 1-cone satisfies $\vec{\omega} \cdot \vec{u} = 1$ where $\vec{u} = (0, 0, 1)$;
- (ii) all maximal cones are three dimensional;
- (iii) $|\Sigma| \cap \{z = 1\}$ is connected, where $|\Sigma| = \bigcup_{\sigma \in \Sigma} \sigma \subset \mathbb{R}^3$ is the support of Σ and z is the third coordinate of \mathbb{R}^3 .

The condition (i) is equivalent to the condition that K_X is trivial (the Calabi–Yau condition) and the condition (ii) implies that $\pi_1(X) = 0$, $A_2(X) \cong \operatorname{Pic}(X) \cong H^2(X,\mathbb{Z})$ and they are torsion free, where $A_2(X)$ is the group of all Weil divisors modulo rational equivalence. (cf.[27, §3.4]). The condition (iii) will be used to define a certain connected graph in §8.2.1. Note that Σ cannot be complete by the condition (i). Hence a toric Calabi–Yau threefold X is never compact.

Let us give an example of a toric Calabi–Yau threefold. Let Δ be the two dimensional nonsingular complete fan, which defines a toric surface S. Let $\{\rho_1, \ldots, \rho_r\}$ be the set of 1cones of Δ and v_i be the primitive lattice vector generating ρ_i . For Δ , we associate a three dimensional fan Σ whose three-dimensional cone is generated by $(v_i, 1) \in \mathbb{R}^3$. Then the toric Calabi–Yau threefold associated to Σ is the total space of the canonical bundle K_S of S (*a local toric surface*). Another example is the total space of a rank 2 vector bundle over \mathbb{P}^1 with degree -2 (*a local toric curve*). More complicated examples can be found in §9.3.

8.1.2 Intersection theory

We briefly describe necessary facts on intersection theory on a toric Calabi–Yau threefold X. Recall that the subset $\Sigma_n \subset \Sigma$ of *n*-cones is in one-to-one correspondence with the set of (3 - n)-dimensional torus invariant subvarieties in X. Let $\Sigma_1 = \{\rho_1, \ldots, \rho_r\}$ be the set of 1-cones. Denote by $D_{\rho_i} \subset X$ $(1 \leq i \leq r)$ the torus invariant Weil divisor corresponding to ρ_i . Let Σ'_2 be the set of 2-cones which lie in the interior of $|\Sigma|$:

$$\Sigma_2' = \{ \tau \in \Sigma_2 | \tau \subset |\Sigma| \setminus \partial |\Sigma| \}.$$

It is in one-to-one correspondence with the set of torus invariant (hence rational) curves in X. Let us write $\Sigma'_2 = \{\tau_1, \ldots, \tau_p\}$ and let $C_{\tau_i} \subset X$ denote the rational curve corresponding to τ_i . We consider the following additive groups

$$TZ^1(X) := \bigoplus_{\rho \in \Sigma_1} \mathbb{Z}D_{\rho}, \quad TZ_1^c(X) := \bigoplus_{\tau \in \Sigma_2'} \mathbb{Z}C_{\tau}.$$

There is a Z-bilinear paring

$$TZ^1(X) \times TZ_1^c(X) \to \mathbb{Z},$$
(8.1)

induced by intersection numbers between D_{ρ_i} 's and C_{τ_i} 's. Numerical equivalence relations on $TZ^1(X)$ and $TZ_1^c(X)$ are defined as follows. For $D \in TZ^1(X)$, $D \equiv 0$ if D.C = 0 for any $C \in TZ_1^c(X)$. For $C \in TZ_1^c(X)$, $C \equiv 0$ if D.C = 0 for any $D \in TZ^1(X)$. We define

$$TN^{1}(X) := TZ^{1}(X) / \equiv, \quad TN_{1}^{c}(X) := TZ_{1}^{c}(X) / \equiv .$$

Note that $TN^1(X)_{\mathbb{R}} := TN^1(X) \otimes_{\mathbb{Z}} \mathbb{R}$ and $TN_1^c(X)_{\mathbb{R}} := TN_1^c(X) \otimes_{\mathbb{Z}} \mathbb{R}$ are dual to each other by the paring (8.1).

Let us explain how to obtain the numerical equivalence relations. By the condition (ii) in Definition 8.1, $TN^1(X)$ is isomorphic to $A_2(X)$. Therefore numerical equivalence relations are given by

$$\sum_{j=1}^{r} a_{ij} D_{\rho_j} = 0, \quad (i = 1, 2, 3),$$

where $A = (a_{ij})$ is the $3 \times r$ matrix

$$A = \left(\vec{\omega}_1, \ldots, \vec{\omega}_r\right),\,$$

and $\vec{\omega}_i$ $(1 \leq i \leq r)$ are the primitive lattice vector generating ρ_i . Let us introduce the following injective map

$$l_X : TN_1^c(X) \to \{l \in \mathbb{Z}^r \,|\, A.l = \vec{0}\}, \quad Z \mapsto (D_{\rho_1}.Z, \dots, D_{\rho_r}.Z).$$
 (8.2)

Then numerical equivalence relations on $TN_1^c(X)$ can be read from linear relations between the lattice vectors $l_X([C_{\tau_1}]), \ldots, l_X([C_{\tau_p}])$.

Remark 8.2. Intersection numbers between D_{ρ_i} 's and C_{τ_i} 's can be computed as follows. First, if ρ_j and τ_i spans a 3-cone, $D_{\rho_j}.C_{\tau_i} = 1$ and if ρ_j and τ_i do not span a cone in the fan, $D_{\rho_j}.C_{\tau_i} = 0$ (cf.[27, §5.1]). If two 1-cones, say ρ_1, ρ_2 , are contained in τ_i , then $D_{\rho_1}.C_{\tau_i}$ and $D_{\rho_2}.C_{\tau_i}$ are obtained by solving the equation $A.l_X([C_{\tau_i}]) = \vec{0}$. In particular, they satisfy the relation $D_{\rho_1}.C_{\tau_i} + D_{\rho_2}.C_{\tau_i} = -2$ by the condition (i) in Definition 8.1.

Let $\tau \in \Sigma'_2$ and $\rho_1, \rho_2 \in \Sigma_1$ be as above. By looking at the gluing of local coordinate systems around C_{τ} , we see that its normal bundle is isomorphic to $\mathcal{O}_{\mathbb{P}^1}(D_{\rho_1}.C_{\tau}) \oplus \mathcal{O}_{\mathbb{P}^1}(D_{\rho_2}.C_{\tau})$. We call C_{τ} a (-1, -1)-curve if its normal bundle is isomorphic to $\mathcal{O}_{\mathbb{P}^1}(-1) \oplus \mathcal{O}_{\mathbb{P}^1}(-1)$.

8.2 Partition functions

Let X be a toric Calabi–Yau threefold and Σ be its fan. We explain how to write down the partition function of X.

8.2.1 Toric graphs

First, consider the following directed graph Γ_X (called *a toric graph*) with labels on edges of a certain type. The vertex set is

$$V(\Gamma_X) = V_3(\Gamma_X) \cup V_1(\Gamma_X),$$

where

$$V_3(\Gamma_X) = \{ v_\sigma | \sigma \in \Sigma_3(X) \}, \quad V_1(\Gamma_X) = \{ v_\tau | \tau \in \Sigma_2(X) \setminus \Sigma_2'(X) \}.$$

The edge set is

$$E(\Gamma_X) = E_3(\Gamma_X) \cup E_1(\Gamma_X),$$

where

$$E_3(\Gamma_X) = \{e_\tau | \tau \in \Sigma'_2(X)\}, \quad E_1(\Gamma_X) = \{e_\tau | \tau \in \Sigma_2(X) \setminus \Sigma'_2(X)\}.$$

An edge $e_{\tau} \in E_3(\Gamma_X)$ joins $v_{\sigma}, v_{\sigma'} \in V_3(\Gamma)$ if and only if $\tau = \sigma \cap \sigma'$ (see Figure 5) and an edge $e_{\tau} \in E_1(\Gamma)$ joins $v_{\sigma} \in V_3(\Gamma_X)$ and $v_{\tau} \in V_1(\Gamma_X)$ if and only if σ is a unique 3-cone such that τ is a face of σ . This defines a finite planner graph. Note that $V_3(\Gamma_X) \neq \emptyset$ by the condition (ii) in Definition 8.1. A vertex in $V_3(\Gamma_X)$ is trivalent and a vertex in $V_1(\Gamma_X)$ is univalent. A graph Γ_X is connected by the condition (iii) in Definition 8.1.

The direction of edges can be taken arbitrarily. The label $n : E_3(\Gamma_X) \to \mathbb{Z}$, called the *framing*, is given as follows:

$$n(e_{\tau}) = \frac{D_{\rho_1} \cdot C_{\tau} - D_{\rho_2} \cdot C_{\tau}}{2}$$

where $\tau \in \Sigma'_2$ and $\rho_1, \rho_2 \in \Sigma_1$ are as shown in Figure 5. Note that C_{τ} is a (-1, -1)-curve if and only if $n(e_{\tau}) = 0$.

Let us consider the case of a local toric surface $X = K_S$. Its toric graph is a cyclic graph formed by $V_3(\Gamma_X)$ and $E_3(\Gamma_X)$ together with exterior vertices $V_1(\Gamma_X)$ and edges $E_1(\Gamma_X)$ (see Figure 6). Its inner edges $E_3(\Gamma_X)$ come from torus invariant curves C in S and their framings are given by $C^2 - 1$, where C^2 is the self-intersection number of C in S, if we take the clockwise orientations on $E_3(\Gamma_X)$ as in Figure 6.



Figure 5: Fan (section at z = 1) and toric graph



Figure 6: Fan (section at z = 1) (left) and toric graph (right) of $K_{\mathbb{P}^1 \times \mathbb{P}^1}$. Numbers attached to $E_3(\Gamma_X)$ are framings with respect to the orientations given in the figure.

8.2.2 Partition functions

Secondly, we write down the partition function from Γ_X . Let

$$\mathcal{P}(\Gamma_X) = \{ \vec{\lambda} : E_3(\Gamma) \to \mathcal{P} \}$$

Take the set of formal variables $\vec{Q} = (Q_e)_{e \in E_3(\Gamma_X)}$ associated to $E_3(\Gamma_X)$. Then the partition function of X is a formal power series in \vec{Q} given by

$$Z_X(q,\vec{Q}) = \sum_{\vec{\lambda}\in\mathcal{P}(\Gamma_X)} \prod_{e\in E_3(\Gamma_X)} (-1)^{|\vec{\lambda}(e)|(n_e+1)} q^{\frac{\kappa(\vec{\lambda}(e))}{2}n(e)} Q_e^{|\vec{\lambda}(e)|} \prod_{v\in V_3(\Gamma_X)} C_{\vec{\lambda}_v}(q) .$$
(8.3)

Here $C_{\vec{\lambda}_v}(q)$ is the topological vertex defined in (7.1) and $\vec{\lambda}_v$ $(v \in V_3(\Gamma_X), \vec{\lambda} \in \mathcal{P}(\Gamma_X))$ is as in Figure 7 (for $e \in E(\Gamma_X) \setminus E_3(\Gamma_X)$, set $\vec{\lambda}(e)$ to \emptyset). We remark that the partition function does not depend on the directions of edges since the framing changes the sign if one gives the opposite direction to an edge $e \in E_3(\Gamma_X)$ and it is compensated by (7.9) and the summation.

Remark 8.3. By Remark 7.2 (i) and the last paragraph in $\S8.2.1$, it is immediate to see that the formula (4.7) can be recovered from the formula (8.3) for local toric surfaces.

8.3 Gromov–Witten invariants of toric Calabi–Yau threefolds

Now we explain how to obtain Gromov–Witten invariants of a toric Calabi–Yau threefold X from the partition function $Z_X(q, \vec{Q})$. First, we give a precise definition of Gromov–Witten invariants for toric Calabi–Yau threefolds.



Figure 7: $\vec{\lambda}_v$

8.3.1 Gromov–Witten invariants of toric Calabi–Yau threefolds

Heuristically, Gromov-Witten invariants of X is defined as

$$\int_{[\overline{\mathcal{M}}_{g,0}(X,\beta)]^{\mathrm{vir}}} 1 \ . \tag{8.4}$$

However the integral is in general not well-defined, since X is non-compact. A rigorous definition of Gromov–Witten invariants for toric Calabi–Yau threefolds has been given in [59], which we now explain. The basic idea is to *define* the integral (8.4) as an equivariant integral by using the virtual localization formula (2.13). Recall that the 3-dimensional algebraic torus $(\mathbb{C}^*)^3$ acts on X. There is a distinguished 2-dimensional subtorus $\mathbb{T} \subset (\mathbb{C}^*)^3$ defined as follows. Let $X^{(\mathbb{C}^*)^3}$ be the fixed point set of the $(\mathbb{C}^*)^3$ -action, which is nonempty by the condition (ii) in Definition 8.1. Let $p \in X^{(\mathbb{C}^*)^3}$ be a fixed point. Then $(\mathbb{C}^*)^3$ acts on T_pX and $\wedge^3 T_pX$, where T_pX is the tangent space at p. The action of $(\mathbb{C}^*)^3$ on the complex line $\wedge^3 T_pX$ gives an irreducible character

$$\alpha: (\mathbb{C}^*)^3 \to \mathbb{C}^*.$$

By conditions (i) and (iii) in Definition 8.1, the character α is independent of the choice of p. We define $\mathbb{T} := \text{Ker } \alpha \cong (\mathbb{C}^*)^2$. Then \mathbb{T} acts on X preserving the holomorphic 3-form on X (the *equivariantly Calabi–Yau* condition). The following is a generalization of Gromov–Witten invariants of local toric surfaces (cf. §4).

Definition 8.4. We define the Gromov–Witten invariant $N_{g,\beta}(X)$ of X with genus g, degree $\beta \neq 0$ as follows :

$$N_{g,\beta}(X) := \int_{[\overline{\mathcal{M}}_{g,0}(X,\beta)^{\mathbb{T}}]^{vir}} \frac{1}{e_{\mathbb{T}}(N^{vir})},\tag{8.5}$$

where the RHS is a T-equivariant integral and $e_{\mathbb{T}}(N^{vir})$ is the T-equivariant Euler class of the virtual normal bundle N^{vir} of $\overline{\mathcal{M}}_{g,0}(X,\beta)^{\mathbb{T}}$ (cf. §2.2.3).

A priori, $N_{g,\beta}(X)$ takes value in $\mathbb{Q}(\alpha_1, \alpha_2)$, where α_1, α_2 are the weights of the standard representations of \mathbb{T} . However, it has been shown in [59, Theorem 5.9] that $N_{g,\beta}(X) \in \mathbb{Q}$, which is a nontrivial consequence from the equivariantly Calabi–Yau condition.

For a toric Calabi–Yau threefold X, the Gopakumar–Vafa invariants $n_{\beta}^{g}(X)$ are defined in terms of Gromov–Witten invariants as in Definition 4.7. It is worth mentioning that Konishi [49] has shown that the same statement as in Theorem 4.8 holds for any toric Calabi–Yau threefolds.



Figure 8: Fans (sections at z = 1): Σ (left), Σ_0 (middle) and Σ^+ (right). The generators $\vec{\omega}_1, \ldots, \vec{\omega}_4$ of ρ_1, \ldots, ρ_4 satisfy the relation $\vec{\omega}_1 + \vec{\omega}_3 = \vec{\omega}_2 + \vec{\omega}_4$.

8.3.2 Partition functions and Gromov–Witten invariants

The claim of [2, 59] is that the Gromov–Witten invariant $N_{g,\beta}(X)$ of X with the genus g and the degree β is obtained form $Z_X(q, \vec{Q})$ as follows:

$$\sum_{g \ge 0} N_{g,\beta}(X) \lambda^{2g-2} = \sum_{\substack{\vec{d} = (d_e)_{e \in E_3(\Gamma_X)}, \\ \vec{d} \mid \vec{C} \mid = \mid \beta \mid}} F_{\vec{d}}(e^{\sqrt{-1}\lambda}),$$
(8.6)

where $[\vec{C}] = ([C_e])_{e \in E_3(\Gamma_X)}, C_e \subset X$ is the rational curve corresponding to $e, \vec{d}[\vec{C}] := \sum_{e \in E_3(\Gamma_X)} d_e[C_e]$, and $F_{\vec{d}}(q)$ is given by

$$\log Z_X(q,\vec{Q}) = \sum_{\vec{d}=(d_e)_{e\in E_3(\Gamma_X)}} F_{\vec{d}}(q)\vec{Q}^{\vec{d}},$$

where $\vec{Q}^{\vec{d}} = \prod_{e \in E_3(\Gamma_X)} Q_e^{d_e}$.

The formula (8.6) can be explained as follows. Since \mathbb{T} and $(\mathbb{C}^*)^3$ have the same 0- and 1-dimensional orbits, their induced actions on $\overline{\mathcal{M}}_{g,0}(X,\beta)$ has the same fixed point set. One can carry out virtual localization as in the case with local toric surfaces in §5. This time, one encounters three-partition special triple Hodge integrals (cf. §2.3), which give rise to the topological vertex $C_{\vec{\lambda}_v}(q)$. This was first observed in [16], and almost established in [59] (see the remark below).

Remark 8.5. Precisely speaking, the partition function obtained in [59] has the expression almost same as (8.3) except that $C_{\vec{\lambda}_v}(q)$ is replaced by $\tilde{\mathcal{W}}_{\vec{\lambda}_v}(q)$. Here $\tilde{\mathcal{W}}_{\lambda_1,\lambda_2,\lambda_3}(q)$ is a rational function in $q^{\frac{1}{2}}$ similar to $C_{\lambda_1,\lambda_2,\lambda_3}(q)$ but has a slightly different expression. It is conjectured that $\tilde{\mathcal{W}}_{\lambda_1,\lambda_2,\lambda_3}(q) = C_{\lambda_1,\lambda_2,\lambda_3}(q)$ [59, Conjecture 8.3]. Here we use $C_{\lambda_1,\lambda_2,\lambda_3}(q)$ assuming that the conjecture is true. (The conjecture is true if at least one λ_i is empty.)

9 Transformations of partition functions

9.1 Flop invariance

In this section, we study the transformation of the partition function of toric Calabi–Yau threefolds under a flop.

9.1.1 Flop

Let X be a toric Calabi–Yau threefold and let Σ be its fan. Assume that X contains at least one (-1, -1)-curve C_0 . Denote the corresponding 2-cone by τ_0 . Near τ_0 , the fan looks like the

	X		X^+	
2-cone	$ au_0, au_1, \dots, au_4$	au	$ au_0^+, au_1,\ldots, au_4$	au
curve	C_0, C_1, \ldots, C_4	C_{τ}	$C_0^+, C_1^+, \dots, C_4^+$	C_{τ}^+
edge	e_0, e_1, \ldots, e_4	e_{τ} or just e	$e_0^+, e_1^+, \dots, e_4^+$	e_{τ} or just e
variable	Q_0, Q_1, \ldots, Q_4	Q_e	$Q_0^+, Q_1^+, \dots, Q_4^+$	Q_e
	I	Table 3:		
$e_3 v_2$	$e_0 \qquad v_1 \qquad e_1 \\ e_0 \qquad n(e_0) = 0$	<u>e</u>	$e_{2}^{\top} n(e_{2}^{+}) = n$ $(e_{3}^{+}) = n$ $v_{2}^{+} n(e_{3}^{+}) = n$ $v_{2}^{+} n(e_{0}^{+})$ $v_{1}^{+} n$ $e_{4}^{+} n$	$(e_{2}) + 1$ $(e_{3}) - 1$ $= 0$ $e_{1}^{e_{1}^{+}}$ $e_{1}^{e_{1}^{+}} = n(e_{1}) - e_{1}^{e_{1}^{+}} = n(e_{4}) + e_{4}^{e_{1}^{+}}$

Figure 9: Toric graphs Γ_X (left) and Γ_{X^+} (right).

left diagram in Figure 8. We set

$$\Sigma_0 = (\Sigma \setminus \{\tau_0, \sigma_1, \sigma_2\}) \cup \{\sigma_0\}, \quad \Sigma^+ = (\Sigma \setminus \{\tau_0, \sigma_1, \sigma_2\}) \cup \{\tau_0^+, \sigma_1^+, \sigma_2^+\}$$

where $\tau_0, \sigma_1, \sigma_2, \sigma_0, \tau_0^+ \sigma_1^+, \sigma_2^+$ are cones shown in Figure 8. Let X_0 be the singular toric variety associated with the fan Σ_0 and X^+ be the toric Calabi–Yau threefold associated with the fan Σ^+ . We denote by C_0^+ the (-1, -1)-curve on X corresponding to τ_0^+ . Then associated to the evident maps $\Sigma \to \Sigma_0$ and $\Sigma^+ \to \Sigma_0$, there are the following birational maps:

$$\begin{array}{cccc} X & \stackrel{- \bullet}{\dashrightarrow} & X^+ \\ f \searrow & \swarrow f^+ \\ & X_0 \end{array}$$

The maps f, f^+ are small contractions whose exceptional sets are C_0 , C_0^+ respectively. The birational map $\phi = (f^+)^{-1} \circ f$ is called the flop with respect to C_0 . Note that ϕ is an isomorphism in codimension 1. Therefore, there is a canonical isomorphism $TN^1(X) \cong TN^1(X^+)$ induced by ϕ . In turn, this induces an isomorphism $\phi_*: TN_1^c(X)_{\mathbb{R}} \to TN_1^c(X^+)_{\mathbb{R}}$ via intersection paring.

 ϕ . In turn, this induces an isomorphism $\phi_* : TN_1^c(X)_{\mathbb{R}} \to TN_1^c(X^+)_{\mathbb{R}}$ via intersection paring. From here on, we proceeds assuming that $\tau_1, \ldots, \tau_4 \in \Sigma'_2$.¹⁰ We use the notations shown in Table 3.

Lemma 9.1. Under the flop $\phi : X \dashrightarrow X^+$, the curve classes transform as follows.

$$\phi_*[C_0] = -[C_0^+], \quad \phi_*[C_i] = [C_i^+] + [C_0^+], \quad \phi_*[C_\tau] = [C_\tau^+] \quad \text{for } \tau \in \Sigma_2'(X) \setminus \{\tau_0, \dots, \tau_4\}.$$

Proof. This follows from computations of intersection numbers between curves and divisors. The first statement follows from $l_X([C_0]) = -l_{X^+}([C_0^+])$. Proofs of the others are similar. \Box

Let Γ_X be a toric graph of X. Near the edge e_0 , the graph looks like the left diagram in Figure 9. Under the flop ϕ , the toric diagram (and the framings) changes as follows.

¹⁰ For other cases, results on partition functions can be recovered by setting to zero the formal variables associated to any of τ_1, \ldots, τ_4 which are not in Σ'_2 .

Lemma 9.2. A graph obtained from Γ_X by replacing the left diagram in Figure 9 with the right is a toric graph of X^+ .

9.1.2 Transformation of partition function

We associate the same formal variables $\vec{Q} = (Q_e)$ to edges in $E_3(\Gamma_X) \setminus \{e_0, \ldots, e_4\}$ and those in $E_3(\Gamma_{X^+}) \setminus \{e_0^+, \ldots, e_4^+\}$ and write the partition functions of X and X^+ as $Z_X(q, \vec{Q}, Q_0, Q_1, Q_2, Q_3, Q_4)$ and $Z_{X^+}(q, \vec{Q}, Q_0^+, Q_1^+, Q_2^+, Q_3^+, Q_4^+)$ respectively. It is immediate to check that

$$Z_X(q, \vec{0}, Q_0, 0, 0, 0, 0) = Z_{(-1, -1)}(q, Q_0),$$

$$Z_{X^+}(q, \vec{0}, Q_0^+, 0, 0, 0, 0) = Z_{(-1, -1)}(q, Q_0^+).$$
(9.1)

We set

$$Z'_X(q, \vec{Q}, Q_0, Q_1, Q_2, Q_3, Q_4) \stackrel{\text{def.}}{=} \frac{Z_X(q, \vec{Q}, Q_0, Q_1, Q_2, Q_3, Q_4)}{Z_X(q, \vec{0}, Q_0, 0, 0, 0, 0)} ,$$
$$Z'_{X^+}(q, \vec{Q}, Q_0^+, Q_1^+, Q_2^+, Q_3^+, Q_4^+) \stackrel{\text{def.}}{=} \frac{Z_{X^+}(q, \vec{Q}, Q_0^+, Q_1^+, Q_2^+, Q_3^+, Q_4^+)}{Z_{X^+}(q, \vec{0}, Q_0^+, 0, 0, 0, 0)}$$

Now we will compare these. To do so, we should identify the formal variables so that the identification is compatible with Lemma 9.1:

$$Q_0 = (Q_0^+)^{-1}, \qquad Q_i = Q_0^+ Q_i^+.$$

Theorem 9.3. (i) The coefficients of $\vec{Q}^{\vec{d}}Q_0^{d_0}Q_1^{d_1}\dots Q_4^{d_4}$ in $Z'_X(q, \vec{Q}, Q_0, Q_1, Q_2, Q_3, Q_4)$ is zero if $d_0 > d_1 + d_2 + d_3 + d_4$. A similar result holds for X^+ .

(ii) Under the identification $Q_0 = (Q_0^+)^{-1}$, $Q_i = Q_0^+ Q_i^+$, we have

$$Z'_X(q, \vec{Q}, Q_0, Q_1, Q_2, Q_3, Q_4) = Z'_{X^+}(q, \vec{Q}, Q_0^+, Q_1^+, Q_2^+, Q_3^+, Q_4^+)$$
.

(This is an equality between two formal power series in $\vec{Q}, Q_0^+, \ldots, Q_4^+$.)

Proof. The statement (i) follows from the first statement of Corollary 7.6.

To see (ii), let

$$\mathcal{P}'(\Gamma_X) = \{ \vec{\nu} : E_3(\Gamma_X) \setminus \{e_0\} \to \mathcal{P} \}$$

and define $\vec{\nu}_v \in \mathcal{P}^3$ for $\vec{\nu} \in \mathcal{P}'(\Gamma_X)$ and $v \in V_3(\Gamma_X) \setminus \{v_1, v_2\}$ in the same way as $\vec{\lambda}_v$ (Figure 7). After (8.3), $Z_X(q, \vec{Q}, Q_0, Q_1, Q_2, Q_3, Q_4)$ is written as follows:

$$Z_{X}(q, \vec{Q}, Q_{0}, Q_{1}, Q_{2}, Q_{3}, Q_{4}) = \sum_{\vec{\nu} \in \mathcal{P}'(\Gamma_{X})} \prod_{e \in E_{3}(\Gamma_{X}) \setminus \{e_{0}, \dots, e_{4}\}} (-1)^{(n(e)+1)|\vec{\nu}(e)|} Q_{e}^{|\vec{\nu}(e)|} \prod_{v \in V_{3}(\Gamma_{X}) \setminus \{v_{1}, v_{2}\}} C_{\vec{\nu}_{v}}(q) \times \underbrace{\prod_{i=1}^{4} (-1)^{(n(e_{i})+1)|\vec{\nu}(e_{i})|} Q_{i}^{|\vec{\nu}(e_{i})|}}_{(a)} \sum_{\mu \in \mathcal{P}} C_{\vec{\nu}(e_{1}), \vec{\nu}(e_{2}), \mu^{t}}(q) C_{\vec{\nu}(e_{3}), \vec{\nu}(e_{4}), \mu}(q) (-Q_{0})^{|\mu|}}_{(a)}$$

Similarly, $Z_{X^+}(q, \vec{Q}, Q_0^+, Q_1^+, Q_2^+, Q_3^+, Q_4^+)$ is written as follows:

$$\begin{split} &Z_{X^{+}}(q,Q,Q_{0}^{+},Q_{1}^{+},Q_{2}^{+},Q_{3}^{+},Q_{4}^{+}) \\ &= \sum_{\vec{\nu}\in\mathcal{P}'(\Gamma_{X^{+}})} \prod_{e\in E_{3}(\Gamma_{X^{+}})\setminus\{e_{0}^{+},\ldots,e_{4}^{+}\}} (-1)^{(n(e)+1)|\vec{\nu}(e)|} Q_{e}^{|\vec{\nu}(e)|} \prod_{v\in V_{3}(\Gamma_{X^{+}})\setminus\{v_{1}^{+},v_{2}^{+}\}} C_{\vec{\nu}_{v}}(q) \\ &\times \underbrace{\prod_{i=1}^{4} (-1)^{(n(e_{i}^{+})+1)|\vec{\nu}(e_{i}^{+})|} (Q_{i}^{+})^{|\vec{\nu}(e_{i}^{+})|} \sum_{\mu\in\mathcal{P}} C_{\vec{\nu}(e_{1}^{+}),\mu^{t},\vec{\nu}(e_{4}^{+})}(q) C_{\vec{\nu}(e_{3}^{+}),\mu,\vec{\nu}(e_{2}^{+})}(q) (-Q_{0}^{+})^{|\mu|}}}_{(b)} \end{split}$$

Here

$$\mathcal{P}'(\Gamma_{X^+}) = \{ \vec{\nu} : E_3(\Gamma_{X^+}) \setminus \{e_0^+\} \to \mathcal{P} \}$$

and for $\vec{\nu} \in \mathcal{P}'(\Gamma_{X^+})$ and $v \in V_3(\Gamma_{X^+}) \setminus \{v_1^+, v_2^+\}, \ \vec{\nu}_v \in \mathcal{P}^3$ is defined in the same way.

Since Γ_X and Γ_{X^+} are identical outside the diagrams described in Figure 9, $E_3(\Gamma_X) \setminus \{e_0, \ldots, e_4\} = E_3(\Gamma_{X^+}) \setminus \{e_0^+, \ldots, e_4^+\}, V_3(\Gamma_X) \setminus \{v_1, v_2\} = V_3(\Gamma_{X^+}) \setminus \{v_1^+, v_2^+\}$ and we have a natural bijection $p : \mathcal{P}'(\Gamma_X) \to \mathcal{P}'(\Gamma_{X^+})$ such that $p(\vec{\nu}) = \vec{\nu}^+$ iff $\vec{\nu}(e) = \vec{\nu}^+(e)$ for all $e \in E_3(\Gamma_X) \setminus \{e_0, \ldots, e_4\}$ and $\vec{\nu}(e_i) = \vec{\nu}^+(e_i^+)$ for $1 \le i \le 4$. Under this identification, we could see that the two partition functions have the same expressions except for the factors (a)and (b). Taking into account the change in framings, we have

$$\frac{(a)}{Z_{(-1,-1)}(q,Q_0)} \Big|_{Q_0 = (Q_0^+)^{-1}, Q_i = Q_0^+ Q_i^+} = \frac{(b)}{Z_{(-1,-1)}(q,Q_0^+)}$$

by Theorem 7.9.

We finish this subsection by restating Theorem 9.3 in terms of Gromov–Witten invariants. Corollary 9.4. For $\beta \in H_2(X, \mathbb{Z})$ such that β is not a multiple of $[C_0]$,

$$N_{g,\phi_*(\beta)}(X^+) = N_{g,\beta}(X).$$

Moreover,

$$N_{g,d[C_0]}(X) = N_{g,d[C_0^+]}(X^+) = N_{g,d[\mathbb{P}^1]}(\mathcal{O}_{\mathbb{P}^1}(-1) \oplus \mathcal{O}_{\mathbb{P}^1}(-1)).$$

The same results also hold for the Gopakumar-Vafa invariants.

Proof. Theorem 9.3 implies that $\log Z_X(q, \vec{Q}, Q_0, \ldots, Q_4)$ and $\log Z_{X^+}(q, \vec{Q}, Q_0^+, \ldots, Q_4^+)$ are written in the following form:

$$\log Z'_X(q, \vec{Q}, Q_0, \dots, Q_4) = \sum_{\vec{d}} \sum_{\substack{d_0, \dots, d_4 \ge 0, \\ d_1 + \dots + d_4 \ge d_0}} F_{\vec{d}, d_0, d_1, d_2, d_3, d_4}(q) \vec{Q}^{\vec{d}} Q_0^{d_0} \dots Q_4^{d_4},$$

$$\log Z'_{X^+}(q, \vec{Q}, Q_0^+, \dots, Q_4^+) = \sum_{\vec{d}} \sum_{\substack{d_0, \dots, d_4 \ge 0, \\ d_1 + \dots + d_4 \ge d_0}} F_{\vec{d}, d_0, d_1, d_2, d_3, d_4}(q) \vec{Q}^{\vec{d}} (Q_0^+)^{d_1 + \dots + d_4 - d_0} (Q_1^+)^{d_1} \dots (Q_4^+)^{d_4}.$$

Comparing with (8.6), we obtain the first statement. The second statement follows from (9.1). \Box

Remark 9.5. In [38, §4.1], Iqbal and Kashani–Poor studied the special case such that the 2-cones $\tau_2, \tau_4 \notin \Sigma'_2$ and the curves C_1, C_3 have normal bundles $\mathcal{O}_{\mathbb{P}^1}(-1) \oplus \mathcal{O}_{\mathbb{P}}(-1)$ or $\mathcal{O}_{\mathbb{P}^1}(-2) \oplus \mathcal{O}_{\mathbb{P}^1}(0)$. They obtained the result of Lemma 9.1 and proved the second statement of Theorem 9.3 in that case.



Figure 10: Fans (sections at z = 1) : Σ (left) and $\hat{\Sigma}$ (right). The generators $\vec{\omega}_1, \ldots, \vec{\omega}_4$ of ρ_1, \ldots, ρ_4 satisfy the relation $\vec{\omega}_1 + \vec{\omega}_3 = \vec{\omega}_2 + \vec{\omega}_4$.



Figure 11: Toric graphs Γ_X (left) and $\Gamma_{\hat{X}}$ (right).

9.2 Application to toric surface and its blowup

9.2.1 Small modification of toric Calabi–Yau threefolds

Let X be a toric Calabi–Yau threefold and Σ be its fan. Let $\sigma \in \Sigma_3$ be a 3-cone such that one of its three 2-dimensional faces τ_0 lies on the boundary of the support of the fan $|\Sigma|$. Let $\hat{\Sigma}$ be the following fan (see Figure 10):

$$\hat{\Sigma} = (\Sigma \setminus \{\tau_0, \sigma\}) \cup \{\rho_4, \hat{\tau}_0, \tau_3, \tau_4, \hat{\sigma}_1, \hat{\sigma}_2\},\$$

and let \hat{X} be the toric Calabi–Yau threefold associated with the fan Σ . We call \hat{X} a toric Calabi–Yau threefold obtained from X by a small modification. We compare the partition function of \hat{X} and that of X. For the rational curves, edges and formal variables, we use the following notations in Table 4. Note that the rational curve \hat{C}_0 corresponding to $\hat{\tau}_0$ is a (-1, -1)-curve.

	X		Â	
2-cone	$ au_1, au_2$	au	$\hat{ au}_0, au_1, au_2$	au
curve	C_1, C_2	$C_{ au}$	$\hat{C}_0, \hat{C}_1, \hat{C}_2$	$\hat{C}_{ au}$
edge	e_1, e_2	e_{τ} or just e	$\hat{e}_0, \hat{e}_1, \hat{e}_2$	e_{τ} or just e
variable	Q_1, Q_2	Q_e	$\hat{Q}_0, \hat{Q}_1, \hat{Q}_2$	Q_e

Table 4:

Let Γ_X be a toric graph of X. Near the edge corresponding to the 2-cone τ_0 , the graph looks like the left diagram in Figure 11. Figure 12 shows an example of small modifications of toric Calabi–Yau threefolds. After the small modification, the toric diagram (and the framings) changes as follows.



Figure 12: Toric graphs for a small modification of local $\mathbb{P}^1 \times \mathbb{P}^1$ (left) and its flop (right). Compare with Figure 6.

Lemma 9.6. A graph obtained from Γ_X by replacing the left diagram in Figure 11 with the right is a toric graph of \hat{X} .

We study the transformation of the partition function under a small modification. We associate the same formal variables $\vec{Q} = (Q_e)$ to edges in $E_3(\Gamma_X) \setminus \{e_1, e_2\}$ and those in $E_3(\Gamma_{\hat{X}}) \setminus \{\hat{e}_0, \hat{e}_1, \hat{e}_2\}$ and write the partition functions of X and \hat{X} as $Z_X(q, \vec{Q}, Q_1, Q_2)$ and $Z_{\hat{X}}(q, \vec{Q}, \hat{Q}_0, \hat{Q}_1, \hat{Q}_2)$. It is immediate to check that

$$Z_{\hat{X}}(q, \vec{0}, \hat{Q}_0, 0, 0) = Z_{(-1, -1)}(q, \hat{Q}_0).$$
(9.2)

We define

$$Z'_{\hat{X}}(q,\vec{Q},\hat{Q}_0,\hat{Q}_1,\hat{Q}_2) = \frac{Z_{\hat{X}}(q,\vec{Q},\hat{Q}_0,\hat{Q}_1,\hat{Q}_2)}{Z_{\hat{X}}(q,\vec{0},\hat{Q}_0,0,0)}$$

The main result of this section is

Theorem 9.7. (i) Coefficients of $\vec{Q}^{\vec{d}}\hat{Q}_{0}^{d_{0}}\hat{Q}_{1}^{d_{1}}\hat{Q}_{2}^{d_{2}}$ in $Z'_{\hat{X}}(q, \vec{Q}, \hat{Q}_{0}, \hat{Q}_{1}, \hat{Q}_{2})$ vanish unless $d_{1} + d_{2} \geq d_{0}$.

(ii)

$$Z_X(q, \vec{Q}, Q_1, Q_2) = Z'_{\hat{X}}(q, \vec{Q}, Q_0^{-1}, Q_1 Q_0, Q_2 Q_0)|_{Q_0 \to 0}$$

Proof. The statement (i) follows from Corollary 7.6. To prove (ii), we consider the toric Calabi– Yau threefold \hat{X}^+ obtained from \hat{X} by the flop of \hat{C}_0 . Let Q_0 , \hat{Q}_1^+ and \hat{Q}_2^+ be the formal variables correspond to the flopped curves \hat{C}_0^+ , \hat{C}_1^+ and \hat{C}_2^+ respectively. Let $\hat{\Sigma}^+$ be the fan of \hat{X}^+ . A natural inclusion $\Sigma \hookrightarrow \hat{\Sigma}^+$ induces that of toric varieties $X \hookrightarrow \hat{X}^+$. Under this map, we identify \hat{Q}_1^+ and \hat{Q}_2^+ with Q_1 and Q_2 respectively. Then by Theorem 7.9, we have

$$Z'_{\hat{X}^+}(q, \vec{Q}, Q_0, Q_1, Q_2) = Z'_{\hat{X}}(q, \vec{Q}, Q_0^{-1}, Q_1 Q_0, Q_2 Q_0)$$

Then (ii) follows, since the $Q_0 \to 0$ limit of the LHS is equal to $Z_X(q, \vec{Q}, Q_1, Q_2)$.

9.2.2 Toric surface and its blowup

Let S be a complete smooth toric surface and \hat{S} its blowup at a torus fixed point. The exceptional curve of $p: \hat{S} \to S$ is denoted by E. Let $X = K_S$ and $\hat{X} = K_{\hat{S}}$. Then \hat{X} is a small modification of X and E is a (-1, -1)-curve on \hat{X} added by the small modification. Applying Theorem 9.7 to this case, we obtain the following

Corollary 9.8. (i) For $\beta \in H_2(\hat{S}, \mathbb{Z})$ such that β is not an multiple of E and satisfying $\beta . E < 0$,

$$N_{g,\beta}(K_{\hat{S}}) = 0$$

(ii) For $\beta \in H_2(\hat{S}, \mathbb{Z})$ such that $\beta . E = 0$,

$$N_{g,\beta}(K_{\hat{S}}) = N_{g,p_*\beta}(K_S)$$
.

(iii) For a multiple of [E],

$$N_{g,d[E]}(K_{\hat{S}}) = N_{g,d}(\mathcal{O}_{\mathbb{P}^1}(-1) \oplus \mathcal{O}_{\mathbb{P}^1}(-1)).$$

Proof. The statements (i) and (ii) follow from Theorem 9.7 (i) and (ii) respectively. The statement (iii) follows from (9.2).

Especially, Gromov–Witten invariants of K_S are obtained from those of $K_{\hat{S}}$. The same statements also hold for the Gopakumar–Vafa invariants. See e.g. [11, Table 1, 10] for some related numerical data.

9.3 Example and geometric engineering

In this section, we first give an example of §9.1. Then we will discuss its relation with Nekrasov's partition function [70] along the same lines with [36, 37, 22, 23, 83].

Let X and X^+ be the toric Calabi–Yau threefolds associated with the left and right toric graphs in Figure 13, respectively. X contains two copies of $\mathbb{P}^1 \times \mathbb{P}^1$ disjoint to each other and X^+ is obtained by a flop of a unique (-1, -1)-curve in X. In this example, formal variables should be assigned as in Figure 13: the five variables for X are independent and the nine variables for X^+ have the four relations $Q_{F_i} = Q_{F_i}^+ Q_0^+$ (i = 1, 2) and $Q_{B_i} = Q_{B_i}^+ Q_0^+$ (i = 1, 2). The variables of X and X^+ should be identified by $Q_0^+ = Q_0^{-1}$ and Q_{F_i}, Q_{B_i} of $X^+ = Q_{F_i}, Q_{B_i}$ of X.

Let us compute the partition function Z_X of X (we omit the variables). By Proposition 7.4, we have

$$Z_X = \sum_{\mu_1^1, \mu_2^1, \mu_1^2, \mu_2^2} \prod_{k=1}^2 (Q_{B_k})^{|\mu_1^k| + |\mu_2^k|} s_{\mu_1^k}^2(q^{\rho}) s_{\mu_2^k}^2(q^{\rho}) \prod_{i,j \ge 1} \left(1 - Q_{F_k} q^{h_{\mu_1^k, (\mu_2^k)^t}(i,j)} \right)^{-2} \sum_{\lambda} (-Q_0)^{|\lambda|} s_{\lambda^t}(q^{\mu_2^1 + \rho}, Q_{F_1} q^{\mu_1^1 + \rho}) s_{\lambda}(q^{(\mu_2^2)^t + \rho}, Q_{F_2} q^{(\mu_1^2)^t + \rho}) .$$

We can perform the sum in the last factor by (7.13):

$$\sum_{\lambda} (-Q_0)^{|\lambda|} s_{\lambda^t} (q^{\mu_2^1 + \rho}, Q_{F_1} q^{\mu_1^1 + \rho}) s_{\lambda} (q^{(\mu_2^2)^t + \rho}, Q_{F_2} q^{(\mu_1^2)^t + \rho})$$

$$= \prod_{i,j \ge 1} \left(1 - Q_0 q^{h_{\mu_2^1, (\mu_2^2)^t}(i,j)} \right) \left(1 - Q_0 Q_{F_1} q^{h_{\mu_1^1, (\mu_2^2)^t}(i,j)} \right)$$

$$\left(1 - Q_0 Q_{F_2} q^{h_{\mu_2^1, (\mu_1^2)^t}(i,j)} \right) \left(1 - Q_0 Q_{F_1} Q_{F_2} q^{h_{\mu_1^1, (\mu_1^2)^t}(i,j)} \right)$$

Therefore Theorem 9.3 implies that the partition function Z_{X^+} is obtained from Z_X by replacing

$$\prod_{i,j\geq 1} \left(1 - Q_0 q^{h_{\mu_2^1,(\mu_2^2)^t}(i,j)} \right) \to \prod_k \left(1 - (Q_0^+)^{-1} q^k \right)^{C_k(\mu_2^1,(\mu_2^2)^t)} \prod_{k\geq 1} \left(1 - Q_0^+ q^k \right)^k$$

and replacing Q_0 in other factors by $(Q_0^+)^{-1}$.

From the discussions in [42, §2.1], it seems natural to expect that the partition function of X reproduces Nekrasov's partition function for a gauge theory with a product gauge group and with a matter. We want to clarify this statement. Let us set

$$Z_X^{\text{inst}} = \frac{Z_X}{Z_X |_{Q_{B_1} = Q_{B_2} = 0}} \,.$$

Then, by the same method with [22, 83], we can show the following

Proposition 9.9. Let

$$q = e^{-2R\hbar}$$
, $Q_{F_1} = e^{-4Ra_1}$, $Q_{F_2} = e^{-4Ra_2}$, $Q_0 = e^{2R(a_1 + a_2 - m)}$

Then we have

$$Z_X^{\text{inst}} = \sum_{\mu_1^1, \mu_2^1, \mu_2^2} \prod_{k=1}^2 \left(\frac{Q_{B_k}}{2^4 Q_{F_k}} \right)^{|\mu_1^k| + |\mu_2^k|} \prod_{l,n=1}^2 \prod_{i,j \ge 1} \frac{\sinh R \left(a_{ln}^{(k)} + \hbar \left(\mu_{1,i}^k - \mu_{2,j}^k + j - i \right) \right)}{\sinh R \left(a_{ln}^{(k)} + \hbar \left(j - i \right) \right)} \\ \times q^{\frac{1}{2} \left(\kappa(\mu_1^1) + \kappa(\mu_2^1) - \kappa(\mu_2^2) - \kappa(\mu_2^2) \right)} \left(2^2 Q_0 \right)^{|\mu_1^1| + |\mu_2^1| + |\mu_2^2|} \left(Q_{F_1}^{\frac{1}{2}} \right)^{2|\mu_1^1| + |\mu_2^2|} \left(Q_{F_2}^{\frac{1}{2}} \right)^{|\mu_1^1| + |\mu_2^1| + |\mu_2^1| + |\mu_2^2|} \\ \prod_{l,n=1}^2 \prod_{i,j \ge 1} \frac{\sinh R \left(a_{ln}^{(1,2)} + m + \hbar \left(j - i \right) \right)}{\sinh R \left(a_{ln}^{(1,2)} + m + \hbar \left(\mu_{l,i}^1 - \mu_{n,j}^2 + j - i \right) \right)}$$

,

where

$$a_{11}^{(k)} = a_{22}^{(k)} = 0$$
, $a_{12}^{(k)} = -a_{21}^{(k)} = 2a_k$

and

$$a_{11}^{(1,2)} = a_1 + a_2$$
, $a_{21}^{(1,2)} = -a_1 + a_2$, $a_{12}^{(1,2)} = a_1 - a_2$, $a_{22}^{(1,2)} = -a_1 - a_2$.

By Proposition 9.9, it is easy to see that the $R \rightarrow 0$ limit of

$$Z_X^{\text{IIISL}}|_{q=e^{-2R\hbar}, Q_{B_k}=2^2\Lambda_k, Q_{F_k}=e^{-4Ra_k}, Q_0=e^{2R(a_1+a_2-m)}$$

is equal to the instanton part of Nekrasov's partition function of 4-dimensional $SU(2) \times SU(2)$ gauge theory with a matter in the bifundamental representation $(\mathbf{2}, \mathbf{\bar{2}})$ (cf. [70, (66)]). It is straightforward to generalize this example to those which engineer more general quiver gauge theories of type A.

Remark 9.10. It is immediate to see that $Z_{X^+}^{\text{inst}} = Z_{X^+}/(Z_{X^+}|_{Q_{B_1}=Q_{B_2}=0})$ also coincides with the same Nekrasov's partition function with a similar variable identification in the limit $R \to 0$. More generally, Theorem 9.3 may imply that if toric Calabi–Yau threefolds X and X^+ are related by flops with respect to (-1, -1)-curves and if the partition function of X reproduces Nekrasov's partition function for a gauge theory, then the partition function of X^+ also reproduces it ¹¹.

¹¹Showing this statement, which was suggested by Yuji Tachikawa, was the author's original motivation for the problem. The author is grateful to him for discussions in August 2005 at Santa Barbara.



Figure 13: Toric Calabi–Yau threefold which contains two disjoint $\mathbb{P}^1 \times \mathbb{P}^1$ connected by a (-1, -1)-curve (left) and its flop (right).

References

- [1] Mina Aganagic, Marcos Mariño and Cumrun Vafa, All loop topological string amplitudes from Chern-Simons theory, Comm. Math. Phys. **247** (2004), 467–512.
- [2] Mina Aganagic, Albrecht Klemm, Marcos Mariño and Cumrun Vafa, *The topological vertex*, Comm. Math. Phys. 254 (2005), 425–478.
- [3] Michel F. Atiyah and Roul Bott, The moment map and equivariant cohomology, Topology 23 (1984), 1–28.
- [4] Wolf P. Barth, Chris A. Peters and Antonius Van de Ven, *Compact complex surfaces*, Ergebnisse der Mathematik und ihrer Grenzgebiete (3) 4, Springer-Verlag, Berlin, 1984.
- [5] Victor V. Batyrev, Dual Polyhedra and Mirror Symmetry for Calabi-Yau Hypersurfaces in Toric Varieties, Preprint, Univ.-GH-Essen, 1992.
- [6] Kai Behrend and Yuri Manin, Stacks of stable maps and Gromov-Witten invariants, Duke Math. J. 85 (1996), 1–60.
- [7] Kai Behrend, Gromov-Witten invariants in algebraic geometry, Invent. Math. 127 (1997), 601–617.
- [8] Kai Behrend and Barbara Fantechi, The intrinsic normal cone, Invent. Math. 128 (1997), 45–88.
- [9] Raoul Bott, A residue formula for holomorphic vector-fields, J. Differential Geometry 1 (1967), 311–330.
- [10] Tom Coates and Alexander Givental, Quantum Riemann-Roch, Lefschetz and Serre, arXiv:math.AG/0110142.

- [11] Ti-Ming Chiang, Albrecht Klemm, Shing-Tung Yau and Eric Zaslow, Local Mirror Symmetry: Calculations and Interpretations, Adv. Theor. Math. Phys. 3 (1999), 495–565.
- [12] David A. Cox and Sheldon Katz, Mirror symmetry and algebraic geometry, Mathematical Surveys and Monographs 68, American Mathematical Society, Providence, RI, 1999.
- [13] Pierre Deligne and David Mumford, The irreducibility of the space of curves of given genus, Inst. Hautes Études Sci. Publ. Math. No. 36 (1969), 75–109.
- [14] Duiliu-Emanuel Diaconescu, Bogdan Florea and Antonella Grassi, Geometric transitions and open string instantons, Adv. Theor. Math. Phys. 6 (2003), 619–642.
- [15] Duiliu-Emanuel Diaconescu, Bogdan Florea and Antonella Grassi, Geometric transitions, del Pezzo surfaces and open string instantons, Adv. Theor. Math. Phys. 6 (2003), 643–702.
- [16] Duiliu-Emanuel Diaconescu and Bogdan Florea, Localization and gluing of topological amplitudes, Comm. Math. Phys. 257 (2005), 119–149.
- [17] Duiliu-Emanuel Diaconescu, Bogdan Florea and Natalia Saulina, A Vertex Formalism for Local Ruled Surfaces, Comm. Math. Phys. 265 (2006), 201–226.
- [18] Duiliu-Emanuel Diaconescu and Bogdan Florea, The Ruled Vertex and Nontoric del Pezzo Surfaces, JHEP 0612, 028 (2006), 18 pp. (electronic).
- [19] Igor Dolgachev, Weyl groups and Cremona transformations, in Singularities, Part 1 (Arcata, Calif., 1981), 283–294, Proc. Sympos. Pure Math., 40, Amer. Math. Soc., Providence, RI, 1983.
- [20] Dan Edidin and William Graham, Equivariant intersection theory, Invent. Math. 131 (1998), 595–634.
- [21] Dan Edidin and William Graham, Localization in equivariant intersection theory and the Bott residue formula, Amer. J. Math. 120 (1998), 619–636.
- [22] Tohru Eguchi and Hiroaki Kanno, Topological strings and Nekrasov's formulas, JHEP 0312, 006 (2003), 30 pp. (electronic).
- [23] Tohru Eguchi and Hiroaki Kanno, Geometric transitions, Chern-Simons gauge theory and Veneziano type amplitudes, Phys. Lett. B585 (2004), 163-172.
- [24] Carel Faber and Rahul Pandharipande, Hodge integrals and Gromov-Witten theory, Invent. Math. 139 (2000), 173–199.
- [25] Kenji Fukaya and Kaoru Ono, Arnold conjecture and Gromov-Witten invariant, Topology 38 (1999), 933–1048.
- [26] William Fulton, Intersection theory, Ergebnisse der Mathematik und ihrer Grenzgebiete (3), 2. Springer-Verlag, Berlin, 1984.
- [27] William Fulton, Introduction to Toric Varieties, Annals of Math. Studies 131, Princeton Univ. Press, 1993.
- [28] William Fulton and Rahul Pandharipande, Notes on stable maps and quantum cohomology, in Algebraic geometry—Santa Cruz 1995, 45–96, Proc. Sympos. Pure Math., 62, Part 2, Amer. Math. Soc., Providence, RI, 1997.

- [29] Rajesh Gopakumar and Cumrun Vafa, *M-Theory and Topological Strings-II*, arXiv:hep-th/9812127.
- [30] Tom Graber and Rahul Pandharipande, Localization of Virtual Classes, Invent. Math. 135 (1999), 487–518.
- [31] Robin Hartshorne, Algebraic Geometry, Graduate Texts in Mathematics 52, Springer-Verlag, New York-Heidelberg, 1977.
- [32] Kentaro Hori, Sheldon Katz, Albrecht Klemm, Rahul Pandharipande, Richard Thomas, Cumrun Vafa, Ravi Vakil and Eric Zaslow, *Mirror symmetry*, Clay Mathematics Monographs, 1, American Mathematical Society, Providence, RI; Clay Mathematics Institute, Cambridge, MA, 2003.
- [33] Shinobu Hosono, Masa-Hiko Saito and Atsushi Takahashi, Relative Lefschetz action and BPS state counting, Internat. Math. Res. Notices 2001, no. 15, 783–816.
- [34] Shinobu Hosono, Counting BPS states via Holomorphic Anomaly Equations, in Calabi-Yau varieties and mirror symmetry (Toronto, ON, 2001), 57–86, Fields Inst. Commun. 38, Amer. Math. Soc., Providence, RI, 2003.
- [35] Amer Iqbal, All Genus Topological String Amplitudes and 5-brane Webs as Feynman Diagrams, arXiv:hep-th/0207114.
- [36] Amer Iqbal and Amir-Kian Kashani-Poor, Instanton counting and Chern-Simons theory, Adv. Theor. Math. Phys. 7 (2004), 457–497.
- [37] Amer Iqbal and Amir-Kian Kashani-Poor, SU(N) geometries and topological string amplitudes, Adv. Theor. Math. Phys. 10 (2006), 1–32.
- [38] Amer Iqbal and Amir-Kian Kashani-Poor, The vertex on a strip, Adv. Theor. Math. Phys. 10 (2006), 317–343.
- [39] Birger Iversen, A fixed point formula for action of tori on algebraic varieties, Invent. Math. 16 (1972), 229–236.
- [40] Hiroaki Kanno, Correspondence of topological string theory and gravity/gauge theories (Japanese), in Correspondence of topological string theory and gravity/gauge theories, 44–113, Sem. Math. Sci., 32, Keio Univ., Yokohama, 2005.
- [41] Hiroaki Kanno, Universal character and large N factorization in topological gauge/string theory, Nucl. Phys. B745 (2006), 165–175.
- [42] Sheldon Katz, Peter Mayr and Cumrun Vafa, Mirror symmetry and exact solution of 4D N = 2 gauge theories. I, Adv. Theor. Math. Phys. 1 (1998), 53–114.
- [43] Sheldon Katz, Albrecht Klemm and Cumurun Vafa, M-Theory, Topological Strings and Spinning Black Holes, Adv. Theor. Math. Phys. 3 (1999), 1445–1537.
- [44] Albrecht Klemm and Eric Zaslow, Local Mirror Symmetry at Higher Genus, in Winter School on Mirror Symmetry, Vector Bundles and Lagrangian Submanifolds (Cambridge, MA, 1999), 183–207, AMS/IP Stud. Adv. Math. 23, Amer. Math. Soc., Providence, RI, 2001.

- [45] Finn F. Knudsen, The projectivity of the moduli space of stable curves. II. The stacks M_{g,n}, Math. Scand. **52** (1983), 161–199.
- [46] Kunihiko Kodaira, Louis Nirenberg and Donald C. Spencer, On the existence of deformations of complex analytic structures, Ann. of Math. (2) 68 (1958), 450–459.
- [47] Kunihiko Kodaira, On stability of compact submanifolds of complex manifolds, Amer. J. Math. 85 (1963), 79–94.
- [48] Yukiko Konishi, Pole structure of topological string free energy, Publ. RIMS, Kyoto Univ. 42 (2006), 173–219.
- [49] Yukiko Konishi, Integrality of Gopakumar-Vafa invariants of toric Calabi-Yau threefolds, Publ. RIMS, Kyoto Univ. 42 (2006), 605–648.
- [50] Yukiko Konishi and Satoshi Minabe, *Flop invariance of the topological vertex*, arXiv:math.AG/0601352.
- [51] Yukiko Konishi and Satoshi Minabe, Local Gromov-Witten invariants of cubic surfaces via nef toric degeneration, arXiv:math.AG/0607187.
- [52] Maxim Kontsevich, Enumeration of rational curves via torus actions, in The moduli space of curves (Texel Island, 1994), 335–368, Progr. Math. 129, Birkhäuser Boston, Boston, MA, 1995.
- [53] Andrew Kresch, Cycle groups for Artin stacks, Invent. Math. 138 (1999), 495–536.
- [54] Wolfgang Lerche, Peter Mayr and Nicholas P. Warner, Non-Critical Strings, del Pezzo Singularities and Seiberg-Witten Curves, Nucl. Phys. B499 (1997), 125–148.
- [55] An-Min Li and Yongbin Ruan, Symplectic surgery and Gromov-Witten invariants of Calabi-Yau 3-folds I, Invent. Math. 145 (2001), 151–218.
- [56] Jun Li and Gang Tian, Virtual moduli cycles and Gromov-Witten invariants of algebraic varieties, J. Amer. Math. Soc. 11 (1998), 119–174.
- [57] Jun Li and Gang Tian, Virtual moduli cycles and Gromov-Witten invariants of general symplectic manifolds, in Topics in symplectic 4-manifolds (Irvine, CA, 1996), 47–83, First Int. Press Lect. Ser., I, Int. Press, Cambridge, MA, 1998.
- [58] Jun Li and Gang Tian, Comparison of the algebraic and the symplectic Gromov-Witten invariants, Asian J. Math. 3 (1999), 689–728.
- [59] Jun Li, Chiu-Chiu Melissa Liu, Kefeng Liu and Jian Zhou, A Mathematical Theory of the Topological Vertex, arXiv:math.AG/0408426.
- [60] Chiu-Chu Melissa Liu, Kefeng Liu and Jian Zhou, A Formula of Two Partition Hodge Integrals, J. Amer. Math. Soc. 20 (2007), 149-184.
- [61] Chiu-Chu Melissa Liu, Formulae of one-partition and two-partition Hodge integrals, Geometry and Topology Monographs 8 (2006), 105–128.
- [62] Ian G. Macdonald, Symmetric functions and Hall polynomials, Second edition, Oxford Mathematical Monographs, Oxford Science Publications, The Clarendon Press, Oxford University Press, New York, 1995.

- [63] Marcos Mariño, Chern-Simons theory, matrix models and topological strings, International Series of Monographs on Physics, 131, The Clarendon Press, Oxford University Press, Oxford, 2005.
- [64] Davesh Maulik, Nikita Nekrasov, Andrei Okounkov and Rahul Pandharipande, Gromov-Witten theory and Donaldson-Thomas theory, I, Compos. Math. 142 (2006), 1263–1285.
- [65] Davesh Maulik, Nikita Nekrasov, Andrei Okounkov and Rahul Pandharipande, Gromov-Witten theory and Donaldson-Thomas theory, II, Compos. Math. 142 (2006), 1286–1304.
- [66] Joseph A. Minahan, Dennis Nemeschansky and Nicholas P. Warner, *Investigating the BPS Spectrum of Non-Critical* E_n Strings, Nucl. Phys. **B508** (1997), 64–106.
- [67] Shigefumi Mori and Shigeru Mukai, On Fano 3-Folds with $B_2 \ge 2$, Adv. Stud. Pure Math. 1 (1983), 101–129.
- [68] James Morrow and Kunihiko Kodaira, Complex manifolds, Holt, Rinehart and Winston, Inc., New York-Montreal, Que.-London, 1971.
- [69] Nikita Nekrasov, Seiberg-Witten prepotential from instanton counting, Adv. Theor. Math. Phys 7 (2003), 831-864.
- [70] Nikita Nekrasov, Lectures on nonperturbative aspects of supersymmetric gauge theories, Class. Quant. Grav. 22 (2005), S77–S105.
- [71] Andrei Okounkov and Rahul Pandharipande, The equivariant Gromov-Witten theory of P¹, Ann. of Math. (2) 163 (2006), 561–605.
- [72] Andrei Okounkov, Nikolai Reshetikhin and Cumrun Vafa, Quantum Calabi-Yau and classical crystals, in The unity of mathematics, 597–618, Progr. Math., 244, Birkhäuser Boston, Boston, MA, 2006.
- [73] Tadao Oda, Convex bodies and algebraic geometry, Ergebnisse der Mathematik und ihrer Grenzgebiete (3) 15, Springer-Verlag, Berlin, 1988.
- [74] Rahul Pandharipande, Hodge integrals and degenerate contributions, Comm. Math. Phys. 208 (1999), 489–506.
- [75] Pan Peng, A simple proof of Gopakumar-Vafa conjecture for local toric Calabi-Yau manifolds, arXiv:math.AG/0410540.
- [76] Bernd Siebert, Gromov-Witten invariants for general symplectic manifolds, Habilitationsschrift, Bochum, 1998. [arXiv:dg-ga/9608005].
- [77] Bernd Siebert, Algebraic and symplectic Gromov-Witten invariants coincide, Ann. Inst. Fourier (Grenoble) 49 (1999), 1743–1795.
- [78] Holger Spielberg, The Gromov-Witten invariants of symplectic toric manifolds, Thèse, Université Louis Pasteur (Strasbourg I), Strasbourg, 1999. [arXiv:math.AG/0006156].
- [79] Kazushi Ueda, Stokes Matrix for the Quantum Cohomology of Cubic Surfaces, arXiv:math.AG/0505350.
- [80] Jian Zhou, Proof and interpretation of a string duality, Math. Res. Lett. 11 (2004), 213– 229.

- [81] Jian Zhou, A conjecture on Hodge integrals, arXiv:math.AG/0310283.
- [82] Jian Zhou, Localizations on Moduli Spaces and Free Field Realizations of Feynman Rules, arXiv:math.AG/0310283.
- [83] Jian Zhou, Curve counting and instanton counting, arXiv:math.AG/0311237, to appear in Adv. Theor. Math. Phys..